

# Oldenburg Discussion Papers in Economics

## The Educational Burden of ADHD: Evidence From Student Achievement Test Scores

Bernhard C. Dannemann Erkan Gören

V - 408 - 18

April 2018

**Department of Economics** 

University of Oldenburg, D-26111 Oldenburg

THE EDUCATIONAL BURDEN OF ADHD:

EVIDENCE FROM STUDENT ACHIEVEMENT TEST SCORES \*

Bernhard C. Dannemann<sup>†</sup>

Erkan Gören<sup>‡</sup>

Carl von Ossietzky University Oldenburg

Carl von Ossietzky University Oldenburg

This Version: April 2018

Abstract

This paper hypothesizes and empirically establishes the educational burden of the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity on aggregate cognitive achievement outcomes. We use a novel compilation of the 2- and 7-repeat allele variants of the human DRD4 exon III gene that candidate gene association studies have identified as an important biomarker in the etiology of childhood ADHD. The main results show a negative and statistically significant association between aggregate international student achievement test scores and the DRD4 exon III 2- and 7-repeat allele frequency measure in a cross-section of 81 countries. This finding is robust to the inclusion of additional country-specific historical, cultural, socioeconomic, biogeographic, health-related, educational, genetic, and diversity factors. Additional estimates suggest the predictive power of the country-level DRD4 exon III 2- and 7-repeat allele frequency measure on cross-country differences of estimated ADHD prevalence rates, confirming the reliability of the proposed biomarker for the measurement of ADHD-related behavioral symptoms in the general population.

Keywords: Human Capital, Cognitive Ability, Cognitive Skills, International Student Achievement Tests,

Education Production Function, ADHD, DRD4 Exon III, Genetic Diversity

JEL Classification Numbers: E70, I10, I20, J24

<sup>\*</sup>We would like to thank seminar participants at the Carl von Ossietzky University Oldenburg 2017 for helpful comments and suggestions.

All remaining errors are our own.

<sup>&</sup>lt;sup>†</sup>Carl von Ossietzky University Oldenburg, School of Computing Science, Business Administration, Economics, and Law (Faculty II), Institute of Economics, Building A5, 26111 Oldenburg, Germany, Tel.: +49-441-798-2625, e-mail: bernhard.dannemann@uni-oldenburg.de.

<sup>&</sup>lt;sup>‡</sup>Carl von Ossietzky University Oldenburg, School of Computing Science, Business Administration, Economics, and Law (Faculty II), Institute of Economics, Building A5, 26111 Oldenburg, Germany, Tel.: +49-441-798-4292, e-mail: erkan.goeren@uni-oldenburg.de.

#### 1 Introduction

The distribution of cognitive skills has been identified as a key determinant of economic and social outcomes, at both the individual and macroeconomic level. The acquisition of cognitive skills significantly predicts individual labor market outcomes, the distribution of income in society, and overall economic growth. On the individual level, increased cognitive skills manifest themselves in higher earnings and job quality, a lower risk of unemployment, and an overall improved socioeconomic status (Cawley et al., 2001; Heckman et al., 2006; Hanushek et al., 2015), among other outcomes. Additionally, evidence from country-level studies suggests that aggregate measures of cognitive skills in society are positively related to higher long-run economic growth rates (Hanushek and Kimko, 2000; Hanushek and Woessmann, 2012). Thus, uncovering the main determinants behind the acquisition of cognitive skills in society should facilitate our understanding of observed differences in standard of living across countries.

The relevant human capital literature identifies a number of possible determinants of cognitive skill outcomes, including family inputs (e.g., parental education, socioeconomic status, and time devoted to child rearing), peer inputs (e.g., social interaction with classmates and friends), schooling inputs (e.g., class size), and innate ability factors (Card, 1999; Cunha and Heckman, 2007). Although empirical studies suggest that parenting, teacher quality, and the institutional framework of the educational system may also play a role in cognitive skills (Schuetz et al., 2008; Hanushek, 1997; Ammermueller and Pischke, 2009), the operationalization and quantitative assessment of innate ability factors has received relatively little attention in the research to date (Todd and Wolpin, 2007; Cunha et al., 2010). This finding is surprising insofar as individual-level studies suggest a considerable role of innate ability factors in the total variation in individual cognitive skill outcomes (Anger and Heineck, 2010; Schuetz et al., 2013).

In this paper, we examine the possible role of childhood Attention-Deficit/Hyperactivity Disorder (ADHD) and its implications for the acquisition of cognitive skills across the life span. The bulk of evidence from molecular genetic studies suggests that childhood ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity have a highly heritable component, thus qualifying them as an innate ability factor in the analysis of individual cognitive skill outcomes. In particular, the child's emotional capacity to pay attention to school tasks has been identified as an important determinant of individual learning success. ADHD represents one of the most common childhood mental disorders with a pooled worldwide prevalence rate in the range of 5.3% to 7.2%, depending on the type of diagnostic criteria used for classification (Polanczyk et al., 2007; Thomas et al., 2015). A large number of neuropsychological studies suggest that children with the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity face serious difficulties in everyday social life, particularly in situations at school and at home, leading to worse educational outcomes, such as less formal schooling, poor grades, and higher school drop-out rates (Mannuzza and Klein, 2000; Bauermeister et al., 2007; Langberg et al., 2011; Birchwood and Daley, 2012). In contrast to popular belief, symptoms of childhood ADHD do not disappear upon reaching adolescence or young adulthood (Mannuzza et al., 1993; Kessler et al., 2006). Instead, the developmental disadvantages and the lack of education and socialization persist even in older ages, leading to long-term consequences for individuals with ADHD, including poor academic achievement, low-skilled occupations, higher illness-related absence from work, and a higher number of workplace accidents (Mannuzza and Klein, 2000; Barkley et al., 2006; Klein et al.,

<sup>&</sup>lt;sup>1</sup>The bulk of evidence reported in Cunha and Heckman (2007) make it clear that early childhood development is critical for the acquisition of important cognitive skills and that any skill deficits can be made up only with much greater individual effort in later life.

#### 2012).

Our empirical approach combines various data sources to quantify the educational impact of ADHD on aggregate cognitive skill outcomes in a cross-section of countries. For the construction of the main dependent variable in the empirical analysis, we use aggregated student achievement test scores from two large-scale international student achievement tests – the Programme for International Student Assessment (PISA) and the Trends in Mathematics and Science Study (TIMSS) - to approximate the average level of cognitive ability in the general population. In these studies, students in a number of countries were evaluated over a number of years. We pool the various international student achievement tests, first across students and then by waves for each of the two studies. This procedure allows the aggregate student achievement test score to be used as a measure of cognitive ability in the general population rather than in the student population. More importantly, we use the frequency of the 2- and 7-repeat allele variants of the human dopamine receptor D4 (DRD4) exon III gene that candidate gene association studies have identified as an important factor in the emergence and etiology of behavioral symptoms of inattention and/or hyperactivity-impulsivity related to childhood ADHD. This measure was constructed in the study of Gören (2017a) by matching the entire distribution of ethnic groups in the Alesina et al. (2003) ethnicity data to the DRD4 exon III population genome data in Gören (2016). We regard the resulting country-level DRD4 exon III 2- and 7-repeat allele frequency measure as a strong proxy variable for the prevalence of ADHD-related behavioral symptoms in society. Additionally, we use a large set of country-specific historical, cultural, socioeconomic, biogeographic, health-related, educational, genetic, and diversity factors to rule out endogeneity concerns in the analysis between aggregate student achievement test scores and the country-level DRD4 exon III 2- and 7-repeat allele frequency measure that might arise from the issue of omitted variables.

In a preliminary step, we report a positive and statistically significant relationship between the country-level DRD4 exon III 2- and 7-repeat allele frequency measure and estimated ADHD prevalence estimates. This finding establishes the key hypothesis proposed in this paper: that the DRD4 exon III gene is able to predict the extent of ADHD-related behavioral symptoms across countries. Considering the estimated magnitudes, the empirical analysis suggests that a one percentage point increase in the country-level DRD4 exon III 2- and 7-repeat allele frequency measure would, ceteris paribus, increase the ADHD prevalence rate in the general population by about 0.4218 percentage points. This finding is robust to the inclusion of region fixed effects, and to a full set of cultural, historical, and socioeconomic controls. Nevertheless, we refrain from using estimated ADHD prevalence rates in the empirical analysis for several reasons. First, as shown below, estimated country-specific ADHD prevalence rates are heavily confounded by study-level methodological characteristics, including ADHD diagnostic criteria, informant source for ADHD behavioral symptoms, type of population sample, whether full or partial diagnostic criteria for ADHD were met, type of ADHD measurement, and the geographical region of the study, accounting for approximately 48% of the total variation in estimated ADHD prevalence rates across individual studies. This finding casts serious doubts on the reliability of reported ADHD cases across studies and therefore countries. Second, estimated ADHD prevalence rates could be prone to population stratification bias, as the studies are not representative of their respective populations (e.g., in terms of gender, age, ethnicity, or socioeconomic status), again limiting the external validity of estimated ADHD prevalence rates in the general population. More importantly, data on estimated ADHD prevalence rates are only available for a limited sample of 42 countries, most of which can be considered as highly developed economically. This might result in inconsistent Ordinary Least Squares (OLS) estimates if the selection process underlying the inclusion of countries is endogenously determined (Wooldridge, 2010, Chapter 19).

The baseline results suggest a negative and statistically significant impact of the country-level DRD4 exon III 2- and 7-repeat allele frequency measure on aggregated student achievement test scores in a cross-section of 81 countries. This finding is consistent with the macroeconomic burden of inattention and/or hyperactivity-impulsivity as behavioral symptoms of ADHD for school learning outcomes and subsequent educational achievement. Regarding the estimated magnitude, the results suggest that a one percentage point increase of the country-level DRD4 exon III 2- and 7-repeat allele frequency measure<sup>2</sup> would, ceteris paribus, result in a reduction in aggregate student achievement test scores of about 3.8765 points. This effect is quite sizable as it corresponds approximately to a  $(0.0388/0.6240) \times 100 \approx 6.2180$  percentage change in the standard deviation of the aggregate student achievement test score variable. It is worth mentioning that the revealed association between the aggregate student achievement test score and the country-level DRD4 exon III 2- and 7-repeat allele frequency measure is robust to the inclusion of historical, biogeographic, health-related, socioeconomic, educational, genetic, diversity, and regional factors. These model specifications effectively rule out endogeneity concerns due to unobserved country-specific heterogeneity that might be correlated with both general cognitive ability and the prevalence of ADHD related behavioral symptoms of inattention and/or hyperactivity-impulsivity in society.

We report additional sensitivity tests to examine the robustness in the association between aggregate student achievement test scores and the country-level DRD4 exon III 2- and 7-repeat allele frequency measure. In a first exercise, we test the robustness of the main findings to the inclusion of various measures of genetic, ethno-linguistic, and religious diversity, to rule out the possibility that the estimated regression coefficients associated with the country-level DRD4 exon III 2- and 7-repeat allele frequency variable captures the impact of various forms of societal diversity on general cognitive ability. Next, we report coefficient estimates based on PISA international student test score outcomes only. This specification is intended to rule out the possibility that the results might be driven by the inclusion of the TIMSS student achievement test scores in the empirical analysis. In addition, we examine heterogeneous effects of the country-level DRD4 exon III 2- and 7-repeat allele frequency measure on the distinct PISA subjects – i.e., aggregated student-level mathematics and science test score outcomes – to ensure that the main findings were not driven by the individual PISA subjects and to underline the educational burden of the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity on general school learning outcomes. Since the association between aggregate student achievement test scores and the country-level DRD4 exon III 2- and 7-repeat allele frequency measure only provides an indirect link to the educational burden of behavioral problems related to ADHD, we report coefficient estimates that directly employ estimated ADHD prevalence rates as the key explanatory variable. Based on a reduced sample of 37 countries, we report coefficient estimates from both simple OLS and Instrumental Variables (IV) estimation approaches to cope with simultaneity in the association between aggregate student achievement test scores and reported ADHD prevalence rates. None of these sensitivity tests alter the main conclusions substantially.

This paper is related to the literature on the main determinants behind the formation of individual cognitive abilities. The theoretical framework of this investigation is a specification of an Education Production Function (EPF) that relates individual cognitive achievement outcomes to various family, peer, schooling, and innate ability factors (Todd and Wolpin, 2003). The bulk of individual-level studies based on within-country and cross-country variation in international student achievement test scores suggest a possible role of individual characteristics (e.g., age, gender, immigration background, and

<sup>&</sup>lt;sup>2</sup>Given our prior finding, this corresponds to a rise in the estimated ADHD prevalence rate of about 0.4218 percentage points.

native language), family background (e.g., parental socioeconomic status), peer inputs (e.g., social interaction with other classmates and friends), and the institutional framework of the educational system (e.g., accountability, school autonomy, and competition from private schools) on students' cognitive achievement outcomes (Fuchs and Woessmann, 2007; Ammermueller and Pischke, 2009; Schuetz et al., 2013). Besides the findings from individual-level studies on cognitive skill formation, macro-level studies based on cross-country variation in aggregated student achievement test scores point to the importance of socioeconomic (e.g., the level of economic development, the government size, and conflict incidence), institutional (e.g., rule of law, democracy, and freedom from corruption), educational (e.g., schooling expenditures), and cultural (e.g., religiosity and gender equality) aspects of international differences in cognitive ability outcomes (Rindermann, 2008; Hanushek and Kimko, 2000; Guiso et al., 2008; Stoet and Geary, 2017). However, the importance of innate ability factors has been widely neglected in the relevant human capital literature, perhaps due to problems of data availability and the difficulty of modelling and quantifying such a latent factor in an empirical analysis that encompasses a wide variety of possible dimensions (e.g., genetic heritability, child socialization, and in utero experiences).

There is only scant research on the importance of innate ability factors for the acquisition of cognitive skills. For example, Anger and Heineck (2010) indirectly reveal the strong predictive power of innate ability factors for various measures of students' cognitive skills through the inclusion of parental cognitive ability into the regression equation. Other studies assess the importance of various genetic controls for cross-country differences in aggregate cognitive abilities (Minkov et al., 2016; Becker and Rindermann, 2016). Although there is substantial evidence in the neuropsychological literature that early childhood development of attention and/or impulse control has a significant and long-lasting impact on a wide array of private (e.g., individual academic, labor market, and health outcomes) and social (e.g., public safety) market returns (Moffitt et al., 2011), this is the first paper – to the best of the authors' knowledge – that empirically establishes the educational burden of ADHD-related behavioral symptoms on the accumulation of human capital at the macroeconomic level. We contribute to the relevant human capital literature by explicitly identifying an important innate ability factor (i.e., the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity) that have been examined frequently in the neuropsychological literature but largely neglected in the literature on the main determinants of observed international differences in cognitive skill outcomes.

The remainder of this paper is organized as follows. Section 2 reviews the relevant literature on diagnosis and symptoms of ADHD-related behavioral symptoms and its implications for subsequent individual economic and social life outcomes. Section 3 provides a detailed discussion of the main country-level controls employed in the empirical analysis. Section 4 provides evidence on the reliability of the country-level DRD4 exon III 2- and 7-repeat allele frequency measure as a strong biomarker indicating the prevalence of ADHD-related behavioral symptoms in society. In addition, we show that estimated ADHD prevalence rates derived from individual ADHD studies are heavily confounded by study-level methodological characteristics, thus casting serious doubts on the external validity of estimated ADHD prevalence rates for the general population. Section 5 highlights the importance of incorporating the issue of innate ability factors into the theoretical framework of the EPF and presents the econometric methodology regarding the association between aggregate student achievement test scores and the country-level DRD4 exon III 2- and 7-repeat allele frequency measure. The main empirical results are reported in Section 6. Section 7 provides additional sensitivity tests. Finally, Section 8 concludes by summarizing the main findings.

#### 2 Theoretical Background: ADHD and Cognitive Ability

According to the Diagnostic and Statistical Manual of Mental Disorders, Fifth Edition (DSM-5) published by the American Psychiatric Association (2013), several different subtypes of ADHD manifest themselves across the domains of inattention, hyperactivity/impulsivity, or combinations of the two in different settings (e.g., at home, school, workplace, and other social activities). A clinical diagnosis of ADHD requires that at least six criteria (out of nine) for inattention and/or six criteria (out of nine) for hyperactivity/impulsivity that significantly interfere with childhood mental development and executive functioning persist for at least six months before the age of 12. For the inattention subtype, individuals are characterized by deficits in focusing on specific tasks, a lack of persistence, poor time management skills, and disorganization. The key diagnostic criteria for the hyperactive type are behavioral symptoms of excessive physical activity, personal disruption, and talkativeness, whereas the impulsivity type is described as making hasty, impetuous movements or decisions, in some cases with negative outcomes for themselves or their immediate environment (e.g., crossing a street without considering the traffic flow) (American Psychiatric Association, 2013). Typically, parents, teachers, or clinical personnel observe ADHD probands in the relevant contextual settings and evaluate their behavioral symptoms based on rating scales.

Reduced academic achievement and performance is by far the most researched outcome associated with the behavioral and executive functioning deficits of individuals diagnosed with ADHD. Childhood behavioral symptoms of inattention (e.g., off-task behavior in the classroom, not listening to classroom tasks, and frequent distractions), hyperactivity (e.g., excessive fidgeting and difficulty remaining seated), and impulsivity (e.g., difficulty in response inhibition) have direct implications for academic productivity (Raggi and Chronis, 2006). There exist numerous studies showing that ADHD probands have significantly lower academic performance, including (i) less formal schooling, (ii) lower final grades, (iii) higher probability of grade retention, (iv) increased use of special education services, (v) higher school drop-out rates, and (vi) lower high-school graduation and university degrees (Mannuzza and Klein, 2000; Bauermeister et al., 2007; Langberg et al., 2011; Birchwood and Daley, 2012). Although recruitment procedures (e.g., clinical vs. non-clinical community samples), investigated age groups (e.g., pre-school children, school-aged children, adolescence and adult samples), experimental design (e.g., cross-sectional vs. longitudinal cohort study), and inclusion of potential confounding factors (e.g., age, gender, socioeconomic status of parents, and presence of comorbid disorders) differ considerably across studies, the finding that ADHD symptoms significantly interfere with an individual's academic performance has been robustly demonstrated in numerous replication studies, with the core results having been synthesized and summarized in various meta-analysis and qualitative review studies (Frazier et al., 2007; Loe and Feldman, 2007; Polderman et al., 2010; Willcutt et al., 2012; Arnold et al., 2015).

Several studies have attempted to explain the negative correlation between ADHD behavioral symptoms and poor academic performance. Results showed that overall cognitive ability is significantly lower among individuals impaired by ADHD. Children diagnosed with ADHD show poor outcomes on a wide range of neuropsychological measures, including intelligence, set shifting ability, impulse control, sustained attention, working memory, processing speed, reaction time, response variability, and vigilance, with findings being more pronounced especially for the ADHD inattention-type group (Pennington et al., 1993; Frazier et al., 2004; Willcutt et al., 2012). Since overall cognitive ability appears to be of pivotal importance to an individual's problem-solving capabilities, the findings further predict poor scores on standardized tests of

achievement in reading, writing, and mathematics among ADHD probands. Interestingly, the negative correlation between ADHD and the various measures of academic achievement (e.g., total mean grade and standardized test scores in reading, writing, and mathematics skills) remain relatively robust in studies accounting for the potential confounding influence of cognitive abilities, indicating that ADHD-related behavioral symptoms significantly predict academic underachievement beyond the influence of cognitive ability (Barry et al., 2002; Diamantopoulou et al., 2007; McConaughy et al., 2011). It is worth mentioning that even in representative community-based samples consisting of non-clinical participants, individuals displaying a continuum of ADHD-related symptoms perform as poorly on standard measures of academic achievement as their clinically diagnosed ADHD counterparts (Currie and Stabile, 2006; Rodriguez et al., 2007; Birchwood and Daley, 2012). The previous finding suggests that ADHD symptoms constitute a behavioral trait that is prevalent in the general population, which, depending on symptom severity, significantly interferes with future academic achievement and performance (Gray et al., 2017).

It has been reported that childhood ADHD persists significantly into adulthood (Mannuzza et al., 1993; Kessler et al., 2006). Consequently, the academic and cognitive deficits associated with behavioral symptoms of childhood ADHD have direct implications for future social, educational, occupational, and health functioning. Specifically, several prospective follow-up studies reported that childhood ADHD symptoms predict poor academic performance and occupational rank, low household income, low self-esteem, poor marital outcomes, give rise to an increase in antisocial personality and substance use disorders, social conflict with peers and family members, the propensity toward incarceration, early sexual intercourse, and the rate of sexually transmitted diseases in adulthood (Mannuzza and Klein, 2000; Barkley et al., 2006; Klein et al., 2012). Moreover, a 10-year prospective follow-up study indicates that ADHD probands had elevated risks of a wide range of adverse psychiatric disorders, including mood, anxiety, antisocial, and substance use disorders (Biederman et al., 2006). Additional studies have shown that mortality rates among ADHD probands are significantly elevated compared to normal controls, perhaps driven by the higher propensity toward risky behavior, traffic accidents, and suicide among individuals with ADHD (James et al., 2004; Olazagasti et al., 2013; Chang et al., 2014; Dalsgaard et al., 2015). In a study using Swedish national registers data, Lichtenstein et al. (2012) showed that ADHD probands had a higher risk of criminal behavior, and that pharmacological treatment results in a significant reduction in the risk of criminality. These results are in accordance with the findings that ADHD significantly worsens the quality of life in adults, including non-medical factors of satisfaction, comfort, resilience, risk avoidance, and achievement (Agarwal et al., 2012). Overall, the findings illustrate that ADHD probands face serious impairments during their life course, resulting in poor social and economic functioning and a significant reduction in their health-related quality of life and individual subjective well-being.

#### 3 Data and Variables

In this section, we provide a detailed discussion on data construction and sources of the main dependent variable associated with the measurement of cognitive ability based on international student achievement test outcomes. Additionally, we motivate the use of the country-level DRD4 exon III 2- and 7-repeat allele frequency measure as a strong biomarker (i.e., proxy variable) for the measurement of the prevalence of ADHD-related behavioral symptoms in the general population, as evidenced by a large number of candidate gene association studies on the etiology of ADHD. In addition to the main dependent and explanatory variable, we also use a large number of country-level controls in the estimation of the EPF

for two main reasons. First, the selectivity of these country-level controls is similar to the set of family, peer, and school inputs usually discussed in the empirical specification of individual EPFs. More importantly, the inclusion of a large set of country-level controls in the regression equation mitigates endogeneity concerns caused by unobserved country-specific heterogeneity that might be correlated with both international student achievement test outcomes and the distribution of the DRD4 exon III 2- and 7-repeat allele frequency measure. Thus, this kind of analysis ensures consistent estimation of the regression coefficients associated with the various input factors in the specification of the country-level EPF.

#### 3.1 Main Dependent Variable: Student Achievement Test Scores

Generally, estimates of an EPF use some measurement of students' educational performance or outcomes for the operationalization of cognitive abilities or skills. Notable examples range from armed forces qualification tests (Heckman et al., 2006) and national or international student assessments (Hanushek and Woessmann, 2008, 2012) to international comparisons of IQ test results Lynn and Vanhanen (2002, 2006, 2011). The meaningfulness and validity of country-level IQ measures has been subject to intense scientific discussion, especially regarding their validity for the measurement of cognitive ability in the general population and issues related to data quality (Volken, 2003; Rindermann, 2007; Lynn and Meisenberg, 2010). We use student achievement test scores derived from International Student Achievement Tests (ISATs) to assess international differences in cognitive abilities for two main reasons. First, student achievement test scores are widely accepted as a measure of the overall cognitive abilities of students and/or the general population. Furthermore, it is widely agreed that student test scores share a common approach to measuring general intelligence that is comparable with IQ tests (Rindermann, 2007; Lynn and Vanhanen, 2011), with IQ being a determinant of cognitive skill formation (Lynn and Mikk, 2009). Second, ISATs are designed to be representative of the corresponding country's population and are thus suitable for measuring cognitive ability in the general population.

For the operationalization of cognitive ability, we use student performance outcomes from two large-scale scholastic assessment tests: the Programme for International Student Assessment (PISA) and Trends in Mathematics and Science Study (TIMSS). Up to the present day, six waves of the PISA study have been conducted at three-year intervals starting in 2000. The PISA study is targeted at the 15-year-old population of the participating countries, with participants varying across the different assessment waves. Despite being designed and maintained by the OECD, a large number of non-OECD member countries have participated in every wave of the study. The main concept is a cross-sectional design without any follow-up test for individual students, although national variants of PISA do exist. The OECD selects schools and students to be representative of each country's population of 15-year-olds. The second study, TIMSS, is conducted by the International Association for the Evaluation of Educational Achievement (IEA) at four-year intervals, starting in 1995, yielding a total of six waves up to the present day.<sup>3</sup> In contrast to PISA, TIMSS is administered to eighth-graders, which means that students are slightly younger than those in the PISA study.<sup>4</sup> The student sample is again chosen to be nationally representative of the respective population.

<sup>&</sup>lt;sup>3</sup>Note that the IEA has also conducted prior studies, FIMS/FISS (1964/1970) and SIMS/SISS (1980,1983). However, these studies were not conducted at regular intervals and differ from today's studies in terms of test design and comparability. Thus, we do not consider these studies in the construction of the aggregate cognitive ability measure.

<sup>&</sup>lt;sup>4</sup>Additional tests are conducted in the fourth grade and in the final year of secondary schooling, twelfth grade, but these are omitted from the current analysis as the age groups in these additional studies do not match those in the PISA studies.

Both ISATs are developed as paper-based student tests with questionnaires designed in the country's local language. Student test scores are calculated using item response theory (IRT) and both tests cover the subjects of mathematics and science. Reading, which is tested in the PISA study, is not part of the eighth grade questionnaire in TIMSS. By addressing a well-defined student sample and through consistent and comparable tasks, these student assessments give both broad and precise estimates of average scholastic achievement in a country. The PISA studies aim to measure real-life problem-solving skills rather than mastery of school curricula or parrot-fashion learning and are regarded as measures of applied ideas (Hanushek and Woessmann, 2008, 660), while TIMSS is based on shorter items that are closer to school curricula than the PISA studies and contain questions involving memorization rather than application (Rindermann, 2007, 670). All PISA studies consist of three subject domains – mathematics, science, and reading – with one of the subjects being the major and the other two being minor domains of the tests across the different waves of the assessment.

The data from both PISA and TIMSS are available for each assessment wave as student-level observations. For the macroeconomic analysis, in the first step, we aggregate student test scores to the country level using the weighted average for the subjects of mathematics and science in each ISAT and assessment wave. We employ the student probability weights provided to ensure representativity. In the second step, we calculate country-level aggregate student achievement test scores across the various assessment waves by applying the unweighted mean of the non-standardized scores rather than standardizing techniques, as suggested in Rindermann (2007).<sup>5</sup> The latter assumption is uncritical, as prior studies indicate that the variation in ISAT scores across the different assessment waves is driven mainly by measurement error rather than by systematic changes in student achievement test scores (Hanushek and Woessmann, 2012; Meisenberg and Woodley, 2013). In addition, both ISATs share a similar methodological background and a common scale, thus facilitating the comparability of student achievement test score outcomes across the various assessment waves.<sup>6</sup>

We use the aggregated country-level data on PISA and TIMSS student assessment test scores following a simple decision rule. If available, we give preference to PISA student achievement test scores over TIMSS, mostly due to the conceptual differences explained above. Accordingly, PISA scores in the domains of science and mathematics are used. As stated above, PISA reading scores, although available, are disregarded in the calculation of country-level aggregated student achievement test scores to ensure compatibility to the corresponding TIMSS data, as the latter lack observations on reading score outcomes in eighth-grade students. We employ the unweighted average of both subjects in the calculation of the country-level aggregated student achievement test scores to arrive at a broad indicator of cognitive skills in the general population.<sup>7</sup> In cases where PISA data are missing, the value of the corresponding TIMSS score is assigned.<sup>8</sup>

<sup>&</sup>lt;sup>5</sup>In contrast to the proposed non-standardized approach, Hanushek and Woessmann (2008, 2012) adjust student achievement test scores in terms of level and variation to attain a score that is comparable across both countries and years.

<sup>&</sup>lt;sup>6</sup>TIMSS and PISA use item-response theory to calculate plausible values for individual student achievement. Moreover, the study-level student achievement test data are calibrated to have a mean value of 500 and a standard deviation of 100 test score points across the sampled countries.

<sup>&</sup>lt;sup>7</sup>In the sensitivity analysis, we show that the main findings are not driven by possible differences in characteristics of school curricula in the individual domains of mathematics and science.

 $<sup>^8</sup>$ To ensure that the main results are not driven by this missing data replacement strategy, we include a recodification indicator for the replacements made in all model specifications to capture potential systematic differences between the two ISATs. In the empirical analysis covering 81 countries, we employ aggregated PISA student achievement test score data in 63 cases. In the remaining 18 cases, we rely on scores from TIMSS. It is worth mentioning that we find a high bivariate correlation coefficient (Corr = 0.9066, Number of countries = 43) between values for the two ISATs, suggesting that both tests capture general cognitive abilities in society in their own ways (Hanushek and Woessmann,

#### 3.2 Main Explanatory Variable: DRD4 Exon III 2- and 7-Repeat Allele Frequency

Twin studies suggest that ADHD is a highly heritable psychiatric disorder, with estimated heritability rates of about 76% (Faraone et al., 2005). Evidence from brain imaging studies, pharmacological medication, and animal models provide strong support for the presence of dopamine dysregulation in the brain as the main cause in the etiology of ADHD symptoms.<sup>9</sup> In line with the dopamine theory of ADHD proposed by Levy (1991), initial candidate gene studies have identified a significant association of two important dopaminergic genes – the dopamine transporter gene DAT1 and the dopamine receptor gene DRD4 exon III – with ADHD (Cook et al., 1995; LaHoste et al., 1996). Although other candidate genes related to biological neurotransmission have been assessed in association with ADHD (e.g., DRD5, SLC6A4, and MAO-A)<sup>10</sup>, DRD4 exon III polymorphism have been the candidate gene studied most extensively in case-control studies of ADHD, with a preponderance of replication studies robustly confirming the initial finding in LaHoste et al. (1996). Several meta-analytical reviews strongly confirmed the positive association between DRD4 exon III polymorphism and ADHD, indicating the importance of the human dopaminergic system for human behavioral outcomes related to locomotion, emotion, reward and cognition (Faraone et al., 2005; Bobb et al., 2006; Gizer et al., 2009; Wu et al., 2012).

The human dopamine receptor D4 gene (DRD4) consists of four encoded regions – called exons in molecular genetics – of which the third region is highly polymorphic. The primary allelic variants found in the human genome occur as a 48-base pair (48-bp) variable number of tandem repeats (VNTR) defined by the 2-repeat, 4-repeat, and 7-repeat alleles that exhibit differences in physiological functioning to dopamine releases between synaptic clefts in the brain (Van Tol et al., 1991, 1992; Lichter et al., 1993). It has been shown that the 7-repeat allele variant shows a blunted response to elevated dopamine levels relative to the ancestral 4-repeat variant. The 2-repeat allele also shows a suboptimal response to dopamine release but one that is less blunted compared to the 7-repeat variant (Asghari et al., 1995; Wang et al., 2004). It has been hypothesized that the blunted response to dopamine levels among the different DRD4 exon III allele variants might result in dopamine dysregulation between synaptic clefts in the human brain that eventually results in neuropsychological disorders such as ADHD (Swanson et al., 2007).

Although the genetic architecture of complex behavioral traits like ADHD is not fully understood, specific biomarkers might provide a reliable way to identify clinical symptoms and/or personality traits related to ADHD (Faraone and Bonvicini, 2014). The DRD4 gene is a promising candidate for this endeavor, since animal models suggest that it is mainly present in brain regions responsible for attention and response inhibition (Noaín et al., 2006). In addition, pharmacological studies suggest that DRD4 exon III 7-repeat carriers need higher doses of psychostimulant drugs such as methylphenidate compared to probands with one or two forms of the DRD4 exon III 4-repeat allele variant (McGough, 2012; Bruxel et al., 2014). This result is consistent with the aforementioned finding that the 7-repeat allele variant has a suboptimal blunted response to elevated dopamine signals compared to the ancestral 4-repeat variant.

Thus, the present study employs the DRD4 exon III 2- and 7-repeat allele variants as a strong biomarker to measure the prevalence of inattention and/or hyperactivity-impulsivity as symptoms of ADHD in society. The inclusion of the 2-repeat variant is motivated by the fact that it shows a similar blunted response to dopamine release in synaptic clefts.

<sup>2012).</sup> 

<sup>&</sup>lt;sup>9</sup>See Swanson et al. (2007) for a comprehensive review of the relevant molecular genetics literature.

<sup>&</sup>lt;sup>10</sup>See Hawi et al. (2015) for a recent discussion of the set of genes that are candidates for the expression of the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity.

This issue is of particular importance among Asian populations that show very low frequencies of the 7-repeat variant but higher frequencies of the 2-repeat allele (Gören, 2016). The grouping of the DRD4 exon III 2- and 7-repeat allele variants is further supported by evidence from a candidate gene study among Chinese ADHD probands, indicating that ADHD was associated with a higher frequency of the 2-repeat variant that is comparable in magnitude to the 7-repeat frequency usually found among ADHD probands of European ancestry (Leung et al., 2005).

We use data on the prevalence of DRD4 exon III 2- and 7-repeat allele frequencies across a large number of countries. These data were constructed in Gören (2017a) by linking the Alesina et al. (2003) ethnicity data to the population genome data in Gören (2016) using information on language classifications from the *Ethnologue* database (Global Mapping International, 2010). Based on the findings from molecular genetics, we hypothesize that a higher prevalence of DRD4 exon III 2- and 7-repeat allele frequency in society is associated with a higher rate of ADHD-related behavioral symptoms.

## 3.3 Additional Control Variables: Biogeographic, Historical, Health-Related, Socioeconomic, Educational, Genetic, and Diversity Factors

Biogeographic Factors. We employ a basic set of biogeographic and climatic controls to examine, among other things, the impact of the soil quality, disease environment, and temperature on the accumulation of human capital in society. In order to assess the impact of thermal stress on cognitive abilities, we use the country's mean temperature value during the period 1960 to 1990. Investigations on paired associate learning shows that cognitive abilities are influenced by ambient temperature (Allen and Fischer, 1978), indicating that student test performance could depend on the location and average temperature of the host country. Additionally, thermal stress due to high temperatures inside buildings is found to negatively affect student achievement by decreasing cognitive performance (Mendell and Heath, 2005; Hancock and Vasmatzidis, 2003; Amasuomo and Amasuomo, 2016).

We use the percentage of inhabitants at risk of contracting malaria to assess the prevailing conditions of the disease environment and its possible implications for cognitive achievement outcomes. Despite showing a rather low case-fatality rate, malaria prevalence is described as having severe symptoms and persistent effects on cognitive development and functioning, either directly through anemia or indirectly through lost time in education due to absenteeism from school (Bleakley, 2010; Gallup and Sachs, 2001). Malaria-related school absenteeism has been found to be an important predictor of increased grade repetition rates, school failure, higher school drop-out rates, and lifelong cognitive impairments (Sachs and Malaney, 2002; Gallup and Sachs, 2001). The eradication of parasite-borne diseases has been found to be associated with significant increases in returns to schooling by raising educational quality through reduced absenteeism from school (Bleakley, 2007).

Furthermore, we consider the percentage of a country's arable land area to account for the importance of the agricultural sector in the overall economy. We hypothesize that a larger share of the workforce engaged in the primary sector is generally associated with low-skilled occupational activities (e.g., cultivation of primary crops) which might require lower investments in the accumulation of human capital. The underlying mechanisms might be comparable to the resource curse hypothesis in development economics (Sachs and Warner, 2001; Easterly and Levine, 2003), in which excessive resource extraction might result in pervasive rent-seeking activities that negatively affect the accumulation of human capital in society (Sachs and Warner, 2001; Easterly and Levine, 2003). Additional studies indicate that a larger share of

agricultural workers in an economy is correlated with reduced overall cognitive performance and a lack of incentives for educational investment. It has been hypothesized that a low cultural value of and lack of esteem for education are the main driving force behind this negative relationship (Barber, 2005; Rindermann, 2007).

Historical Factors. We hypothesize that education is affected by the kind of values, norms, and beliefs promoted by religious groups. Historically, Protestant and Catholic religious leaders have promoted and discouraged a variety of socioeconomic outcomes, such as political and civic engagement but also literacy (Putnam et al., 1993; Landes, 1998; Becker and Woessmann, 2009). In addition, religiosity has been shown to affect knowledge creation, especially through bans on the teaching of specific bodies of scientific thought (e.g., evolutionary theory in public schools) and through time devoted to the practice of religion (Stoet and Geary, 2017).

Furthermore, we examine the effect of legal traditions on the formation of cognitive skills. We assume that cultural, political, and economic conditions, but also the historical development of institutions, are reflected in a country's national legal and regulatory systems (La Porta et al., 1999). The effect of legal origins on present-day economic performance is described to operate mainly through the transmission channel of education (Rostowski and Stacescu, 2006), whereby the broad ideas underlying common and civic law have influenced not only the formulation of specific rules but also the formation of human capital. Thus, the spreading of legal traditions has also affected the acquisition of education through slow-changing expectations, beliefs, modes of thought, and ideologies (La Porta et al., 2008). For instance, the French legal tradition has been described as detrimental to educational policies due to interventions into school curricula designed to promote French language and culture and to repress "deviant" ideas that might undermine the legitimacy of the central government in society (La Porta et al., 2008). In addition, socialist legal traditions have been found to significantly affect the level of and preference for education in society through repression and a lack of individual vocational freedom (Alesina and Fuchs-Schuendeln, 2007; Fuchs-Schuendeln and Masella, 2016).

Health-Related Factors. As suggested by the literature on innate abilities, in utero experiences, such as overall maternal health, substance abuse (e.g., alcohol, tobacco, medications, and drugs) or insufficient provision of micro-nutrients affect prenatal health and children's physical and cognitive development. Child birth weight is considered a widespread measure of perinatal health, which, for full-term birth, substantially reflects the state of prenatal development. Low birth weight (LBW) is typically defined as infant birth weight being lower than 2,500 grams. However, this definition does not explicitly enable a comparable measurement of reasons for LBW, which can either be caused by intrauterine growth restriction, i.e., low weight for gestational age (Black et al., 2013) or by an interruption of the normal intrauterine growth cycle, i.e., through preterm birth (Goldenberg et al., 2000, 2008). Despite these and further conceptual difficulties associated with LBW as an indicator of perinatal health LBW has proven to be the most frequently used and readily available measure which provides valuable information about environmental and in utero influences during gestation and possesses a high predictive power for later health outcomes and cognitive development (Bhutta et al., 2002; Gluckman et al., 2008; Barker, 1995).

<sup>&</sup>lt;sup>11</sup>See diagnosis code P07 of the World Health Organization ICD-10 criteria, which lists disorders of newborns related to short gestation and low birth weight.

<sup>&</sup>lt;sup>12</sup>See Wilcox (2001) for a more detailed discussion of this issue.

Another control variable linked to the quality of the country's health system is the infant mortality rate, which is largely determined by the supply of basic medication, sanitation, and nutrition in society. A large body of literature is dedicated to the impacts of malnutrition on children's mental and physical development. Stunting and wasting, as well as lack of micro-nutrients such as iodine, zinc, or iron in early childhood are found to have dramatic effects on brain development and thereby on later cognitive abilities. Deficiency of nutrients, nutrition, and medication in early childhood are observed to cause sickness-related impairments and the failure to reach the individual's full developmental potential, resulting in reduced performance in cognitive and IQ tests, lower educational attainment, poor labor market outcomes, and the emergence of inattentive and hyperactive behavior (Grantham-McGregor et al., 2007; Chang et al., 2002; Walker et al., 2007). It has been reported that an insufficient supply of medication, nutrition, and sanitation accounts for up to 60% of early infant deaths in developing countries (Black et al., 2008; Tate et al., 2012; Kotloff et al., 2013; Esrey et al., 1991). Accordingly, we expect the infant mortality rate to act as a heavy burden on educational achievement by way of mortality, disease, disabilities, and impaired mental and physical development.

Using the same rationale, the longevity of the adult population, measured as the life expectancy at birth, is in large part determined by a country's health system and environmental health risks (Pritchett and Summers, 1996). Empirical studies and theoretical models suggest that, regardless of the educational environment, a higher life expectancy leads to increases in schooling demand, as individuals are able to draw benefits from their returns to skills for a longer working period due to a longer healthy life (Ben-Porath, 1967; Bils and Klenow, 2000; Strulik and Werner, 2016). Conversely, the expectation of early mortality reduces the incentive for educational investments and instead fosters unsustainable, reckless behavior (Lorentzen et al., 2008).

Similarly, disability-adjusted life years (DALY) lost provide a measure of the working power lost due to diseases, disabilities, and death and therefore serve as an extension to the life expectancy at birth variable. In contrast to life expectancy, educational decisions are affected not only by considerations of longevity but also by mental and physical health. Apart from that, the same mechanisms mentioned above are at work here, too, suggesting that in this case, a lower number of DALY implies increased incentives for investing in education.

Last, we consider the country's homicide rate (i.e., the number of intentional homicides per 100,000 people) as an additional health-related factor in the analysis of cognitive achievement outcomes. We hypothesize that an everyday experience of violent crime creates an unfavorable environment for educational investment due to uncertainty, and discourages learning and school effort. A higher incidence of crime is linked to distraction and the creation of criminal opportunities, with the consequence that efforts directed at socially beneficial and productive activities is reduced (Pritchett, 2001). In general, a negative relationship has been observed between violent crime and the level of education (Lochner and Moretti, 2004; Lochner, 2004). Studies show that witnessing violent crime or intentional killings significantly reduces cognitive ability and school attendance of children, as a result of stress, trauma, and fear (Brown and Velásquez, 2017; Sharkey, 2010).

**Socioeconomic Factors.** In this section, we discuss a set of economic environment and social value factors that might significantly predict observed differences in international student achievement outcomes.

To start with the set of economic environment factors, it has been reported that cross-country income differences are significantly correlated with differences in scholastic achievement. Richer countries tend to perform better in ISATs, sug-

gesting that the level of economic development plays a role in determining educational achievement outcomes (Hanushek and Woessmann, 2012). The gross domestic income per person employed serves as a broad proxy for the overall standard of living in society. Additionally, economic development is closely related to infrastructure and institutional quality, suggesting that an economically more developed country might have an advantage in the accumulation of human capital. Next, we hypothesize that both the country's total fertility rate as well as the growth rate of the total population could affect cognitive ability outcomes, as they portray the familiar quality versus quantity trade-off in child rearing (Becker and Lewis, 1973). A higher quantity of children implies that family inputs into the production process (e.g., time spent on child rearing, tuition fees, and tutoring costs) have to be distributed across a large number of children, leading to reduced investment per child and, thus, to subsequently reduced educational achievement (Cunha et al., 2006).<sup>13</sup>

We further consider the share of government consumption to approximate the size of the country's government apparatus and its possible implications for the accumulation of human capital in society. On the one hand, a higher government consumption share is assumed to be an indicator of the country's fiscal capacity, which enables the provision of key public goods and services such as schools, libraries, and Internet access. On the other hand, these benefits are to some extent counterbalanced by possible growth-distorting government taxes and increased centralization, which have been found to significantly reduce student achievement outcomes (Falch and Fischer, 2008).

Another aspect of the economic environment in which people coexist in society is the country's degree of openness to trade. The argumentation is that countries more open to trade, and thus to international competition, might invest more in the accumulation of cognitive skills that are complementary to technological progress (Jamison et al., 2007).

In the remaining part of this section, we provide a discussion of the set of social value factors that might explain observed differences in international student achievement outcomes.

First, we consider a set of gender equality controls to account for the fact that a gender-biased cultural environment in society might have large and detrimental effects on students' test performance outcomes (Guiso et al., 2008; Fryer and Levitt, 2010). If girls are discouraged from making educational investments, their overall cognitive abilities are expected to decrease. Accordingly, a higher degree of women's emancipation has been found to be accompanied by a smaller achievement gap between boys and girls in mathematics test score outcomes (Guiso et al., 2008). To control for the role of women in the society, the share of females in the total labor force is used to proxy for the overall participation and presence of women in everyday work life. If female graduates have no access to the labor market, the main incentive for the acquisition of cognitive skills is reduced. This also applies in the case that certain occupational positions are not accessible to the women in a society. Specifically, we use the proportion of women in national parliaments as an indicator for this phenomenon of women not being able to reach higher positions, which has become known as the glass ceiling effect (Cotter et al., 2001).

Second, we use a set of institutional environment controls in the analysis of international aggregate student achievement outcomes. Human capital investments can turn out to be inefficient when they are not complemented by favorable political institutions. Without the provision of basic institutional structures, education might be misused for personal gain (e.g., bribery or unproductive rent-seeking activities) instead of for socially beneficial work (Pritchett, 2001).

<sup>&</sup>lt;sup>13</sup>Given the observed positive association between mortality and fertility, a country's higher total fertility rate might to some extent also reflect the parent's precautionary demand for children as a means of compensating for a higher risk of child mortality in society (Lorentzen et al., 2008).

For example, it has been shown that higher decentralization improves the provision of schooling services that contribute significantly to student test score outcomes (Falch and Fischer, 2012). Thus, we hypothesize that countries with a federalist system of government might perform better in terms of student achievement test outcomes in comparison to countries with a unitary or centralized political system.<sup>14</sup>

The early human capital literature found the property rights of workers over their own skills to be a fundamental incentive for investments in training (Becker, 1975). Furthermore, the protection of intellectual property rights guarantees inventors the ability to reap the benefits from their inventions and thus constitutes a prerequisite for innovation activity in the economy (Helpman, 1993). We hypothesize that the establishment of property rights in a country's legislation and the enforcement thereof by the government are crucial for educational investment decisions. Planning certainty is important for balancing short-term costs (e.g., tuition fees and foregone earnings) against long-term returns to education (e.g., increased wages and profits from patents).

Freedom from corruption provides a measure of government integrity, which could affect the institutional framework of the educational system. Despite the educational sector being described as insusceptible to bribery, educational expenditures are found to be negatively affected by the presence of corruption (Mauro, 1998). Furthermore, the presence of corruption is found to significantly reduce school enrollment and student learning (Reinikka and Svensson, 2005). It has been observed that corruption is prevalent in countries with lower average rates of higher education, where high-skilled individuals are often primarily engaged in bribe-collection, perhaps due to the lack of economic opportunities (Murphy et al., 1991; Glaeser and Saks, 2006; Mocan, 2008).

Finally, we employ the number of civil conflicts to assess the detrimental influence of political instability on the accumulation of human capital in society. We hypothesize that the incidence of conflicts affects cognitive abilities through three main transmission channels. First, civil conflicts induce uncertainty due to political changes and threats against property rights and might thereby reduce individual incentives for educational investment (Ichino and Winter-Ebmer, 2004). Second, it has been reported that an environment of unrest, civil conflict, or war negatively affects the development of socio-emotional and cognitive abilities by causing stress, anxiety, and trauma (Wolff et al., 1995). Finally, the incidence of civil conflicts might divert both private and public resources from the educational sector to covering the costs of warfare, collateral damage, and scarcity of resources (Shemyakina, 2011).

Educational Factors. Previous studies suggest that international differences in access to basic schooling services or school participation rates might affect the aggregate level of cognitive ability outcomes in society (Rindermann, 2007). Besides being an important determinant in the analysis of aggregate student achievement outcomes, the inclusion of basic measures of educational attainment (i.e., the share of a country's population aged 15 and above with primary, secondary or no education) in the empirical analysis further accounts for the issue of sample selection bias in the variation of student test score outcomes across countries. Specifically, even though PISA and TIMSS are designed to be representative of the respective student populations, it is conceivable that countries with lower school enrollment rates and thus lower overall

<sup>&</sup>lt;sup>14</sup>It is worth mentioning that a higher degree of decentralization might also proxy for the autonomy in educational policy decisions (e.g., school autonomy in important areas of decision-making such as budget allocation and school curricula). The interested reader is referred to Hanushek and Woessmann (2011) for a discussion of the relevant literature on the institutional structures of the educational system and possible implications for student test score outcomes.

basic educational attainment might select a non-representative sample of high-performing students with a privileged socioeconomic status. Thus, if variations in student test score outcomes were indeed subject to sample selection bias, we would expect a negative regression coefficient associated with the various educational attainment controls (Hanushek and Kimko, 2000).<sup>15</sup>

In order to control for the effect of formal schooling quantity on educational achievement, we consider a compound human capital index in the analysis of aggregate student test score outcomes. We assume schooling quantity to have a twofold effect on scholastic achievement. On the one hand, an increased duration of education results in a larger overall quantity of learning and implies a higher level of education, more sophisticated subject matter, as suggested by the returns-to-schooling literature, for instance, Psacharopoulos (1994). This ultimately translates into gains in cognitive abilities (Kaarsen, 2014). Therefore, more schooling should proxy for both an increased depth and breadth of education. On the other hand, average years of schooling reflect the importance of education in an economy. A high average number of schooling years is related to a high-skilled labor force and indicates a certain pressure to achieve and the relevance of particular high-skilled cognitive abilities in the production process (Rindermann, 2007). Additionally, the overall level of education in society might reflect the knowledge stock of the parent (or teacher) generation and affect the formation of cognitive skills as a direct school or family inputs.

As a further, frequently discussed input into the educational production function, we employ the student/teacher ratio (STR) as an indicator of average class size. A small class size is widely accepted as beneficial for student achievement, as it increases the amount of time a teacher can spend meeting the needs of a possibly heterogeneous group of students (Ehrenberg et al., 2001; Wright et al., 1997).<sup>16</sup>

In many countries, educational services are provided either by state-run public schools or by independent private schools. To assess the implications of this competition between publicly and privately provided education in a country for cognitive achievement outcomes, we consider the country's share of private enrollment in primary education. This variable serves as an indicator of individuals or parents choosing privately operated educational facilities over public institutions (Couch et al., 1993; Hsieh and Urquiola, 2006). Private schools could be chosen over public schools because private provision of educational services is more efficient, implying better individual learning outcomes (e.g., higher qualifications of teachers, better classroom equipment, and curricular differences) and higher social prestige (e.g., of graduates and academic rankings). Individuals may therefore expect a higher market return to their educational investment when after graduation from private schools (Goldhaber, 1996; De La Croix and Doepke, 2009). A second reason to choose private schools is that they may act as a substitute where publicly provided educational services are insufficient. Such choices could be driven either by the low quality of public schools (e.g., poor equipment, few or low-skilled teachers) or by a lack of school provision (e.g., no public schools in rural areas or remote districts) (Couch et al., 1993; Rouse, 1998; Chaudhary et al., 2012).

In order to assess the effect of financial inputs in the formation of cognitive ability outcomes, we consider the country's educational expenditures, measured as a share of total GDP. We hypothesize that a higher share of educational expen-

<sup>&</sup>lt;sup>15</sup>This finding would be consistent with the hypothesis that aggregate student achievement test score outcomes in countries with low educational attainment might reflect the cognitive abilities of a small, elite group rather than the general population.

<sup>&</sup>lt;sup>16</sup>This variable is usually perceived as a natural instrument of educational policy, mainly achieved through the hiring of more teachers (Hoxby, 2000).

ditures expresses the governmental valuation and priority of the provision of public education in society. Furthermore, additional spending on educational facilities increases the utilization of government resources for the accumulation of cognitive skills. Increased educational expenditures could entail hiring more teachers (resulting in smaller class sizes) and better quality teachers (paid higher salaries), improved classroom equipment (e.g., supplies of books and computers), reduction of tuition fees, or simply more schools (i.e., nationwide provision of educational services) available to the public (Hanushek, 1997). Through these improved learning facilities and environments, an increase in student achievement is expected.

Genetic Factors. Although it is widely acknowledged that environmental and social factors predict future cognitive performance and educational attainment, estimates from twin studies suggest that up to 58% of the variation of educational achievement is highly heritable (Shakeshaft et al., 2013). Specifically, a number of Genome-Wide Association Studies (GWAS) identified various independent genome-wide loci (typically differences in single-nucleotide polymorphisms – i.e., SNPs) that significantly predict an individual's educational attainment measured as the number of completed years of schooling (Rietveld et al., 2013; Okbay et al., 2016). Genome-wide polygenic score (GPS) analysis has been used to assess the predictive power of the relevant SNPs for various measures of educational attainment. GPS aggregate the effect of a large number of particular SNPs that are ranked according to the estimated effect sizes. For example, the estimated GPS of the most recent GWAS of educational attainment account for up to 4% of the total variance in completed years of education (Okbay et al., 2016). The predictive power of GPS of schooling years will rise further as the availability of larger sample sizes identifies additional genetic traits associated with educational attainment (Chabris et al., 2015). It is worth mentioning that derived GPS of education years correlates significantly with various measures of educational achievement in independent samples, including reading and mathematics ability and general cognitive ability (Selzam et al., 2017a,b). Furthermore, gene-based analysis has identified several genes expressed in the human brain that significantly correlate with human intelligence (Sniekers et al., 2017).

Overall, the above studies point to the importance of genetic factors for observed differences in educational and cognitive performance across individuals. Since there is a lack of data on genome-wide polygenic scores of educational years at the aggregated country level, we instead employ different measures of genetic diversity to account for the heritable nature of student achievement test score outcomes. It is worth noting that the use of various genetic diversity measures implicitly assumes that variation in student achievement test scores across countries is to some extent rooted in genetic differences across people in society. Observed differences in genetic traits across countries might either enhance or suppress the country's cognitive performance. However, theoretical and empirical models suggest that genetic diversity is associated with a larger continuum of specific human traits that might facilitate knowledge creation in society (Ashraf and Galor, 2013).

Besides using an overall measure of genetic diversity, we also consider a specific set of genetic markers that were subject to strong selection pressure in prehistoric times. In particular, we examine a possible association between genes located in the human leukocyte antigen (HLA) region and student achievement test performance. The HLA region is a cluster of more than 200 genes located on chromosome 6 that consists of three important sub-regions, referred to as class I, class II, and class III (Shiina et al., 2004). The HLA region is one of the most polymorphic regions found in the human genome, with genetic variation in the class I and class II regions providing inherent resistance to many infectious diseases.

Genetic variability in the HLA region supports the individual's immune system in the identification and elimination of intracellular pathogens. Among the possible natural selection models that could explain the extraordinary polymorphism within the HLA region, the heterozygote advantage (or overdominant selection) hypothesis has received special attention. According to this theory, heterozygote populations would more efficiently resist an array of pathogens due to a larger range of possible antigens associated with HLA heterozygosity (Doherty and Zinkernagel, 1975; Hughes and Yeager, 1998). Consistent with this hypothesis, Prugnolle et al. (2005) have shown that regional pathogen richness is positively and significantly correlated with HLA class I diversity in a worldwide sample of 61 human populations.<sup>17</sup> Further evidence has been provided by Cook (2015), who found a significant and positive association between HLA diversity and life expectancy prior to the international epidemiological transition in a large cross-section of countries.

In addition to health-related effects, HLA diversity has been found to be associated with language impairment, decreased cognitive ability in later life, susceptibility to psychiatric disorders (e.g., schizophrenia), and high intelligence in independent samples, even though the latter finding lacks significant evidence (Nudel et al., 2014; Payton et al., 2015; Brucato et al., 2015; Zabaneh et al., 2016). As argued above, health-related factors have a strong and significant effect on general cognitive ability, and the inclusion of HLA diversity in the empirical analysis accounts for the innate resistance of populations to many disorders. We use the country-level estimates of HLA diversity from Cook (2015) to examine its association with student achievement test scores.

Diversity Factors. Theoretical models suggest that members of diverse teams may outperform members of homogenous ones because of enhanced problem-solving capabilities (Hong and Page, 2001, 2004). On the other hand, a large number of studies in development economics point out that ethno-linguistic diversity is associated with poor socioeconomic outcomes, such as low economic growth, low provision of public goods, and a higher incidence of conflicts (Alesina et al., 2003; Desmet et al., 2012). In relation to the issue of human capital accumulation, it has been argued that educational achievement might be lower in ethnically diverse countries due to higher disagreement on issues such as school curricula and language of instruction (Easterly and Levine, 1997; Gören, 2014).

Besides being an important explanatory variable in the analysis of general cognitive ability, the inclusion of ethno-linguistic diversity further accounts for a possible source of confounding in the relationship between the key genetic controls (i.e., the DRD4 exon III 2- and 7-repeat allele frequency measure, overall genetic diversity, and HLA diversity) and cognitive achievement. For example, Ashraf and Galor (2013) argue that genetic diversity is also correlated with cultural diversity (usually measured as ethno-linguistic and/or religious diversity). More precisely, the association between the country-level  $DRD4^{R2R7}$  measure and student achievement test score outcomes might simply capture the negative effect of ethno-linguistic diversity on cognitive performance rather than the issue of ADHD prevalence. To rule out this possibility, we examine the robustness of the main findings to the inclusion of various measures of ethno-linguistic and religious diversity, using the relevant country-level estimates provided by Alesina et al. (2003).

<sup>&</sup>lt;sup>17</sup>Other studies reported the protective effect of HLA diversity against specific human infectious diseases, such as malaria, hepatitis B, and HIV (Hill et al., 1991; Thursz et al., 1997; Trachtenberg et al., 2003).

#### 4 Relationship Between DRD4 Exon III and ADHD Prevalence

The aim of this section is twofold. In a first exercise, we establish the key hypothesis proposed in this paper about the strong explanatory power of the DRD4 exon III 2- and 7-repeat allele frequency measure of inattention and/or hyperactivity-impulsivity as behavioral symptoms of ADHD, as indicated by aggregated ADHD prevalence rates derived from individual ADHD studies. Second, we argue against the use of estimated ADHD prevalence rates in the empirical analysis. The reason is that these prevalence rates are heavily confounded by the issue of study-level methodological characteristics, as evidenced by a systematic meta-regression analysis of 179 individual ADHD studies conducted in 52 countries in the period 1981 to 2014. We discuss these and additional caveats associated with the use of estimated ADHD prevalence rates derived from individual ADHD studies.

### 4.1 Reliability of ADHD Prevalence Estimates: The Issue of Study Methodological Characteristics

ADHD prevalence estimates vary significantly across studies both between and within countries, fueling concerns about possible over- and under-diagnosis of reported ADHD cases in the general population (Charach et al., 2011; Fulton et al., 2009). A series of meta-regression studies identified that ADHD prevalence estimates are explained largely by methodological characteristics of the study, including ADHD diagnostic criteria, source of information on ADHD behavioral symptoms (i.e., child, parent, teacher, or clinician), type of population sample (i.e., community, school, or whole population), whether full or partial diagnostic criteria for ADHD were met, type of ADHD measurement (i.e., diagnostic report, interview, or symptom-only checklist), and the geographical region of the study (i.e., America, Asia, Europe, Africa, or Oceania), casting serious doubts on the reliability of reported ADHD cases across studies (Polanczyk et al., 2007, 2014; Thomas et al., 2015).

In the following, we conduct a systematic meta-regression analysis to examine the severity of study-level characteristics on estimated ADHD prevalence rates. This step of the analysis would provide information on the reliability of reported ADHD cases across studies and whether these estimates can be used to provide a valid measure of ADHD prevalence in the general population. To accomplish this task, we use a worldwide compilation of ADHD prevalence rates from 179 individual studies conducted in 52 countries during the period 1981 to 2014 (Thomas et al., 2015). The review includes studies reporting point ADHD prevalence estimates with ADHD cases identified based on the various editions of the DSM diagnostic criteria from community, schooling, or national representative survey samples. Studies reporting ADHD cases from healthy adults (i.e., age > 18) were excluded as well as studies reporting clinical intervention samples. Specifically, we estimate the following weighted meta-regression model to assess the confounding influence of study-level methodological characteristics

$$ADHD_{scrt}\sqrt{\omega_{scrt}} + \beta_1'\Gamma_{scrt}\sqrt{\omega_{scrt}} + \beta_2'\Delta_{scrt}\sqrt{\omega_{scrt}} + \beta_3'\Theta_{scrt}\sqrt{\omega_{scrt}} + \beta_4CM_{scrt}\sqrt{\omega_{scrt}} + \beta_5'\Delta_{scrt}\sqrt{\omega_{scrt}} + \beta_6'\Phi_{scrt}\sqrt{\omega_{scrt}} + \lambda_r\sqrt{\omega_{scrt}} + \lambda_t\sqrt{\omega_{scrt}} + v_{scrt}\sqrt{\omega_{scrt}},$$

$$(1)$$

where  $ADHD_{scrt}$  refers to the ADHD prevalence rate of study s conducted in country c within region r at time period t. Although ADHD outcomes vary at the individual level, observations on reported cases of ADHD and control variables are only available at the aggregated study level. This requires study-level observations to be weighted according to the

reported sample size  $\omega_{scrt}$ , thus accounting for the possibility that the validity of reported cases of ADHD might be an increasing function of sample size. Estimating Equation (1) on study-level rather than individual observations accounts for the unexplained variation in reported cases of ADHD resulting from methodological characteristics of the study rather than individual characteristics.

We consider a full set of possible confounding methodological factors that might explain reported cases of ADHD across studies. The vector  $\Gamma$  includes a set of control variables indicating a possible Risk of Bias (RoB) in ADHD prevalence estimates. Specifically, each study was assessed based on eight RoB questionnaires reflecting the representativeness of sample, the sampling frame, whether individuals were randomly selected, minimum non-response bias, whether observations were directly collected from subjects, and whether data collection and construction fulfills a minimum level of measurement reliability and validity. The calculated RoB score ranges from 0 (= very high risk of bias) to 8 (= low risk of bias score). We classified studies having a very high RoB score (i.e.,  $0 \le RoB \le 1$ ), high RoB score (i.e.,  $2 \le RoB \le 3$ ), moderate RoB score (i.e.,  $4 \le RoB \le 5$ ), and low RoB score (i.e.,  $6 \le RoB \le 8$ ).

The vector  $\Delta$  includes a set of indicator variables that reports the various editions of the DSM diagnostic criteria (i.e., DSM-III, DSM-III-R, and DSM-IV) used to identify ADHD cases in the sample,  $\Theta$  refers to a set of control variables indicating the source of sampled population (i.e., schooling, community, or national representative survey sample), the variable CM indicates whether the study employs the full diagnostic criteria of the various editions of the DSM to identify ADHD cases, and  $\Lambda$  indicates the type of ADHD measurement (i.e., diagnostic report, interview, or symptom-only checklist).

In addition, we examine the source of information or informant (e.g., child, parent, teacher, or clinician) reporting ADHD-related symptoms, as summarized in the vector  $\mathbf{\Phi}$ . Most studies employ more than one informant to identify ADHD cases in the sample. We follow the relevant literature and classify studies according to an 'AND' rule that is positive if the reported ADHD case is confirmed by at least two informants (e.g., by parents and teachers) and according to an 'OR' rule that is positive if confirmed by any one informant (e.g., parent or teacher). In cases where the ADHD diagnostic assessment was supervised by clinical personnel, the study's source of information was completely coded as clinician.

The regression model further includes a full set of region fixed effects  $\lambda_r$  (i.e., indicator variables for North America, South America and the Caribbean, Africa, Asia, Middle East, and Europe) to account for differences in ADHD prevalence rates across regions and a set of publication year fixed effects  $\lambda_t$  (i.e., indicator variables for the years 1980 – 1984, 1985 – 1989,..., 2010 – 2014) to account for a possible increase in ADHD prevalence rates across time. Finally,  $v_{scrt}$  is a study-specific error term. The method of estimation is Ordinary Least Squares (OLS) with standard errors clustered by countries to account for the likely non-independence of reported ADHD prevalence estimates within countries.

Table 1 reports the results on the association between ADHD prevalence estimates and study-level methodological characteristics. The estimates reported in column (1) examine the influence of the various RoB indicators on estimated ADHD rates. All three indicators are significantly and negatively (reference category: low RoB score) related to the study's estimated ADHD prevalence rate, with an overall  $R^2$  of about 19%. The estimated coefficients suggest that studies lacking various criteria on, for example, representativeness, random sampling, or measurement reliability and validity tend to have lower reported ADHD prevalence estimates compared to studies with a low risk of bias assessment.

In column (2), we examine the possible confounding influence of the various editions of the DSM diagnostic manual used to identify ADHD-related symptoms in individuals. Specifically, it has been reported that estimated ADHD prevalence

rates may differ substantially even within the same study, with estimated prevalence rates being higher when using DSM-IV diagnostic criteria in comparison to DSM-III-R caused by significant heterogeneity in the definition of ADHD-related subtypes (Baumgaertel et al., 1995; Wolraich et al., 1996). Consistent with this observation, the estimated results show that studies using DSM-IV diagnostic criteria report about 2.2731% higher ADHD prevalence rates than studies based on the ADHD symptom checklists in the DSM-III-R diagnostic manual.

The results presented in column (3) suggest that studies using community or nationally representative survey samples have significantly higher ADHD prevalence estimates on average relative to studies employing school samples, even though the reported association is sensitive to subsequent model specifications.

Moreover, ADHD prevalence estimates might differ as to whether studies employ full or partial diagnostic criteria for ADHD-related symptoms. Hence, in column (4), we include an indicator variable that is positive if the respective study employs the full ADHD diagnostic criteria to identify ADHD cases versus a partial diagnostic assessment. Although the corresponding estimate is not statistically significant at conventional significance levels, it turns highly significant in later model specifications, confirming its predictive power for ADHD prevalence estimates across studies.

In column (5), we investigate how the type of ADHD measurement affects ADHD prevalence rates across studies. The results robustly show that studies using a symptom-only checklist have significantly higher ADHD prevalence rates in comparison to a detailed diagnostic report in the range of 4% to 6% depending on the model specification. The inclusion of the various types of measurement controls increases the  $R^2$  considerably from 27% in column (4) to about 38% in column (5), qualifying it as a key study-level confounding factor in ADHD prevalence estimates.

In addition, ADHD prevalence estimates might be sensitive to the informant (i.e., child, parent, teacher, and clinician) or the type of method used to combine information from multiple informants (i.e., 'AND' versus 'OR' rule) (Willcutt, 2012). According to the estimates in column (6), the results suggest that studies using children, parents, or a combination of children, parents, or teachers reporting ADHD symptoms tend to underestimate ADHD prevalence rates compared with studies using ADHD symptoms reported by a clinician. The importance of the source of information for estimated ADHD prevalence rates across studies is evidenced by an additional 10% increase of the  $R^2$  in column (6) relative to column (5). The previous analysis suggests that study-level methodological characteristics rather than individual factors account for approximately 48% in the variation of reported ADHD prevalence rates across studies. In the remaining two model specifications, we assess the robustness of study-level confounding factors to the presence of time and region fixed effects. In column (7), we include a full set of time fixed effects (i.e., indicator variables for each five-year period in 1980 – 1984, 1985 – 1989,..., 2010 – 2014). The inclusion of time fixed effects accounts for a possible increase in ADHD prevalence estimates caused, for example, by increased awareness of the diagnosis of ADHD-related symptoms and/or access to improved medical services for the treatment of ADHD (Polanczyk et al., 2014). Nevertheless, the results remain qualitatively unaffected by the inclusion of time fixed effects in the regression model. It is worth mentioning that the inclusion of time fixed effects further captures significant changes in DSM diagnostic criteria for ADHD over time, which would explain the insignificant coefficient of the DSM-III-R indicator variable when the regression equation includes time fixed effects. Finally, the main findings remain rather unaffected when including a full set of region fixed effects in the regression model. as shown in column (8). However, the large and significant increase of the  $R^2$  from 50% in column (7) to about 65% in column (8) provides strong evidence that ADHD prevalence estimates vary considerably across regions, even accounting for the possible confounding influence of study-level methodological characteristics.

#### 4.2 Association Between ADHD Prevalence Estimates and DRD4 Exon III 2- and 7-Repeat Allele Frequency

The previous discussion has shown that study-level methodological rather than individual characteristics contribute significantly to reported ADHD cases, thus limiting the external validity of estimated ADHD prevalence rates for the general population. Despite these obvious limitations, we nevertheless employ ADHD prevalence rates to examine a key assumption in the present study that the proposed DRD4 exon III biomarker is significantly correlated with the extent of ADHD-related symptoms in society.

To accomplish this task, ADHD prevalence rates for each country were calculated from the corresponding study-level data, with studies having a larger sample size being assigned a higher weight in the calculation. The pooled ADHD prevalence rate across the available set of 52 countries was about 0.0686 with a standard deviation of 0.0480. To provide a direct test of the positive association between ADHD prevalence rates and the DRD4 exon III 2- and 7-repeat allele frequency measure, we estimate the following regression model in a cross-section of countries:

$$ADHD_c = \beta_0 + \beta_1 DRD4_c^{R2R7} + \boldsymbol{\beta}_2' \boldsymbol{X}_c + \boldsymbol{\beta}_3' \boldsymbol{R}_c + e_c,$$
(2)

where  $ADHD_c$  refers to the ADHD prevalence rate of country c derived from study-level data,  $DRD4_c^{R2R7}$  is the country-level DRD4 exon III 2- and 7-repeat allele frequency measure, the vector  $\mathbf{X}_c$  is a vector of country-level control variables,  $\mathbf{R}_c$  refers to a full set of region fixed effects, and  $e_c$  is the usual error term. The vector  $\mathbf{X}_c$  includes a set of socioeconomic, historical, institutional, and cultural variables that might explain differences in ADHD prevalence rates across countries and to ensure that the proposed key association is not sensitive to the issue of omitted variables.

Table 2 presents OLS estimates between the ADHD prevalence rates and the DRD4 exon III 2- and 7-repeat allele frequency measure in a cross-section of 42 countries. The reduced sample size is due to missing observations of the set of country-level controls included in Equation (2). In the baseline specification presented in column (1), the regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure is positive and highly statistically significant at the 5% significance level conditional on a full set of region fixed effects. This result is consistent with findings from molecular genetic studies, emphasizing the importance of the human dopamine system – and especially a possible role of the DRD4 exon III gene – for ADHD-related behavioral symptoms. To the best of the authors' knowledge, this is the first study showing that this association even holds at the aggregated macro-level and among culturally distinct countries, qualifying the DRD4 exon III gene as a strong biomarker for the measurement of ADHD-related symptoms in the general population. In column (2), we include a set of legal origin factors (i.e., indicator variables that refer to British, French, Scandinavian, German, and Socialist legal origin) to control for the institutional environment in which people live and cooperate in a given country. We hypothesize that differences in estimated ADHD prevalence rates across countries might to some extent reflect differences in the country's legal tradition. For example, the collectivist environment in socialist countries might result in a higher awareness of antisocial behavioral outcomes of ADHD-impaired individuals in comparison to countries with a British legal tradition that is relatively tolerant of individualistic self-expression. Reassuringly, the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure remains rather robust and precisely estimated at the 1% significance level.

In column (3), we address potential concerns in the relevant psychiatric literature arguing that variation of ADHD prevalence rates across countries might reflect differences in cultural values, norms, and beliefs regarding the recognition and

treatment of ADHD related behavioral symptoms in society (Timimi and Taylor, 2004; Singh, 2008). We use a set of major religion shares (i.e., percentage of the population that is Protestant, Muslim, or Catholic) to measure differences in cultural values across countries. Again, the association between ADHD prevalence rates and the country-level  $DRD4^{R2R7}$  measure remains precisely estimated even in this model specification.

Next, we consider another source of confounding factors caused by differences in standards of living across countries. Specifically, it is conceivable that estimated ADHD prevalence rates are underreported in countries that are less developed economically and overdiagnosed in countries with a higher standard of living. The possible endogeneity of estimated ADHD prevalence rates to the country's level of economic development might further reflect the impairing nature of ADHD-related behavioral symptoms in an economically more complex environment, as manifested in an increased use of prescribed ADHD medications in economically advanced countries (Scheffler et al., 2007). To account for this type of endogeneity, the results reported in column (4) includes the natural logarithm of real GDP per capita as a measure of the country's standard of living in the regression model. Although the coefficient associated with the log of real GDP per capita has the expected positive sign, it is not statistically significant at conventional significance levels. The effect of the country-level  $DRD4^{R2R7}$  measure on estimated ADHD prevalence rates remains relatively strong and precisely estimated. To examine whether differences in reported ADHD prevalence rates across countries result from differences in the country's health system, column (5) includes the mean life expectancy at birth into the regression model. The hypothesis is that a highly advanced health system – approximated by using the country's life expectancy at birth – might provide improved medical services both for the diagnosis and treatment of ADHD cases. The life expectancy variable enters insignificantly into the regression model, whereas the impact of the country-level  $DRD4^{R2R7}$  measure on estimated ADHD prevalence rates remains rather robust and statistically significant at the 1% significance level.

Another concern could be that ADHD prevalence rates reflect differences in a country's educational system. For example, awareness of ADHD-related behavioral symptoms in society probably increases with the amount of time people spend in school. To rule out this possibility, the specification presented in column (6) controls for a human capital index, which enters insignificantly into the regression model. The regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure remains virtually unchanged to the inclusion of the human capital variable.

Finally, the remaining two model specifications presented in columns (7) and (8) are intended to account for the possibility that ADHD-impaired individuals might experience serious challenges in educational and occupational life in more mature societies that have developed complex hierarchical organizational systems in which people coexist and interact (Williams and Taylor, 2006). In modern societies, this requires a minimum of strategic self-control for an individual's economic and social well-being (Moffitt et al., 2011). To account for the concern that estimated ADHD prevalence rates might be higher in more mature societies, in column (7) we first consider a measure of early state experience between the year 1 AD and 1500 AD. This variable indicates the present-day historical experience of a country's population with early state formation and centralization during the pre-colonial era. This measure has been ancestry-adjusted to account for global migration flows across countries that has occurred since 1500 AD using data on international migration flows from the Putterman and Weil (2010) database. Second, in column (8) we control for the country's Neolithic transition timing (i.e., the number of years when the country shifted to agricultural practices up to the year 2000 AD) to account for the relative importance of sedentary practices in society. Accordingly, this variable has been ancestry-adjusted to control for large international migration flows since 1500 AD. Including these measures subsequently in the regression model leaves the main findings

regarding the positive association between estimated ADHD prevalence rates and the country-level  $DRD4^{R2R7}$  measure unchanged. Regarding the estimated magnitudes, the regression coefficients reported in the complete model specification of column (8) suggest that a one percentage point increase in the country-level  $DRD4^{R2R7}$  measure would, ceteris paribus, increase the ADHD prevalence rate in the general population by about 0.4218 percentage points.

In summary, the results presented in Table 2 qualify the  $DRD4^{R2R7}$  measure as a strong biomarker for measuring the prevalence of ADHD-related behavioral symptoms across countries. We advocate the use of the country-level  $DRD4^{R2R7}$  measure over estimated ADHD prevalence rates from study-level data, since the latter variable seems to be heavily confounded by study-level methodological characteristics. Additionally, in contrast to the proposed country-level  $DRD4^{R2R7}$  measure, the identification and diagnosis of ADHD cases might be sensitive to unobserved individual characteristics (e.g., age, gender, ethnic background, parents' socioeconomic status, and family structure of ADHD probands), thus limiting the external validity of estimated ADHD prevalence rates from study-level data for the general population. More importantly, data coverage of estimated ADHD prevalence rates is limited in scope for both countries and ethnic groups, thus restricting their practical value in empirical cross-country studies.

#### 5 Education Production Function, ADHD, and Identification Approach

This section sets the theoretical framework for the analysis of the relationship between cognitive ability and the various family, peer, school, and innate ability factors. We point out that accounting for the issue of innate abilities in the empirical analysis of the EPF is crucial for consistent estimation of the regression coefficients associated with the various input factors that explain cognitive ability outcomes. We then proceed by specifying a cross-country variant of the EPF relating aggregate student achievement test scores to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure – as an important innate ability factor in the specification of the EPF – that is robust to a large set of possible confounding factors arising from unobserved country-specific heterogeneity.

#### 5.1 Education Production Function

For the theoretical foundation regarding the relationship between cognitive achievement and the prevalence of ADHD related behavioral symptoms in society, we rely on a large body of studies in the human capital literature. According to Todd and Wolpin (2003), this literature can be classified into one branch examining how early childhood development (ECD) is influenced by home environment and parental inputs (Cunha and Heckman, 2007), and a second branch, where an EPF is constructed to describe the relationship between family, peers, schooling, and innate endowment inputs and the corresponding cognitive achievement outcome – for example, as measured by the results of ISATs (Hanushek and Woessmann, 2011). In the following, we discuss potential challenges associated with the estimation of EPFs in the presence of unobserved childhood innate ability factors. The motivation is that childhood behavioral symptoms, such as continued attention to school tasks, are determined by unobserved innate ability factors that significantly interfere with the child's learning success and, thus, subsequent educational and occupational achievement. We highlight the importance of the proper modelling of innate ability factors in the specification of the EPF in order to consistently estimate the

<sup>&</sup>lt;sup>18</sup>See Hanushek (2002) for a review of the relevant literature.

remaining regression coefficients associated with the set of schooling, family, and peer inputs.

The EPF methodology can be applied either to evaluate policy measures as an experimental data design or to identify and interpret parameters of the production function using non-experimental data. In the present case, the latter consideration applies. Thus, we follow the methodology outlined in Hanushek (2002) and use the following general microeconomically based framework describing the key determinants of an individual's educational outcomes:

$$CAO_{sa} = f\left(\mathbf{F}_{sa}, \mathbf{P}_{sa}, \mathbf{S}_{sa}, \mathbf{A}_{s}, v_{sa}\right),\tag{3}$$

where  $CAO_{sa}$  is the individual cognitive achievement outcome of student s at age a that is determined by a general function  $f(\cdot)$  consisting of a set of family inputs  $\mathbf{F}_{sa}$ , peer inputs  $\mathbf{P}_{sa}$ , school inputs  $\mathbf{S}_{sa}$ , and the child's innate ability factors  $\mathbf{A}_s$  that is usually modelled as being invariant across ages (e.g., due to the child's genetic endowment). Finally,  $v_{sa}$  is an idiosyncratic error term.

The empirical application of Equation (3) crucially hinges on the availability of the various family, peer, and school input variables. The conceptual framework developed in Todd and Wolpin (2003) indicates that in the analysis of student cognitive achievement, both current and past inputs matter, suggesting the cumulative nature of the acquisition of future cognitive skill outcomes. However, in empirical applications of the EPFs, the relevant data usually lack the longitudinal dimension of individual cognitive achievement outcomes among different ages. Up to the point at which individual cognitive ability is assessed, students receive or use a variety of educational inputs that constitute the status quo of their performance. However, prior educational investments and cognitive achievement cannot be measured retrospectively, or if at all, only by proxy variables. The estimation of the EPF is further complicated by the facts that data on the set of family, peer, and school inputs are relatively incomplete, that student achievement test scores are subject to measurement error problems, and that the data on the child's heritable (i.e., innate) endowment are usually unobserved and/or difficult to model in empirical applications.

The problem of missing data on children's innate endowments poses a serious obstacle to the estimation of general EPFs when using non-experimental observational data. The reason is that innate ability factors are important for the development of a child's mental capacity and behavior, which in turn determine the child's learning outcomes in school and subsequent educational achievement (Cunha and Heckman, 2007). The identifying assumption usually set in the consistent estimation of the EPF is that the unobserved student innate ability factor is uncorrelated with the remaining family, schools, and peer inputs. However, this assumption is likely to be violated when using non-experimental observational data. For example, most childhood development studies suggest that adverse in utero experiences, including poor maternal health, substance abuse, and malnutrition can have severe and long-lasting implications for future childhood cognitive outcomes. Additionally, Cunha and Heckman (2007) point out that the old "nature" versus "nurture" distinction has become obsolete and has been replaced by the modern epigenetic view in which a child's cognitive ability is heavily influenced by a complex interaction between genetic endowments and susceptibility to environmental conditions. Thus, the modern epigenetic view would reject the widespread assumption in the human capital literature that innate ability factors are uncorrelated with the child's family and educational environment. In other words, family and educational

<sup>&</sup>lt;sup>19</sup>For example, Hanushek and Kimko (2000) estimate a cross-country variant of the EPF shown in Equation (3) based on the restrictive assumption that average ability of students does not vary across countries or is at least orthogonal to the remaining input factors.

inputs cannot be analyzed independently of the child's innate endowment in the analysis of traditional EPFs. 20

It is obvious that not accounting for the child's innate ability factors would result in inconsistent estimates of the various family and school input factors. The severity of the omitted variables bias in the estimation of the EPF is evidenced in a series of empirical studies analyzing the determinants in the variation of individual cognitive achievement outcomes. For example, the empirical analysis in Schuetz et al. (2013) reveals that student characteristics (e.g., age, gender, and grade outcomes), family background (e.g., parents' marital, working, and economic status), school inputs (e.g., school's location in the community, educational expenditures per student, and class size), and institutional structures of schools (e.g., private versus public school, accountability, and school autonomy) together only account for between 30 and 40 percent of the overall variation in individual student achievement test scores, indicating the importance of (unobserved) innate student-level ability factors in the estimation of EPFs. Additionally, Anger and Heineck (2010) show that the explanatory power in the analysis of child's cognitive ability increases considerably once the regression model accounts for the parent's cognitive abilities, suggesting the importance of intergenerational transmitted genetic and cultural traits in the acquisition of students' cognitive skill outcomes.

Different estimation approaches have been proposed to overcome the endogeneity problem in the estimation of EPFs depending on the availability of student-level data.<sup>21</sup> The most prominent approaches are fixed-effects and first-difference specifications, which effectively account for endogeneity problems in the estimation of the EPF resulting from the presence of unobserved individual level heterogeneity that is invariant across different ages and partly determined by the child's initial innate ability at birth. Unfortunately, the missing longitudinal dimension in the collection of student achievement test score outcomes prevents the application of the aforementioned model specifications for the current analysis. In addition, even if the data fulfill the necessary longitudinal dimension, we are explicitly (rather than implicitly) interested in uncovering the specific innate ability factors – particularly the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity – that give rise to differences in individual cognitive achievement outcomes.

#### 5.2 Econometric Specification

We tackle the issue of endogeneity in the relationship between aggregate student achievement test scores and the various input factors by estimating a cross-country variant of Equation (3) similar to the specification proposed in Hanushek and Kimko (2000). Thus, we relate aggregate student achievement test scores to country-specific socioeconomic and educational characteristics analogous to parental and school resource variables in the specification of student-level EPFs. Specifically, the following cross-country education production function is estimated:

$$SATS_c = \beta_0 + \beta_1 DRD4_c^{R2R7} + \beta_2' BG_c + \beta_3' MR_c + \beta_4' LO_c + \beta_5' Z_c + \beta_6' R_c + \varepsilon_c, \tag{4}$$

where  $SATS_c$  is the aggregate student achievement test score (averaged across the two subjects mathematics and science) of country c during the various assessment waves from 1995 to 2015, and  $DRD4_c^{R2R7}$  refers to the country-level DRD4

<sup>&</sup>lt;sup>20</sup>This assumption is also consistent with the optimizing behavior of parents, indicating that the amount of financial resources and educational investment is subject to the child's initial cognitive endowment (Becker and Lewis, 1973). Moreover, even schools use different class intervention measures that are designed to match the child's general cognitive ability (Todd and Wolpin, 2003).

<sup>&</sup>lt;sup>21</sup>See Todd and Wolpin (2003) for an excellent discussion of the various approaches to estimate educational production functions and the underlying assumptions for consistent estimation of the various input factors.

exon III 2- and 7-repeat allele frequency measure of country c. In addition to the key dependent and explanatory variable, we employ a large set of country-specific controls in the regression equation to cope with endogeneity concerns in the association between aggregate cognitive achievement and the country-level DRD4 exon III 2- and 7-repeat allele frequency measure. Specifically, the baseline specification includes a full set of biogeographic  $BG_c$ , major religion  $MR_c$ , legal origin  $LO_c$ , and region  $R_c$  effects in the education production function. We conduct a series of additional sensitivity tests to examine the robustness in the association between aggregate student achievement test scores and the country-level  $DRD4^{R2R7}$  measure through the inclusion of the vector  $Z_c$ , which refers to a large battery of health-related, socioeconomic, educational, genetic, and diversity factors. Finally, the variable  $\varepsilon_c$  is a stochastic error term.

It is worth noting that the empirical analysis of aggregate cognitive achievement is not confounded by the presence of overall human genetic diversity, as we explicitly account for this kind of confounding factor in the relationship between aggregate cognitive achievement and the country-level DRD4 exon III 2- and 7-repeat allele frequency measure. <sup>22</sup>

Theory suggests a negative sign for the regression coefficient  $\beta_1$ , which would indicate a negative relation between the country-level  $DRD4^{R2R7}$  measure and aggregate student achievement test scores. This finding would be consistent with the educational burden of the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity for cognitive achievement outcomes in society.

#### 6 Main Empirical Results

Biogeographic and Historical Factors. Table 3 presents the first results on the association between the average cognitive achievement test score variable and the country-level  $DRD4^{R2R7}$  measure in a cross-section of 81 countries. The estimates presented in column (1) show the univariate relationship between both key variables conditional on a full set of region, OPEC, and TIMSS fixed effects. The latter variable accounts for possible differences in the PISA and TIMSS achievement test scores across countries. However, given the high correlation between the two achievement test scores, the estimated regression coefficient associated with the TIMSS indicator variable is statistically insignificant at conventional significance levels across the various model specifications (results not shown). The inclusion of the OPEC indicator variable accounts for the natural resource curse hypothesis, which suggests that countries with a higher amount of proven mineral resources might have poorer growth performance due to widespread corruption and rent-seeking activities that eventually undermine the individual's incentive to invest in the accumulation of human capital. More importantly, the influence of the country-level  $DRD4^{R2R7}$  measure on the dependent variable is negative and highly statistically significant at the 1% significance level, consistent with the educational burden hypothesis of ADHD-related symptoms in society overall. The point estimate suggests that increasing the median value of the  $DRD4^{R2R7}$  measure in the 81-country sample (i.e., Singapore = 0.2376) to the  $DRD4^{R2R7}$  value in the United States (i.e., 0.2636) would, ceteris paribus, decrease the average composite cognitive skill measure by about  $(0.2636 - 0.2376) \times 2.8806 \times 100 \approx 7.4896$  test score points.<sup>23</sup> This effect is

<sup>&</sup>lt;sup>22</sup>The inclusion of overall genetic diversity in the estimation of the EPF captures the notion of international differences in average ability of students consistent with the hypothesis that a higher continuum of human traits in society might result in the emergence of those traits that bring forth individuals with high cognitive abilities (Ashraf and Galor, 2013).

 $<sup>^{23}</sup>$ According to the estimates in column (8) of Table 2 this change in the value of the  $DRD4^{R2R7}$  measure would correspond to a rise in the estimated ADHD prevalence rate of about  $(0.2636 - 0.2376) \times 0.4218 \times 100 \approx 1.0967$  percentage points.

quite sizeable and corresponds approximately to a  $(0.0749/0.6240) \times 100 \approx 12.0032$  percentage change in the standard deviation of the average student achievement test score variable.

Next, the estimates reported in columns (2) and (3) are intended to assess the potential burden of the disease environment on general cognitive ability through the inclusion of various biogeographical controls. None of these control variables enter significantly into the regression model, leaving the regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure qualitatively unaffected both in magnitude and statistical significance.

In column (4), we include the percentage of a country's arable land area to account for the possibility that a favorable agricultural land endowment might result in a higher level of economic activity in the agricultural sector that pays less attention to the accumulation of human capital in society. However, the point estimate of the country-level  $DRD4^{R2R7}$  measure remain rather stable and highly statistically significant at the 5% significance level.

The results presented in column (5) examine the potential confounding influence of major religion effects on average student achievement test scores. Interestingly, the regression coefficient associated with the Protestant religion share enters into the regression model positively and statistically significantly at the 1% significance level, consistent with the notion of Weber's Protestant work ethic. Accordingly, the coefficient associated with the share of Catholics in the population is significant, although only at the 10% significance level, and considerably smaller in magnitude. Controlling for these rough cultural dimension controls in the empirical EPF even results in a significant increase of the estimated magnitude associated with the country-level  $DRD4^{R2R7}$  measure.

Finally, the results shown in column (6) reveal that the association between the aggregate student achievement test score and the country-level  $DRD4^{R2R7}$  measure is rather robust to the inclusion of a full set of legal origin factors in the regression model. Interestingly, the results show quite significantly that countries with a German legal tradition have on average higher student achievement test scores than countries with a socialist legal origin. This finding provides evidence that the institutional framework in which people coexist and cooperate significantly influences individual decisions about the accumulation of human capital over the life course. In addition, the regression coefficient associated with the shares of Protestants and Catholics in the population turns insignificant in this augmented model specification. This finding is consistent with the notion that culture shapes the institutional framework in society, and that when conditioning on the country's legal tradition, the impact of cultural factors on the accumulation of human capital becomes less pronounced (Gorodnichenko and Roland, 2017). According to this final model specification, the estimated regression coefficients suggest that increasing the country-level  $DRD4^{R2R7}$  value of Singapore (i.e., 0.2376) to the level of the United States (i.e., 0.2636) would, ceteris paribus, result in a reduction of the average student achievement test score by about  $(0.2636 - 0.2376) \times 3.8765 \times 100 \approx 10.0789$  points. Again, this effect is quite sizable, as it corresponds roughly to a  $(0.1008/0.6240) \times 100 \approx 16.1538$  percentage change in the standard deviation of the average student achievement test score variable.

**Health-Related Factors.** Table 4 examines the robustness of the regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure to the inclusion of various health-related factors. The results presented in column (1) refer to the baseline specification reported in Table 3, column (8) and is reproduced for comparison purposes.

Column (2) includes the country's share of infants born with low birth weight. As described in greater detail above, this variable captures the general provision of health care and nutrition during pregnancy and can therefore be interpreted as

the result of the newborn's in utero experiences. Theory suggests that negative in utero experiences inhibit intrauterine development and therefore result in reduced subsequent postnatal development potential (Bhutta et al., 2002; Gluckman et al., 2008). This finding is confirmed by a negative estimated regression coefficient associated with the low birth weight variable, which is statistically significant at the 5% significance level. More importantly, the estimated regression coefficient and statistical significance associated with the key country-level  $DRD4^{R2R7}$  measure remain qualitatively unchanged by this model specification.

In column (3), the infant mortality rate per 1,000 infants is included in the regression equation, which proxies for the general provision of medical services, nutrition, and sanitation for infants in society and its possible consequences for subsequent cognitive achievement outcomes. As expected, the infant mortality rate enters into the regression model with a negative and highly statistically significant regression coefficient. The estimated magnitude suggests that an average increase in infant deaths by one child per 1,000 infants would, ceteris paribus, result in a reduction of the country-level student achievement test score by about 2.8382 points. Or to put it differently, decreasing the infant mortality rate of Uruguay (the country closest to the median infant mortality rate of 14.0308 infant deaths per 1,000 infants in the 81-country sample) to the level of Germany (i.e., 4.4115 infant deaths per 1,000 infants) would, ceteris paribus, increase Uruguay's student achievement test score by about  $(4.4115-14.03808) \times (-0.0284) \times 100 \approx 27.3395$  points. Although the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure is slightly reduced in magnitude, it is still of the expected negative sign and remains highly statistically significant at the 1% significance level.

In model specification (4), life expectancy at birth is included in the regression equation. As predicted by life-cycle models of schooling and earnings, longevity creates incentives for investment in education as the phase of working life is longer. As predicted, the estimated regression coefficient associated with life expectancy is both positive and highly statistically significant at conventional significance levels, confirming that longevity is beneficial to educational achievement outcomes, similar to previous findings reported in other studies (Strulik and Werner, 2016). An additional expected year of life increases aggregate student achievement test scores by about 6.6526 points. In addition, the latter finding is further consistent with the notion that higher life expectancy partly reflects the quality of the country's health system and its corresponding beneficial effect on overall cognitive ability. Again, the main findings associated with the country-level  $DRD4^{R2R7}$  variable remain relatively robust and precisely estimated.

Column (5) includes the disability-adjusted life years lost as an additional variable indicating the country's overall health situation (i.e., disease environment and quality of the health system). As the DALY refer not only to early death but also to expected life years lost due to illness or disability, they share the same transmission channels as the life expectancy variable, but additionally account for the burden of diseases and their implications for labor productivity. Severe diseases or disabilities decrease the expectancy of healthy life and therefore result in an increase in DALY. Again, this might discourage individuals from investing in education, but could also measure impairments caused by or occurring as a co-morbidity of ADHD. As hypothesized, the regression coefficient associated with the DALY variable enters with the expected negative sign into the regression equation and is highly statistically significant at the 1% significance level. Regarding the estimated magnitude, the coefficient estimates suggest that decreasing the value of the DALY for Latvia (i.e., DALY = 0.4479) to the level of Germany (i.e., DALY = 0.3019) would, ceteris paribus, increase aggregate student achievement test scores by about  $(0.3019 - 0.4479) \times (-2.2591) \times 100 \approx 32.9829$  points. As in the previous specification, the regression coefficient associated with the country-level  $DRD4^{R2R7}$  variable remains qualitatively unchanged in this

model specification.

In column (6), we include a country's homicide rate (i.e., the number of intentional homicides per 100,000 inhabitants) in the regression equation. This variable proxies for the criminal environment and the overall presence of crime in daily life. Crime is thought to have a direct effect on education by creating an unfavorable environment and discouraging students from pursuing a lawful career. Additionally, the inclusion of this control variable in the regression equation accounts for an important source of confounding in the association between aggregate student achievement test scores and the country-level  $DRD4^{R2R7}$  measure, as the latter might to some extent reflect the detrimental influence of criminal acts in society on general cognitive ability triggered by a higher prevalence of the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity. As expected, the regression coefficient associated with the homicide rate is negative and weakly statistically significant at the 10% significance level. The estimates suggest that reducing Colombia's homicide rate (42.4533) to the level of Germany (0.8200) would, all else equal, boost student achievement test scores by about  $(0.8200 - 42.4533) \times (-0.0113) \times 100 \approx 47.0456$  points, which is a relatively large effect, amounting to approximately a  $(0.4705/0.6240) \times 100 \approx 75.4006$  percentage change in the standard deviation of the average student achievement test score variable. Interestingly, the inclusion of the country's homicide rate in the regression equation results in a considerable increase in the estimated magnitude associated with the country-level  $DRD4^{R2R7}$  measure, suggesting that the previous model specification suffered from a serious omitted variables bias problem.

In column (7), we simultaneously include all of the aforementioned health-related measures reported in model specifications (2) to (6) in the regression model. The infant mortality rate, as before, enters with a negative point estimate and remains highly statistically significant at the 1% significance level. The estimated regression coefficient for the homicide rate remains qualitatively unaltered, while the remaining coefficients associated with low birth weight, life expectancy, and DALY now turn statistically insignificant at conventional significance levels in this final model specification, perhaps due to the presence of multi-collinearity among the various health related control variables. Nevertheless, the main findings regarding the key country-level  $DRD4^{R2R7}$  measure remain qualitatively unaffected even in this augmented model specification, suggesting that this variable does not unintentionally measure either of the health-related factors mentioned above, but instead provides, as intended, a valid approximation of the educational burden of the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity on overall cognitive ability.

**Economic Environment Factors.** In this section, we examine the robustness of the main findings to the inclusion of a set of economic environment controls in the regression equation. The corresponding estimates are shown in Table 5. As before, the results presented in column (1) portray the baseline model specification, as previously shown in column (8) of Table 3.

A potential concern might be that educational achievement is actually driven by socioeconomic factors and that the country-level  $DRD4^{R2R7}$  variable captures international differences in economic development and knowledge creation to some extent (Gören, 2017a,b). As the main indicator of economic prosperity, we include the country's logarithm of real GDP per capita in the regression model, as shown in column (2). As expected, the regression coefficient associated with economic development enters with the expected positive sign and is highly statistically significant at the 1% significance level, consistent with the notion that economically more prosperous countries achieve higher student achievement test score outcomes, perhaps due to better-quality infrastructure. As hypothesized, the relationship between aggregate stu-

dent achievement test scores and the country-level  $DRD4^{R2R7}$  measure is partly confounded by the omission of GDP per capita from the regression model, as indicated by the substantial decrease in the estimated magnitude associated with the country-level  $DRD4^{R2R7}$  measure. However, the regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure still has the expected negative sign and remains statistically significant at the 5% significance level.

Next, we examine the sensitivity of the main findings to the inclusion of the government consumption share in the regression model. Again, the assumption is that higher government spending captures to some extent the positive effect of a country's fiscal capacity to finance the provision of basic infrastructure services (e.g., kindergarten, museums, and libraries) on aggregate student achievement outcomes, but also captures growth-distorting government taxes that may discourage the accumulation of human capital in society. Thus, the estimated sign of the regression coefficient associated with the government consumption share variable would provide information as to which effect predominates. The results shown in column (3) indicate that although the government consumption share variable enters negatively into the regression model, it is statistically insignificant at conventional significance levels. More importantly, the main findings regarding the country-level  $DRD4^{R2R7}$  measure remain qualitatively unchanged to this model specification.

The estimates conducted in columns (4) to (5) include the total fertility rate (i.e., the number of births per woman) and the population growth rate in the regression model. These model specifications account for the educational effect of the quality vs. quantity trade-off in child rearing and its consequences for aggregate student achievement outcomes. With more children, limited financial resources and parenting time have to be distributed across a larger number of children, thus theory would predict that lower per-child investments in education might have adverse effects on individual cognitive achievement outcomes.<sup>24</sup> The regression coefficient associated with the country's total fertility rate enters with the expected negative sign into the regression model and is highly statistically significant at the 1% significance level. Although the regression coefficient associated with the key country-level  $DRD4^{R2R7}$  measure is reduced in magnitude, it retains the expected negative sign and remains highly statistically significant at the 1% significance level. Likewise, the regression coefficient associated with the population growth rate is negative and statistically significant at the 5% significance level, as shown in column (4). In contrast to column (3), the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure even increases in magnitude as compared with the baseline specification, and remains highly statistically significant at the 1% significance level.

In column (6), we account for the potential positive effect of the country's openness to international trade on aggregate student achievement test scores. It has been argued that participation in international production chains might positively affect educational achievement outcomes and thereby help to maintain a country's competitiveness in international markets (Jamison et al., 2007). Consistent with this hypothesis, the estimated regression coefficient associated with the trade openness variable enters with the expected positive sign into the regression model and is statistically significant at conventional significance levels. Reassuringly, the main results regarding the country-level  $DRD4^{R2R7}$  measure remain relatively robust to the inclusion of trade openness in the regression model and precisely estimated.

The estimates reported in model specification (7) jointly include the set of economic environment factors in the regression equation. In this augmented model specification, only the regression coefficient associated with the population growth

 $<sup>^{24}</sup>$ It is worth mentioning that the inclusion of these variables further accounts for an important source of confounding factors in the association between student achievement test scores and the country-level  $DRD4^{R2R7}$  measure, as ADHD studies suggest that ADHD-impaired individuals have earlier sexual encounters and more frequent extramarital and/or unprotected intercourse.

rate variable remains statistically significant at the 5% significance level. Although the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure falls in magnitude, it still has the expected negative sign and remains highly statistically significant at the 1% significance level.

Finally, the results reported in column (8) confirm the robustness of the previous findings to the inclusion of the various health-related factors in the regression model.

Social Value Factors. In Table 6, we present coefficient estimates that account for various social value factors in the empirical analysis of aggregate student achievement test score outcomes. As in prior tables, we examine the sensitivity of the main findings regarding the country-level  $DRD4^{R2R7}$  measure shown in column (2) relative to the inclusion of the set of social value controls.

In the following two model specifications, we consider various gender equality indicators to assess the impact of a genderbiased cultural environment on the accumulation of human capital in society. Specifically, the results shown in column (2) include the proportion of seats held by women in national parliaments in the regression model. Although this variable enters into the regression model with the expected positive sign, it is statistically insignificant at conventional significance levels. In addition, the estimates presented in column (3) examine the impact of the proportion of females in the total labor force on aggregate student achievement test score outcomes. Again, this variable does not enter significantly into the regression model. In both model specifications, the results regarding the key country-level  $DRD4^{R2R7}$  variable remain unchanged, in terms of both the estimated magnitude and statistical significance.

In column (4), we include a binary indicator variable for the presence of a federalist political system. As stated above, a higher degree of decentralization might be associated with a more efficient provision of public educational services (Falch and Fischer, 2012). In line with the theory, the estimated regression coefficient enters positively and statistically significantly into the regression model, although only at the 10% significance level. Again, the main findings regarding the key country-level  $DRD4^{R2R7}$  variable remain qualitatively unchanged to this model specification.

In specifications (5) and (6), we include two indicators assessing the country's institutional environment. In both model specifications, the regression coefficients associated with the protection of property rights and the freedom from corruption variable, respectively, enter with the expected positive sign into the regression model. Furthermore, both regression coefficients are highly statistically significant at the 1% significance level. This result is in accordance with the findings from previous studies, consistent with the hypothesis that a favorable institutional environment facilitates the accumulation of human capital in society due to higher future market returns resulting from educational investment decisions (Reinikka and Svensson, 2005).

In column (7), we include the logarithm of the number of civil conflicts in the regression model to account for the detrimental effects of a war-prone environment on cognitive achievement outcomes. Although the estimated regression coefficient associated with the civil conflicts variable has the expected negative sign, it enters statistically insignificantly into the regression model. The main results regarding the key country-level  $DRD4^{R2R7}$  measure remain qualitatively unchanged in this model specification.

In column (8), we jointly include the full set of social value factors in the regression equation. None of the social value controls included remain statistically significant in this 'horse race' specification. The point estimate regarding the country-level  $DRD4^{R2R7}$  measure decreases slightly in magnitude, yet still remains statistically significant at the 1% significance

level.

After including the health-related factors examined in column (7) of Table 4, the estimated regression coefficients associated with the various social value controls still remain statistically insignificant at conventional significance levels, as shown in the last column (9). Again, the estimated regression coefficient associated with the key country-level  $DRD4^{R2R7}$  measure remains relatively robust and precisely estimated, even in this augmented model specification.

Educational Factors. In Table 7, we examine the predictive power of a full set of educational input factors in the regression model as the *prima facie* most obvious determinants for the accumulation of cognitive abilities. In addition, we account for an additional source of confounding in the association between student achievement test scores and the country-level  $DRD4^{R2R7}$  measure, as the latter variable might to some extent reflect the adverse effect on cognitive achievement resulting from the poor educational achievement of ADHD impaired students.

In column (1) we report, again, the baseline specification previously shown in column (6) of Table 3, but this time restricted to a limited sample of 73 country observations. The reason for reducing the sample is the lack of data available on the various educational input factors for some countries in the extended 81-country sample.

In columns (2) to (4), we include various measures indicating the share of the population aged 15 and above with primary, secondary, or no educational attainment. However, none of the three measures enter significantly into the regression model. More importantly, the main results regarding the country-level  $DRD4^{R2R7}$  measure remain qualitatively unchanged to these model specifications.

In column (5), we include the logarithm of the human capital index as a proxy for the overall quantity of schooling in a country in the regression model. However, the estimated regression coefficient of the human capital index is statistically insignificant, suggesting that it does not contribute to the accumulation of cognitive skills once a full set of biogeographic, historical, and region factors are accounted for in the regression equation. Again, the regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure remains virtually unchanged in magnitude and statistical significance, suggesting that the omission of school quantity does not significantly confound the association between aggregate student achievement test score outcomes and the country-level  $DRD4^{R2R7}$  variable.

The results presented in column (6), examines the robustness of the main findings to the inclusion of the student-teacher ratio in secondary education, as a real resource measure of classroom size. The estimated regression coefficient associated with the student-teacher ratio has a negative sign but is statistically insignificant at conventional significance levels. This finding is in line with results from other studies, which have found that the student-teacher ratio is a rather poor proxy variable for schooling quality (Wright et al., 1997; Hoxby, 2000). Again, the estimated regression coefficient associated with the  $DRD4^{R2R7}$  measure remains relatively robust and precisely estimated.

Next, we consider the effect of education expenditures on cognitive skills formation by including governmental expenditures on education expressed as a percentage of total GDP, as shown in model specification (7). Theory suggests that increased schooling resources lead to better learning opportunities and thus to improved educational performance. However, the educational expenditure variable enters into the regression model statistically insignificantly, consistent with the hypothesis that the influence of financial resources on the educational process is largely dependent on the timing of spending (Hanushek, 1997, 2002; Häkkinen et al., 2003). Once educational expenditures are included in the regression model, the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  decreases in magnitude but remains statis-

tically significant at the 5% significance level.

In column (8), the share of private enrollment in primary education is included to account for the possibility that private schools might be more efficient in the provision of basic school services. Additionally, this variable might capture competition effects between the private and public school sector, with potentially positive effects on the formation of cognitive skill outcomes in society. However, the estimated regression coefficient associated with the private enrollment rate variable enters statistically insignificantly into the regression equation<sup>25</sup>, leaving the main findings associated with the key country-level  $DRD4^{R2R7}$  measure qualitatively unchanged.

Entering the set of educational input factors jointly into the regression equation, as shown in column (9), does not alter the main findings substantially. All of the included educational factors remain statistically insignificant at conventional significance levels.

In column (10), we examine the robustness of the previous findings to the inclusion of the various health-related factors, as examined in column (7) of Table 4. Upon controlling for these factors, the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure decreases moderately in magnitude, but still remains statistically significant at the 1% significance level. Interestingly, the estimated regression coefficient associated with the country's population share aged 15 and above with no educational attainment is positive and now turns highly statistically significant at the 1% significance level. This finding suggests that countries with lower educational attainment might select a non-representative sample of high-performing students across the various international student assessment waves.

#### 7 Robustness Analysis

Genetic and Diversity Factors. In this part of the analysis, we examine the sensitivity of the main results to the presence of various aspects of diversity in society. Specifically, it is conceivable that the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure captures the potential influence of genetic, ethnic, or religious diversity on overall cognitive ability. To rule out these additional confounding factors, Table 8 presents coefficient estimates controlling for various diversity measures in the regression model. Again, column (1) presents the baseline specification and is shown for comparison purposes.

In column (2), we consider an ancestry-adjusted measure of overall genetic diversity to account for the possibility that observed differences in cognitive abilities across countries reflect differences in the genetic composition across countries to some extent. This model specification is intended to rule out the possibility that the country-level genetic  $DRD4^{R2R7}$  measure simply captures the issue of overall genetic diversity on student achievement test scores. The regression coefficient associated with the ancestry-adjusted overall genetic diversity measure enters insignificantly into the regression model, leaving the regression coefficient of the country-level  $DRD4^{R2R7}$  measure virtually unchanged with respect to magnitude and statistical significance.

Next, we consider a possible association between HLA heterozygosity and student achievement test performance. It has been argued that HLA diversity might affect student achievement test scores either indirectly through increased inherent resistance to many infectious diseases (e.g., malaria, hepatitis B, and HIV) or directly due to increased susceptibility to

<sup>&</sup>lt;sup>25</sup>This finding is in line with the results reported in Hsieh and Urquiola (2006), who find no support for the hypothesis of rivalry induced by the provision of private schooling in society enhances educational quality.

psychological disorders (e.g., language impairment or cognitive decline in later life). In addition to the aforementioned pathway, HLA diversity might also capture regional pathogen richness and the associated detrimental impact on cognitive ability. However, the estimates reported in column (3) reveal that HLA diversity enters insignificantly into the regression model. Hence, once we have controlled for a full set of cultural, historical, and biogeographic factors, there seems to be no separate statistically significant effect of HLA diversity on student achievement test scores. This observation is further consistent with the notion that the diffusion of key medical discoveries since the global epidemiological transition have lessened the benefits of HLA heterozygote individuals on an array of possible infectious diseases (Cook, 2015). Reassuringly, the regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure remains rather robust and precisely estimated.

In column (4), we examine the robustness of the main findings to the inclusion of an ethno-linguistic diversity measure. Specifically, it is conceivable that the country-level  $DRD4^{R2R7}$  measure merely reflects the detrimental effect of ethno-linguistic diversity on the accumulation of human capital rather than the prevalence of ADHD. Although the ethno-linguistic diversity measure enters into the regression model negatively and statistically significantly, the negative association between the country-level  $DRD4^{R2R7}$  measure and general cognitive ability remains qualitatively unaffected by the potential confounding influence of ethno-linguistic diversity.

In addition, the baseline results remain largely unaffected when including a measure of religious diversity – capturing the notion of the cultural differences within a country originating from different religious values, norms, and beliefs – in the regression model, as indicated in column (5).

Finally, to account for the possible conflating influence of the various genetic and cultural controls, the estimates reported in column (6) include all variables simultaneously in the regression model. Even in this augmented model specification, the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure remains relatively robust and precisely estimated.

Stratified Analysis by PISA Subjects. In Table 9, we analyze whether the main results are affected by the way we constructed the main dependent variable in the analysis of international student achievement test score outcomes. As stated above, our preferred data source is the OECD PISA study, which contains, apart from reading, the subjects of mathematics and science. We replicate the main empirical results, but this time restrict the analysis to a limited sample of 63 country observations for which we have student achievement test scores from PISA only (Panel A). We therefore examine whether the main findings are driven by the inclusion of the TIMSS aggregate student achievement test scores and thus if qualitative differences in both ISATs drive the main results. Additionally, we stratify the empirical analysis by the various PISA subjects, i.e., mathematics and science (Panel B and C), to address possible concerns of subject choice affecting the main findings.

Column (1) corresponds to the bivariate baseline specification in the relationship between the various student achievement test score outcomes and the country-level  $DRD4^{R2R7}$  measure, conditional on a full set of region, OPEC, biogeographic, major religion, and legal origin effects (this corresponds to the baseline model specification (6) of Table 3). Even in this reduced PISA sample of 63 countries, the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure is of the expected negative sign and highly statistically significant at the 1% significance level. In all panels A to C, the coefficients are slightly larger in magnitude, yet do not differ qualitatively from the main findings.

Next, in columns (2) to (4), we examine the robustness of the previous findings by subsequently including the various healthrelated, economic environment, and social value factors in the regression equation. None of these model specifications alter the main findings regarding the key country-level  $DRD4^{R2R7}$  measure substantially. It is worth noting that in all model specifications (2) to (4), the estimated regression coefficient associated with the country-level  $DRD4^{R2R7}$  measure increases in magnitude compared to the main findings based on the extended 81-country sample.

In the remaining columns (5) to (7), we restrict the empirical analysis to the reduced schooling sample, in which both PISA student achievement test scores and data on the various education input factors are available for a total number of 57 countries. In model specification (5), we again only use the basic set of region, OPEC, biogeographic, major religion, and legal origin effects in the regression model. Next, we examine the sensitivity of the main findings to the inclusion of the various education input factors, as shown in column (6). In column (7), we add the full set of health-related factors to the previous model specification. None of these model specifications alter the main findings considerably.

Overall, the stratified analysis by PISA subjects reveals that the main findings are robust to variations of the sample size (i.e., restriction to the PISA sample) and to the inclusion of a large set of additional country-specific biogeographic, historical, regional, health-related, socioeconomic, and educational factors. Across the panel specifications A to C, the influence of the country-level  $DRD4^{R2R7}$  measure on aggregate student achievement test scores is of the expected negative sign and remains statistically significant at conventional significance levels.

Instrumental Variables Estimation Approach. The results discussed so far point to a negative and statistically significant effect of the country-level  $DRD4^{R2R7}$  measure on general cognitive ability. Given the positive association between the DRD4 exon III 2- and 7-repeat allele frequency measure and ADHD-related behavioral symptoms, this finding is suggestive of the proposed hypothesis that ADHD prevalence might play a causal role in predicting cognitive problems. In the following, we present coefficient estimates from an Instrumental Variables (IV) estimation approach to provide a more direct test of this causal link.

Several causal pathways underlying the association between ADHD-related behavioral symptoms and academic achievement have been proposed in the relevant literature. In particular, it is conceivable that frequent frustration with educational failures might lead to decreased motivation and the emergence of behavioral problems associated with inattention, hyperactivity, and impulsivity, which in turn would result in even lower educational outcomes (Arnold, 1997). Such a simultaneous relationship between academic achievement and ADHD-related behavioral problems undermine the identification of causal inferences. Even if the underlying hypothetical model runs unidirectionally from ADHD to academic achievement, omitted variables might result in a spurious relationship between the two outcomes. For example, Hinshaw (1992) argue that the presence of common factors (e.g., language difficulties) might explain both the presence of ADHD-related symptoms and academic underachievement. Both hypothetical causal mechanisms result in endogeneity problems that can only be overcome with the use of a credible IV estimation approach.

Thus, we estimate the following regression equation regarding the association between student achievement test scores and ADHD prevalence:

$$SATS_c = \beta_0 + \beta_1 ADHD_c + \beta_2' \mathbf{X}_c + \beta_3' \mathbf{R}_c + e_c, \tag{5}$$

where X is a vector of country-level controls, R refers to a vector of region controls, and e is the usual error term. Again, simultaneity bias and/or the existence of common factors might blur the relationship between student achievement test

scores and ADHD prevalence. To account for this source of endogeneity, we utilize a set of excluded instrumental variables that satisfy the necessary relevance and exclusion condition in a standard IV estimation approach.

In panel A of Table 10 we report coefficient estimates associated with the country-level ADHD prevalence rate. The results presented in columns (1) to (4) were derived using the OLS estimator, while columns (5) to (8) utilize the Two-Stage Least Square (2SLS) estimator of the proposed IV approach. In panel B, we report the first-stage IV diagnostic tests for under-identification and weak identification of the set of excluded instruments. Given the discussion about the possible determinants of cross-country differences in estimated ADHD prevalence rates, we employ the country-level  $DRD4^{R2R7}$  measure, state history experience, and Neolithic transition timing as excluded instrumental variables.

In line with the proposed hypothetical mechanism, the results presented in columns (1) to (4) point to a negative and statistically significant association between student achievement test scores and ADHD prevalence, which is robust to the inclusion of continent and country-level controls. However, the OLS coefficient estimates regarding the ADHD prevalence rate might be biased due to the presence of simultaneity and/or omitted variables bias. Thus, columns (5) to (8) provide coefficient estimates based on the proposed IV estimation approach. Across the various model specifications, only the results reported in columns (7) and (8) show that the set of excluded instruments are relevant and excludable, as indicated by the corresponding Kleibergen and Paap (2006) rk LM and the Hansen J statistic, respectively. In addition, testing for weak identification of IV, the non-robust Cragg-Donald and heteroskedasticity-robust Kleibergen-Paap rk Wald F statistic indicates that the reported 2SLS coefficient estimate of ADHD in columns (7) and (8) has a maximum relative OLS bias of 10% and 20%, respectively. Although the findings are supportive of a causal link between student achievement test scores and ADHD prevalence, they should be interpreted with caution given the relatively small number of 37 country observations.

### 8 Conclusion

The ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity significantly interfere with child-hood learning success and eventually lead to individual educational underachievement. In contrast to popular belief, ADHD-related behavioral symptoms persist into adolescence and adulthood with direct implications for future educational, occupational, and social life outcomes.

Although a large number of neuropsychological studies have confirmed the impairing nature of ADHD-related behavioral symptoms in daily life activities throughout the life course, no study, to the best of the authors' knowledge, has addressed the macroeconomic burden of ADHD for the accumulation of human capital in a sample of culturally distinct countries. We contribute to the relevant human capital literature on the formation of cognitive skills in society by identifying an important innate ability factor that significantly determines the child's inclination to pay attention to schooling tasks, which is regarded as an essential prerequisite for individual learning success.

In particular, we examine the relationship between international student achievement test score outcomes and a set of

<sup>&</sup>lt;sup>26</sup>The critical values for assessing the weakness of IV are tabulated in Stock and Yogo (2005) for various combinations of the number of endogenous variables, instruments, and the relative size of the OLS bias. Although these values only apply to the non-robust Cragg-Donald Wald F statistic assuming the presence of *i.i.d.* errors, we follow the suggestion in Baum et al. (2007, p. 490) and apply these critical values with caution to the Kleibergen-Paap Wald F statistic, which is robust to the presence of non-*i.i.d.* errors.

specific allele variants of the human DRD4 exon III gene, which candidate gene association studies have found to be a robust biomarker in the etiology of the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity. The negative and statistically significant coefficient associated with the country-level DRD4 exon III 2- and 7-repeat allele frequency measure on aggregate student achievement test score outcomes is consistent with the educational burden of the ADHD-related behavioral symptoms inattention and/or hyperactivity-impulsivity on cognitive achievement outcomes. This finding is robust to the inclusion of a full set of country-specific regional, biogeographic, historical, health-related, socioeconomic, educational, genetic, and diversity factors and across various model specifications.

Several policy implications can be drawn from the findings in this paper. The psychological literature discusses a series of academic and behavioral classroom intervention measures designed to increase children's attention and motivation that may be beneficial for individual learning outcomes and for subsequent academic and occupational achievement, especially for children with ADHD-related behavioral symptoms. These intervention measures include class-wide peer tutoring, task or instructional modifications, parental involvement in completing and checking homework, and individualized academic interventions tailored to the specific needs of ADHD-impaired students at school.<sup>27</sup> Given that the emotional skill of impulse-control significantly predicts an individual's economic, health, and social outcomes, the teaching of such skills in early life seems of pivotal importance to avoid later disabilities, especially in an increasingly complex modern society (Moffitt et al., 2011).

<sup>&</sup>lt;sup>27</sup>See Raggi and Chronis (2006) for an excellent discussion of the various intervention measures designed to improve the educational performance of ADHD-impaired children and adolescents.

### References

- Agarwal, R., Goldenberg, M., Perry, R., and Ishak, W. W. (2012). The Quality of Life of Adults with Attention Deficit Hyperactivity Disorder: A Systematic Review. *Innovations in Clinical Neuroscience*, 9:10–21.
- Alesina, A., Devleeschauwer, A., Easterly, W., Kurlat, S., and Wacziarg, R. (2003). Fractionalization. *Journal of Economic Growth*, 8:155–194.
- Alesina, A. and Fuchs-Schuendeln, N. (2007). Good-Bye Lenin (or Not?): The Effect of Communism on People's Preferences. *The American Economic Review*, 97(4):1507–1528.
- Allen, M. A. and Fischer, G. J. (1978). Ambient Temperature Effects on Paired Associate Learning. Ergonomics, 21(2):95–101.
- Amasuomo, T. T. and Amasuomo, J. O. (2016). Perceived Thermal Discomfort and Stress Behaviours Affecting Students' Learning in Lecture Theatres in the Humid Tropics. *Buildings*, 6(18):1–17.
- American Psychiatric Association (2013). Diagnostic and Statistical Manual of Mental Disorders. Arlington, VA, American Psychiatric Association, 5 edition.
- Ammermueller, A. and Pischke, J.-S. (2009). Peer Effects in European Primary Schools: Evidence from the Progress in International Reading Literacy Study. *Journal of Labor Economics*, 27(3):315–348.
- Anger, S. and Heineck, G. (2010). Do Smart Parents Raise Smart Children? The Intergenerational Transmission of Cognitive Abilities. *Journal of Population Economics*, 23:1255–1282.
- Arnold, D. H. (1997). Co-Occurrence of Externalizing Behavior Problems and Emergent Academic Difficulties in Young High-Risk Boys: A Preliminary Evaluation of Patterns and Mechanisms. *Journal of Applied Developmental Psychology*, 18:317–330.
- Arnold, L. E., Hodgkins, P., Kahle, J., Madhoo, M., and Kewley, G. (2015). Long-Term Outcomes of ADHD: Academic Achievement and Performance. *Journal of Attention Disorders*, pages 1–13.
- Asghari, V., Sanyal, S., Buchwaldt, S., Paterson, A., Jovanovic, V., and Van Tol, H. H. M. (1995). Modulation of Intracellular Cyclic AMP Levels by Different Human Dopamine D4 Receptor Variants. *Journal of Neurochemistry*, 65:1157–1165.
- Ashraf, Q. and Galor, O. (2013). The 'Out of Africa' Hypothesis, Human Genetic Diversity, and Comparative Economic Development. The American Economic Review, 103:1–46.
- Barber, N. (2005). Educational and Ecological Correlates of IQ: A Cross-National Investigation. Intelligence, 33:273–284.
- Barker, D. J. P. (1995). Fetal Origins of Coronary Heart Disease. British Medical Journal, 311:171-174.
- Barkley, R. A., Fischer, M., Smallish, L., and Fletcher, K. (2006). Young Adult Outcome of Hyperactive Children: Adaptive Functioning in Major Life Activities. *Journal of the American Academy of Child and Adolescent Psychiatry*, 45:192–202.
- Barro, R. J. and Lee, J. W. (2013). A New Data Set of Educational Attainment in the World, 1950–2010. *Journal of Development Economics*, 104:184–198.
- Barry, T. D., Lyman, R. D., and Klinger, L. G. (2002). Academic Underachievement and Attention-Deficit/Hyperactivity Disorder: The Negative Impact of Symptom Severity on School Performance. *Journal of School Psychology*, 40:259–283.

- Bauermeister, J. J., Shrout, . P. E., Ramírez, R., Bravo, M., Alegría, M., Martínez-Taboas, A., Chávez, L., Rubio-Stipec, M., García, P., Ribera, J. C., and Canino, G. (2007). ADHD Correlates, Comorbidity, and Impairment in Community and Treated Samples of Children and Adolescents. *Clinical Child and Family Psychology Review*, 35:883–898.
- Baum, C. F., Schaffer, M. E., and Stillman, S. (2007). Enhanced Routines for Instrumental Variables/Generalized Method of Moments Estimation and Testing. Stata Journal, 7:465–506.
- Baumgaertel, A., Wolraich, M. L., and Dietrich, M. (1995). Comparison of Diagnostic Criteria for Attention Deficit Disorders in a German Elementary School Sample. *Journal of the American Academy of Child and Adolescent Psychiatry*, 34:629–638.
- Becker, D. and Rindermann, H. (2016). The Relationship Between Cross-National Genetic Distances and IQ-Differences. *Personality and Individual Differences*, 98:300–310.
- Becker, G. S. (1975). Investment in Human Capital: Effects on Earnings, pages 13-44. NBER.
- Becker, G. S. and Lewis, H. G. (1973). On the Interaction between the Quantity and Quality of Children. *Journal of Political Economy*, 81(2, Part 2):279–288.
- Becker, S. O. and Woessmann, L. (2009). Was Weber Wrong? A Human Capital Theory of Protestant Economic History. *Quarterly Journal of Economics*, 124(2):531–596.
- Ben-Porath, Y. (1967). The Production of Human Capital and the Life Cycle of Earnings. *Journal of Political Economy*, 75(4, Part 1):352–365.
- Bhutta, A. T., Cleves, M. A., Casey, P. H., Cradock, M. M., and Anand, K. J. S. (2002). Cognitive and Behavioral Outcomes of School-Aged Children who were born Preterm A Meta-Analysis. *Journal of the American Medical Association*, 288(6):728–737.
- Biederman, J., Monuteaux, M. C., Mick, E., Spencer, T., Wilens, T. E., Silva, J. M., Snyder, L. E., and Faraone, S. V. (2006). Young Adult Outcome of Attention Deficit Hyperactivity Disorder: A Controlled 10-Year Follow-Up Study. *Psychological Medicine*, 36:167–179.
- Bils, M. and Klenow, P. J. (2000). Does Schooling Cause Growth? American economic review, pages 1160–1183.
- Birchwood, J. and Daley, D. (2012). Brief Report: The Impact of Attention Deficit Hyperactivity Disorder (ADHD) Symptoms on Academic Performance in an Adolescent Community Sample. *Journal of Adolescence*, 35:225–231.
- Black, R. E., Allen, L. H., Bhutta, Z. A., Caulfield, L. E., de Onis, M., Ezzati, M., Mathers, C., Rivera, J., and Study, M. . C. U. (2008). Maternal and Child Undernutrition 1 Maternal and Child Undernutrition: Global and Regional Exposures and Health Consequences. *LANCET*, 371(9608):243–260.
- Black, R. E., Victora, C. G., Walker, S. P., Bhutta, Z. A., Christian, P., de Onis, M., Ezzati, M., Grantham-McGregor, S., Katz, J., Martorell, R., Uauy, R., and the Maternal and Child Nutrition Study Group (2013). Maternal and Child Undernutrition and Overweight in Low-Income and Middle-Income Countries. *Lancet*, 382:427–451.
- Bleakley, H. (2007). Disease and Development: Evidence from Hookworm Eradication in the American South. *Quarterly Journal of Economics*, 122(1):73–117.
- Bleakley, H. (2010). Health, Human Capital, and Development. Annual Review of Economics, 2:283-310.

- Bobb, A. J., Castellanos, F. X., Addington, A. M., and Rapoport, J. L. (2006). Molecular Genetic Studies of ADHD: 1991 to 2004.

  American Journal of Medical Genetics Part B (Neuropsychiatric Genetics), 141B:551–565.
- Brown, R. and Velásquez, A. (2017). The Effect of Violent Crime on the Human Capital Accumulation of young Adults. *Journal of Development Economics*, 127:1–12.
- Brucato, N., Guadalupe, T., Franke, B., Fisher, S. E., and Francks, C. (2015). A Schizophrenia-Associated HLA Locus Affects Thalamus Volume and Asymmetry. *Brain, Behavior, and Immunity*, 46:311–318.
- Bruxel, E. M., Akutagava-Martins, G. C., Salatino-Oliveira, A., Contini, V., Kieling, C., Hutz, M. H., and Rohde, L. A. (2014).

  ADHD Pharmacogenetics Across the Life Cycle: New Findings and Perspectives. *American Journal of Medical Genetics Part B:*Neuropsychiatric Genetics, 165:263–282.
- Cann, H. M., de Toma, C., Cazes, L., Legrand, M. F., Morel, V., Piouffre, L., Bodmer, J., Bodmer, W. F., Bonne-Tamir, B., Cambon-Thomsen, A., Chen, Z., Chu, J., Carcassi, C., Contu, L., Du, R., Excoffier, L., Ferrara, G. B., Friedlaender, J., Groot, H., Gurwitz, D., Jenkins, T., Herrera, R. J., Kidd, X. H. J., Kidd, K. K., Langaney, A., Lin, A. A., Mehdi, S. Q., Parham, P., Piazza, A., Pistillo, M. P., Qian, Y., Shu, Q., Xu, J., Zhu, S., Weber, J. L., Greely, H. T., Feldman, M. W., Thomas, G., Dausset, J., and Cavalli-Sforza, L. L. (2002). A Human Genome Diversity Cell Line Panel. Science, 296:261–262.
- Card, D. (1999). The Causal Effect of Education on Earnings. In Handbook of Labor Economics, volume 3, pages 1801–1863.
  Elsevier.
- Cawley, J., Heckman, J., and Vytlacil, E. (2001). Three Observations on Wages and Measured Cognitive Ability. *Labour Economics*, 8:419–442.
- Chabris, C. F., Lee, J. J., Cesarini, D., Benjamin, D. J., and Laibson, D. I. (2015). The Fourth Law of Behavior Genetics. *Current Directions in Psychological Science*, 24:304–312.
- Chanda, A. and Putterman, L. (2007). Early Starts, Reversals and Catch-Up in the Process of Economic Development. *Scandinavian Journal of Economics*, 109:387–413.
- Chang, S. M., Walker, S. P., Grantham-McGregor, S., and Powell, C. A. (2002). Early Childhood Stunting and Later Behaviour and School Achievement. *Journal of Child Psychology and Psychiatry*, 43(6):775–783.
- Chang, Z., Lichtenstein, P., D'Onofrio, B. M., Sjölander, A., and Larsson, H. (2014). Serious Transport Accidents in Adults with Attention-Deficit/Hyperactivity Disorder and the Effect of Medication: A Population-Based Study. *JAMA Psychiatry*, 71:319–325.
- Charach, A., Dashti, B., Carson, P., Booker, L., Lim, C. G., Lillie, E., Yeung, E., Ma, J., Raina, P., and Schachar, R. (2011). Attention Deficit Hyperactivity Disorder: Effectiveness of Treatment in At-Risk Preschoolers; Long-Term Effectiveness in All Ages; and Variability in Prevalence, Diagnosis, and Treatment. Available at https://www.ncbi.nlm.nih.gov/books/NBK82377/. Accessed on 14/11/2017.
- Chaudhary, L., Muscchio, A., Nafziger, S., and Yan, S. (2012). Big BRICs, weak Foundations: The Beginning of Public Elementary Education in Brazil, Russia, India, and China. *Explorations in Economic History*, 49:221–240.
- Cohen, D. and Leker, L. (2014). Health and Education: Another Look with the Proper Data. CEPR Discussion Paper No. DP9940.
- Cohen, D. and Soto, M. (2007). Growth and Human Capital: Good Data, Good Results. Journal of Economic Growth, 12(1):51-76.

- Cook, C. J. (2015). The Natural Selection of Infectious Disease Resistance and its Effect on Contemporary Health. The Review of Economics and Statistics, 97:742–757.
- Cook, E. H., Stein, M. A., Krasowski, M. D., Cox, N. J., Olkon, D. M., Kieffer, J. E., and Leventhal, B. L. (1995). Association of Attention-Deficit Disorder and the Dopamine Transporter Gene. *American Journal of Human Genetics*, 56:993–998.
- Cotter, D. A., Hermsen, J. M., Ovadia, S., and Vanneman, R. (2001). The Glass Ceiling Effect. Social forces, 80(2):655-681.
- Couch, J. F., Shughart II, W. F., and Williams, A. L. (1993). Private School Enrollment and Public School Performance. Public Choice, 76:301–312.
- Cunha, F. and Heckman, J. J. (2007). The Technology of Skill Formation. American Economic Review, 97(2):31–47.
- Cunha, F., Heckman, J. J., Lochner, L., and Masterov, D. V. (2006). Interpreting the Evidence on Life Cycle Skill Formation.

  Handbook of the Economics of Education, 1:697–812.
- Cunha, F., Heckman, J. J., and Schennach, S. M. (2010). Estimating the Technology of Cognitive and Noncognitive Skill Formation. Econometrica, 78(3):883–931.
- Currie, J. and Stabile, M. (2006). Child Mental Health and Human Capital Accumulation: The Case of ADHD. *Journal of Health Economics*, 25:1094–1118.
- Dahlberg, S., Holmberg, S., Rothstein, B., Alvarado Pachon, N., and Svensson, R. (2018). The Quality of Government Basic Dataset, version Jan18. *University of Gothenburg: The Quality of Government Institute*.
- Dalsgaard, S., Østergaard, S. D., Leckman, J. F., Mortensen, P. B., and Pedersen, M. G. (2015). Mortality in Children, Adolescents, and Adults with Attention Deficit Hyperactivity Disorder: A Nationwide Cohort Study. *The Lancet*, 385:2190–2196.
- De La Croix, D. and Doepke, M. (2009). To Segregate or to Integrate: Education Politics and Democracy. *The Review of Economic Studies*, 76(2):597–628.
- Desmet, K., Ortuño-Ortín, I., and Wacziarg, R. (2012). The Political Economy of Linguistic Cleavages. *Journal of Development Economics*, 97:322–338.
- Diamantopoulou, S., Rydell, A. M., Thorell, L. B., and Bohlin, G. (2007). Impact of Executive Functioning and Symptoms of Attention Deficit Hyperactivity Disorder on Children's Peer Relations and School Performance. *Developmental Neuropsychology*, 32:521–542.
- Doherty, P. C. and Zinkernagel, R. M. (1975). A Biological Role for the Major Histocompatibility Antigens. Lancet, 305:1406–1409.
- Easterly, W. and Levine, R. (1997). Africa's Growth Tragedy: Policies and Ethnic Divisions. *The Quarterly Journal of Economics*, 112:1203–1250.
- Easterly, W. and Levine, R. (2003). Tropics, Germs, and Crops: How Endowments influence Economic Development. *Journal of Monetary Economics*, 50:3–39.
- Ehrenberg, R. G., Brewer, D. J., Gamoran, A., and Willms, J. D. (2001). Class Size and Student Achievement. *Psychological Science in Public Interest*, 2(1):1–30.

- Esrey, S. A., Potash, J. B., Roberts, L., and Shiff, C. (1991). Effects of Improved Water-Supply and Sanitation on Ascariasis, Diarrhea, Dracunculiasis, Hookworm Infection, Schistosomiasis, and Trachoma. *Bulletin of the World Health Organization*, 69(5):609–621.
- Falch, T. and Fischer, J. A. V. (2008). Does a Generous Welfare State Crowd out Student Achievement? Panel Data Evidence from International Student Tests. CESifo Working Paper, (2383).
- Falch, T. and Fischer, J. A. V. (2012). Public Sector Decentralization and School Performance: International Evidence. *Economics Letters*, 114:276–279.
- Faraone, S. V. and Bonvicini, C. (2014). Biomarkers in the Diagnosis of ADHD Promising Directions. *Molecular Neurobiology*, 45:497.
- Faraone, S. V., Perlis, R. H., Doyle, A. E., Smoller, J. W., Goralnick, J. J., Holmgren, M. A., and Sklar, P. (2005). Molecular Genetics of Attention-Deficit/Hyperactivity Disorder. *Biological Psychiatry*, 57:1313–1323.
- Feenstra, R. C., Inklaar, R., and Timmer, M. P. (2015). The Next Generation of the Penn World Table. *American Economic Review*, 105(10):3150–3182.
- Frazier, T. W., Demaree, H. A., and Youngstrom, E. A. (2004). Meta-Analysis of Intellectual and Neuropsychological Test Performance in Attention-Deficit/Hyperactivity Disorder. *Neuropsychology*, 18:543–555.
- Frazier, T. W., Youngstrom, E. A., Glutting, J. J., and Watkins, M. W. (2007). ADHD and Achievement: Meta-Analysis of the Child, Adolescent and Adult Literatures and a Concomitant Study with College Students. *Journal of Learning Disabilities*, 40(1):49–65.
- Fryer, R. G. and Levitt, S. D. (2010). An Empirical Analysis of the Gender Gap in Mathematics. *American Economic Journal:*Applied Economics, 2(2):210–240.
- Fuchs, T. and Woessmann, L. (2007). What accounts for International Differences in Student Performance? a Re-Examination using PISA Data. *Empirical Economics*, 32:433–464.
- Fuchs-Schuendeln, N. and Masella, P. (2016). Long-Lasting Effects of Socialist Education. *The Review of Economics and Statistics*, 98(3):428–441.
- Fulton, B. D., Scheffler, R. M., Hinshaw, S. P., Levine, P., Stone, S., Brown, T. T., and Modrek, S. (2009). National Variation of adhd Diagnostic Prevalence and Medication Use: Health Care Providers and Education Policies. *Psychiatric Services*, 60:1075–1083.
- Gallup, J. L. and Sachs, J. D. (2001). The Economic Burden of Malaria. *The American Journal of Tropical Medicine and Hygiene*, 64:85–96.
- Gizer, I. R., Ficks, C., and Waldman, I. D. (2009). Candidate Gene Studies of ADHD: A Meta-Analytical Review. *Human Genetics*, 126:51–90.
- Glaeser, E. L. and Saks, R. E. (2006). Corruption in America. Journal of Public Economics, 90:1053–1072.
- Global Mapping International (2010). World Language Mapping System, Version 16. http://worldgeodatasets.com/language/index.html.

- Gluckman, P. D., Hanson, M. A., Cooper, C., and Thornburg, K. L. (2008). Effect of In Utero and Early-Life Conditions on Adult Health and Disease. *The New England Journal of Medicine*, 359:61–73.
- Goldenberg, R. L., Culhane, J. F., Iams, J. D., and Romero, R. (2008). Epidemiology and Causes of Preterm Birth. *Lancet*, 371:75–84.
- Goldenberg, R. L., Hauth, J. C., and Andrews, W. W. (2000). Intrauterine Infection and Preterm Delivery. *The New England Journal of Medicine*, 342(20):1500–1507.
- Goldhaber, D. D. (1996). Public and Private High Schools: Is School Choice an Answer to the Productivity Problem? *Economics of Education Review*, 15(2):93–109.
- Gören, E. (2014). How Ethnic Diversity Affects Economic Growth. World Development, 59:275-297.
- Gören, E. (2016). The Biogeographic Origins of Novelty-Seeking Traits. Evolution and Human Behavior, 37:456-469.
- Gören, E. (2017a). The Persistent Effects of Novelty-Seeking Traits on Comparative Economic Development. *Journal of Development Economics*, 126:112–126.
- Gören, E. (2017b). The Role of Novelty-Seeking Traits in Contemporary Knowledge Creation. *University of Oldenburg, Department of Economics, Working Paper No. V-402-17.*
- Gorodnichenko, Y. and Roland, G. (2017). Culture, Institutions and the Wealth of Nations. *The Review of Economics and Statistics*, 99:402–416.
- Grantham-McGregor, S., Cheung, Y. B., Cueto, S., Glewwe, P., Richter, L., Strupp, B., and the International Child Development Steering Group (2007). Developmental Potential in the first 5 Years for Children in Developing Countries. *LANCET*, 369:60–70.
- Gray, S. A., Dueck, K., Rogers, M., and Tannock, R. (2017). Qualitative Review Synthesis: The Relationship Between Inattention and Academic Achievement. *Journal of Health Economics*, 59:17–35.
- Guiso, L., Monte, F., Sapienza, P., and Zingales, L. (2008). Culture, Gender, and Math. Science, 320:1164–1165.
- Häkkinen, I., Kirjavainen, T., and Uusitalo, R. (2003). School Resources and Student Achievement Revisited: New Evidence from Panel Data. *Economics of Education Review*, 22:329–335.
- Hancock, P. A. and Vasmatzidis, I. (2003). Effects of Heat Stress on Cognitive Performance: The Current State of Knowledge. International Journal of Hyperthermia, 19(3):355–372.
- Hanushek, E. A. (1997). Assessing the Effects of School Resources on Student Performance: An Update. *Educational Evaluation and Policy Analysis*, 19(2):141–164.
- Hanushek, E. A. (2002). Publicly Provided Education. In Auerbach, A. J., Chetty, R., Feldstein, M., and Saez, E., editors, *Handbook of Public Economics*, volume 4, pages 2045–2141. Elsevier.
- Hanushek, E. A. and Kimko, D. D. (2000). Schooling, Labor-Force Quality, and the Growth of Nations. *American Economic Review*, 90(5):1184–1208.
- Hanushek, E. A., Schwerdt, G., Wiederhold, S., and Woessmann, L. (2015). Returns to Skills around the World: Evidence from PIAAC. *European Economic Review*, 73:103–130.

- Hanushek, E. A. and Woessmann, L. (2008). The Role of Cognitive Skills in Economic Development. *Journal of Economic Literature*, 46(3):607–668.
- Hanushek, E. A. and Woessmann, L. (2011). The Economics of International Differences in Educational Achievement. In Hanushek, E. A., Machin, S., and Woessmann, L., editors, Handbook of the Economics of Education. Band 3, volume 26, pages 89–200. Elsevier.
- Hanushek, E. A. and Woessmann, L. (2012). Do Better Schools lead to more Growth? cognitive Skills, Economic Outcomes, and Causation. *Journal of Economic Growth*, 17(4):267–321.
- Hawi, Z., Cummins, T. D. R., Tong, J., Johnson, B., Lau, R., Samarrai, W., and Bellgrove, M. A. (2015). The Molecular Genetic Architecture of Attention Deficit Hyperactivity Disorder. *Molecular Psychiatry*, 20:289–297.
- Heckman, J. J., Stixrud, J., and Urzua, S. (2006). The Effects of Cognitive and Noncognitive Abilities on Labor Market Outcomes and Social Behavior. *Journal of Labor Economics*, 24(3):411–482.
- Helpman, E. (1993). Innovation, Imitation, and Intellectual Property Rights. Econometrica, 61(6):1247–1280.
- Hill, A. V. S., Allsopp, C. E. M., Kwiatkowski, D., Anstey, N. M., Twumasi, P., Rowe, P. A., Bennett, S., Brewster, D., McMichael, A. J., and Greenwood, B. M. (1991). Common West African HLA Antigens are Associated with Protection from Severe Malaria. Nature, 352:595–600.
- Hinshaw, S. P. (1992). Externalizing Behavior Problems and Academic Underachievement in Childhood and Adolescence: Causal Relationships and Underlying Mechanisms. *Psychological Bulletin*, 111:127–155.
- Hong, L. and Page, S. E. (2001). Problem Solving by Heterogenous Agents. Journal of Economic Theory, 91:123-163.
- Hong, L. and Page, S. E. (2004). Groups of Diverse Problem Solvers Can Outperform Groups of High-Ability Problem Solvers.

  Proceedings of the National Academy of Sciences, 101:16385–16389.
- Hoxby, C. M. (2000). The Effects of Class Size on Student Achievement. The Quarterly Journal of Economics, 115(4):1239–1285.
- Hsieh, C.-T. and Urquiola, M. (2006). The Effects of Generalized School Choice on Achievement and Stratification: Evidence from Chile's Voucher Program. *Journal of Public Economics*, 90:1477–1503.
- Hughes, A. L. and Yeager, M. (1998). Natural Selection at Major Histocompatibility Complex Loci of Vertebrates. Annual Review of Genetics, 32:415–435.
- Ichino, A. and Winter-Ebmer, R. (2004). The Long-Run Educational Cost of World War II. *Journal of Labor Economics*, 22(1):57–86.
- James, A., Lai, F. H., and Dahl, C. (2004). Attention Deficit Hyperactivity Disorder and Suicide: A Review of Possible Associations. Acta Psychiatrica Scandinavica, 110:408–415.
- Jamison, E. A., Jamison, D. T., and Hanushek, E. A. (2007). The Effects of Education Quality on Income Growth and Mortality Decline. Economics of Education Review, 26:772–789.
- Kaarsen, N. (2014). Cross-Country Differences in the Quality of Schooling. Journal of Development Economics, 107:215-224.

- Kessler, R. C., Adler, L., Barkley, R., Biederman, J., Conners, C. K., Demler, O., Faraone, S. V., Greenhill, L. L., Howes, M. J., Secnik, K., Spencer, T., Ustun, T. B., Walters, E. E., and Zaslavsky, A. M. (2006). The Prevalence and Correlates of Adult ADHD in the United States: Results from the National Comorbidity Survey Replication. *American Journal of Psychiatry*, 163:716–723.
- Kleibergen, F. and Paap, R. (2006). Generalized Reduced Rank Tests Using the Singular Value Decomposition. *Journal of Econometrics*, 133:97–126.
- Klein, R. G., Mannuzza, S., Olazagasti, M. A. R., Roizen, E., Hutchison, J. A., Lashua, E. C., and Castellanos, F. X. (2012). Clinical and Functional Outcome of Childhood Attention-Deficit/Hyperactivity Disorder 33 Years Later. *Archives of General Psychiatry*, 69:1295–1303.
- Kotloff, K. L., Nataro, J. P., Blackwelder, W. C., Nasrin, D., Farag, T. H., Panchalingam, S., Wu, Y., Sow, S. O., Sur, D., Breiman, R. F., Faruque, A. S. G., Zaidi, A. K. M., Saha, D., Alonso, P. L., Tamboura, B., Sanogo, D., Onwuchekwa, U., Manna, B., Ramamurthy, T., Kanungo, S., Ochieng, J. B., Omore, R., Oundo, J. O., Hossain, A., Das, S. K., Ahmed, S., Qureshi, S., Quadri, F., Adegbola, R. A., Antonio, M., Hossain, M. J., Akinsola, A., Mandomando, I., Nhampossa, T., Acacio, S., Biswas, K., O'Reilly, C. E., Mintz, E. D., Berkeley, L. Y., Muhsen, K., Sommerfelt, H., Robins-Browne, R. M., and Levine, M. M. (2013). Burden and Aetiology of Diarrhoeal Disease in Infants and Young Children in Developing Countries (The Global Enteric Multicenter Study, GEMS): A Prospective, Case-Control Study. LANCET, 382(9888):209–222.
- La Porta, R., Lopez-de Silanes, F., Shleifer, A., and Vishny, R. (1999). The Quality of Government. *The Journal of Law, Economics Organization*, 15(1):222–279.
- La Porta, R., Lopez-de Silanes Molina, F., and Shleifer, A. (2008). The Economic Consequences of Legal Origins. *Journal of Economic Literature*, 46(2):285–332.
- LaHoste, G. J., Swanson, J. M., Wigal, S. B., Glabe, C., Wigal, T., King, N., and Kennedy, J. L. (1996). Dopamine D4 Receptor Gene Polymorphism is Associated with Attention Deficit Hyperactivity Disorder. *Molecular Psychiatry*, 1:121–124.
- Landes, D. S. (1998). The Wealth and Poverty of Nations: Why Some Countries are so Rich and Some so Poor. New York, NY: WW Noton.
- Langberg, J. M., Molina, B. S., Arnold, L. E., Epstein, J. N., Altaye, M., Hinshaw, S. P., Swanson, J. M., Wigal, T., and Hechtman, L. (2011). Patterns and Predictors of Adolescent Academic Achievement and Performance in a Sample of Children with Attention-Deficit/Hyperactivity Disorder (ADHD). Journal of Clinical Child and Adolescent Psychology, 40:519–531.
- Leung, P. W., Lee, C., Hung, S., Ho, T., Tang, C., Kwong, S., Leung, S., Yuen, S., Lieh-Mak, F., Oosterlaan, J., Grady, D., Harxhi, A., Ding, Y., Chi, H., Flodman, P., Schuck, S., Spence, M. A., Moyzis, R., and Swanson, J. (2005). Dopamine Receptor D4 (DRD4) Gene in Han Chinese Children With Attention-Deficit/Hyperactivity Disorder (ADHD): Increased Prevalence of the 2-Repeat Allele. American Journal of Medical Genetics Part B (Neuropsychiatric Genetics), 133B:54–56.
- Levy, F. (1991). The Dopamine Theory of Attention Deficit Hyperactivity Disorder (ADHD). Australian and New Zealand Journal of Psychiatry, 25:277–283.
- Lichtenstein, P., Halldner, L., Zetterqvist, J., Sjölander, A., Serlachius, E., Fazel, S., Långström, N., and Larsson, H. (2012). Medication for attention deficit-hyperactivity disorder and criminality. New England Journal of Medicine, 367:2006–2014.
- Lichter, J. B., Barr, C. L., Kennedy, J. L., Tol, H. H. V., Kidd, K. K., and Livak, K. J. (1993). A Hypervariable Segment in the Human Dopamine Receptor D<sub>4</sub> (DRD4) Gene. *Human Molecular Genetics*, 2:767–773.

- Lochner, L. (2004). Education, Work, and Crime: A Human Capital Approach. International Economic Review, 45(3):811–843.
- Lochner, L. and Moretti, E. (2004). The Effect of Education on Crime: Evidence from Prison Inmates, Arrests, and Self-Reports.

  The American Economic Review, 94(1):155–189.
- Loe, I. M. and Feldman, H. M. (2007). Academic and Educational Outcomes of Children with ADHD. *Journal of Pediatric Association*, 32:643–654.
- Lorentzen, P., McMillan, J., and Wacziarg, R. (2008). Death and Development. Journal of Economic Growth, 13:81–124.
- Lynn, R. and Meisenberg, G. (2010). National IQs Calculated and Validated for 108 Nations. Intelligence, 38:353–360.
- Lynn, R. and Mikk, J. (2009). National IQs predict Educational Attainment in Math, Reading and Science across 56 Nations. Intelligence, 37:305–310.
- Lynn, R. and Vanhanen, T. (2002). IQ and the Wealth of Nations. Greenwood Publishing Group.
- Lynn, R. and Vanhanen, T. (2006). IQ and Global Inequality. Washington Summit Publishers.
- Lynn, R. and Vanhanen, T. (2011). National IQs: A Review of their Educational, Cognitive, Economic, Political, Demographic, Sociological, Epidemiological, Geographic and Climatic Correlates. *Intelligence*, 40:226:234.
- Mannuzza, S. and Klein, R. G. (2000). Long-Term Prognosis in Attention-Deficit/Hyperactivity Disorder. *Child and Adolescent Psychiatric Clinics of North America*, 9:711–726.
- Mannuzza, S., Klein, R. G., Bessler, A., Malloy, P., and LaPadula, M. (1993). Adult Outcome of Hyperactive Boys: Educational Achievement, Occupational Rank, and Psychiatric Status. *Archives of General Psychiatry*, 50:565–576.
- Mauro, P. (1998). Corruption and the Composition of Government Expenditure. Journal of Public Economics, 69:263–279.
- McConaughy, S. H., Volpe, R. J., Antshel, K. M., Gordon, M., and Eiraldi, R. B. (2011). Academic and Social Impairments of Elementary School Children with Attention Deficit Hyperactivity Disorder. *School Psychology Review*, 100:200–225.
- McGough, J. J. (2012). Attention Deficit Hyperactivity Disorder Pharmacogenetics: The Dopamine Transporter and D4 Receptor. *Pharmacogenomics*, 13:365–368.
- Meisenberg, G. and Woodley, M. A. (2013). Are Cognitive Differences Between Countries Diminishing? Evidence from TIMSS and PISA. *Intelligence*, 41:808–816.
- Mendell, M. J. and Heath, G. A. (2005). Do Indoor Pollutants and Thermal Conditions in Schools influence Student Performance?

  A critical Review of the Literature. *Indoor Air*, 15:27–52.
- Minkov, M., Welzel, C., and Bond, M. H. (2016). The Impact of Genes, Geography, and Educational Opportunities on National Cognitive Achievement. *Learning and Individual Differences*, 47:236–243.
- Mocan, N. (2008). What determines Corruption? international Evidence from Microdata. Economic Inquiry, 46(4):493–510.
- Moffitt, T. E., Arseneault, L., Belsky, D., Dickson, N., Hancox, R. J., Harrington, H., Houts, R., Poulton, R., Roberts, B. W., Ross, S., Sears, M. R., Thomson, W. M., and Caspi, A. (2011). A Gradient of Childhood Self-Control Predicts Health, Wealth, and Public Safety. *Proceedings of the National Academy of Sciences*, 108:2693–2698.
- Murphy, K. M., Shleifer, A., and Vishny, R. W. (1991). The Allocation of Talent. The Quarterly Journal of Economics, 106:503-530.

- New, M., Lister, D., Hulme, M., and Makin, I. (2002). A High-Resolution Data Set of Surface Climate Over Global Land Areas.

  Climate Research, 21:1–25.
- Noaín, D., Avale, M. E., Wedemeyer, C., Calvo, D., Peper, M., and Rubinstein, M. (2006). Identification of Brain Neurons Expressing the Dopamine D4 Receptor Gene using BAC Transgenic Mice. *European Journal of Neuroscience*, 24:2429–2438.
- Nordhaus, W. D. (2006). Geography and Macroeconomics: New Data and New Findings. Proceeding of the National Academy of Sciences, 103:3510–3517.
- Nudel, R., Simpson, N. H., Baird, G., O'Hare, A., Conti-Ramsden, G., Bolton, P. F., Hennessy, E. R., The SLI Consortium, Monaco, A. P., Knight, J. C., Winney, B., Fisher, S. E., and Newbury, D. F. (2014). Associations of HLA Alleles with Specific Language Impairment. *Journal of Neurodevelopmental Disorders*, 6:1.
- Okbay, A., Beauchamp, J. P., Fontana, M. A., Lee, J. J., Pers, T. H., Rietveld, C. A., Turley, P., Chen, G.-B., Emilsson, V., Meddens, S. F. W., Oskarsson, S., Pickrell, J. K., Thom, K., Timshel, P., de Vlaming, R., Abdellaoui, A., Ahluwalia, T. S., Bacelis, J., Baumbach, C., Bjornsdottir, G., Brandsma, J. H., Concas, M. P., Derringer, J., Furlotte, N. A., Galesloot, T. E., Girotto, G., Gupta, R., Hall, L. M., Harris, S. E., Hofer, E., Horikoshi, M., Huffman, J. E., Kaasik, K., Kalafati, I. P., Karlsson, R., Kong, A., Lahti, J., van der Lee, S. J., de Leeuw, C., Lind, P. A., Lindgren, K.-O., Liu, T., Mangino, M., Marten, J., Mihailov, E., Miller, M. B., van der Most, P. J., Oldmeadow, C., Payton, A., Pervjakova, N., Peyrot, W. J., Qian, Y., Raitakari, O., Rueedi, R., Salvi, E., Schmidt, B., Schraut, K. E., Shi, J., Smith, A. V., Poot, R. A., St Pourcain, B., Teumer, A., Thorleifsson, G., Verweij, N., Vuckovic, D., Wellmann, J., Westra, H.-J., Yang, J., Zhao, W., Zhu, Z., Alizadeh, B. Z., Amin, N., Bakshi, A., Baumeister, S. E., Biino, G., Bonnelykke, K., Boyle, P. A., Campbell, H., Cappuccio, F. P., Davies, G., De Neve, J.-E., Deloukas, P., Demuth, I., Ding, J., Eibich, P., Eisele, L., Eklund, N., Evans, D. M., Faul, J. D., Feitosa, M. F., Forstner, A. J., Gandin, I., Gunnarsson, B., Halldorsson, B. V., Harris, T. B., Heath, A. C., Hocking, L. J., Holliday, E. G., Homuth, G., Horan, M. A., Hottenga, J.-J., de Jager, P. L., Joshi, P. K., Jugessur, A., Kaakinen, M. A., Kahonen, M., Kanoni, S., Keltigangas-Jarvinen, L., Kiemeney, L. A. L. M., Kolcic, I., Koskinen, S., Kraja, A. T., Kroh, M., Kutalik, Z., Latvala, A., Launer, L. J., Lebreton, M. P., Levinson, D. F., Lichtenstein, P., Lichtner, P., Liewald, D. C. M., Loukola, A., Madden, P. A., Magi, R., Maki-Opas, T., Marioni, R. E., Marques-Vidal, P., Meddens, G. A., McMahon, G., Meisinger, C., Meitinger, T., Milaneschi, Y., Milani, L., Montgomery, G. W., Myhre, R., Nelson, C. P., Nyholt, D. R., Ollier, W. E. R., Palotie, A., Paternoster, L., Pedersen, N. L., Petrovic, K. E., Porteous, D. J., Raikkonen, K., Ring, S. M., Robino, A., Rostapshova, O., Rudan, I., Rustichini, A., Salomaa, V., Sanders, A. R., Sarin, A.-P., Schmidt, H., Scott, R. J., Smith, B. H., Smith, J. A., Staessen, J. A., Steinhagen-Thiessen, E., Strauch, K., Terracciano, A., Tobin, M. D., Ulivi, S., Vaccargiu, S., Quaye, L., van Rooij, F. J. A., Venturini, C., Vinkhuyzen, A. A. E., Volker, U., Volzke, H., Vonk, J. M., Vozzi, D., Waage, J., Ware, E. B., Willemsen, G., Attia, J. R., Bennett, D. A., Berger, K., Bertram, L., Bisgaard, H., Boomsma, D. I., Borecki, I. B., Bultmann, U., Chabris, C. F., Cucca, F., Cusi, D., Deary, I. J., Dedoussis, G. V., van Duijn, C. M., Eriksson, J. G., Franke, B., Franke, L., Gasparini, P., Gejman, P. V., Gieger, C., Grabe, H.-J., Gratten, J., Groenen, P. J. F., Gudnason, V., van der Harst, P., Hayward, C., Hinds, D. A., Hoffmann, W., Hyppnen, E., Iacono, W. G., Jacobsson, B., Jarvelin, M.-R., Jockel, K.-H., Kaprio, J., Kardia, S. L. R., Lehtimaki, T., Lehrer, S. F., Magnusson, P. K. E., Martin, N. G., McGue, M., Metspalu, A., Pendleton, N., Penninx, B. W. J. H., Perola, M., Pirastu, N., Pirastu, M., Polasek, O., Posthuma, D., Power, C., Province, M. A., Samani, N. J., Schlessinger, D., Schmidt, R., Sorensen, T. I. A., Spector, T. D., Stefansson, K., Thorsteinsdottir, U., Thurik, A. R., Timpson, N. J., Tiemeier, H., Tung, J. Y., Uitterlinden, A. G., Vitart, V., Vollenweider, P., Weir, D. R., Wilson, J. F., Wright, A. F., Conley, D. C., Krueger, R. F., Smith, G. D., Hofman, A., Laibson, D. I., Medland, S. E., Meyer, M. N., Yang, J., Johannesson, M., Visscher, P. M., Esko, T., Koellinger, P. D., Cesarini, D., Benjamin, D. J., and Study, L. C. (2016). Genome-Wide Association Study Identifies 74 Loci Associated with Educational Attainment. Nature, 533:539-542.

- Olazagasti, M. A. R., Klein, R. G., Mannuzza, S., Belsky, E. R., Hutchison, J. A., Lashua-Shriftman, E. C., and Castellanos, F. X. (2013). Does Childhood Attention-Deficit/Hyperactivity Disorder Predict Risk-Taking and Medical Illnesses in Adulthood? Journal of the American Academy of Child and Adolescent Psychiatry, 52:153–162.
- Payton, A., Dawes, P., Platt, H., Morton, C. C., Moore, D. R., Massey, J., Horan, M., Ollier, W., Munro, K. J., and Pendleton, N. (2015). A Role for HLA-DRB1\*1101 and DRB1\*0801 in Cognitive Ability and Its Decline with Age. American Journal of Medical Genetics Part B, 171B:209-214.
- Pennington, B. F., Groisser, D., and Welsh, M. C. (1993). Contrasting Cognitive Deficits in Attention Deficit Hyperactivity Disorder Versus Reading Disability. *Developmental Psychology*, 29(3):511–523.
- Polanczyk, G., de Lima, M. S., Horta, B. L., Biederman, J., and Rohde, L. A. (2007). The Worldwide Prevalence of ADHD: A Systematic Review and Metaregression Analysis. *American Journal of Psychatry*, 164(6):942–948.
- Polanczyk, G. V., Willcutt, E. G., Salum, G. A., Kieling, C., and Rohde, L. A. (2014). ADHD Prevalence Estimates Across Three Decades: An Updated Systematic Review and Meta-Regression Analysis. *International Journal of Epidemiology*, 43:434–442.
- Polderman, T. J. C., Boomsma, D. I., Bartels, M., Verhulst, F. C., and Huizink, A. C. (2010). A systematic Review of Prospective Studies on Attention Problems and Academic Achievement. *Acta Psychiatrica Scandinavica*, 122:271–284.
- Pritchett, L. (2001). Where has all the Education gone? World Bank Economic Review, 15(3):367-391.
- Pritchett, L. and Summers, L. H. (1996). Wealthier is Healthier. Journal of Human Resources, 31(4):841–868.
- Prugnolle, F., Manica, A., Charpentier, M., Guégan, J. F., Guernier, V., and Balloux, F. (2005). Pathogen-Driven Selection and Worldwide HLA Class I Diversity. *Current Biology*, 15:1022–1027.
- Psacharopoulos, G. (1994). Returns to Investment in Education: A Global Update. World Development, 22(9):1325–1343.
- Putnam, R. D., Leonardi, R., and Nanetti, R. Y. (1993). Making Democracy Work: Civic Traditions in Modern Italy. Princeton university press.
- Putterman, L. (2008). Agriculture, Diffusion and Development: Ripple Effects of the Neolithic Revolution. Economica, 75:729–748.
- Putterman, L. and Weil, D. N. (2010). Post-1500 Population Flows and the Long Run Determinants of Economic Growth and Inequality. *The Quarterly Journal of Economics*, 125:1627–1682.
- Raggi, V. L. and Chronis, A. M. (2006). Interventions to Address the Academic Impairment of Children and Adolescents with ADHD. Clinical Child and Family Psychology Review, 9(2):85–111.
- Rajeevan, H., Soundararajan, U., Kidd, J. R., Pakstis, A. J., and Kidd, K. K. (2012). ALFRED: An Allele Frequency Ressource for Research and Teaching. *Nucleic Acids Research*, 40:D1010–D1015.
- Ramachandran, S., Deshpande, O., Roseman, C. C., Rosenberg, N. A., Feldman, M. W., and Cavalli-Sforza, L. L. (2005). Support from the Relationship of Genetic and Geographic Distance in Human Populations for a Serial Founder Effect Originating in Africa. Proceedings of the National Academy of Sciences, 44:15942–15947.
- Reinikka, R. and Svensson, J. (2005). Fighting Corruption to Improve Schooling: Evidence from a Newspaper Campaign in Uganda.

  Journal of the European Economic Association, 3(2–3):259–267.

- Rietveld, C. A., Medland, S. E., Derringer, J., Yang, J., Esko, T., Martin, N. W., Westra, H.-J., Shakhbazov, K., Abdellaoui, A., Agrawal, A., Albrecht, E., Alizadeh, B. Z., Amin, N., Barnard, J., Baumeister, S. E., Benke, K. S., Bielak, L. F., Boatman, J. A., Boyle, P. A., Davies, G., de Leeuw, C., Eklund, N., Evans, D. S., Ferhmann, R., Fischer, K., Gieger, C., Gjessing, H. K., Hägg, S., Harris, J. R., Hayward, C., Holzapfel, C., Ibrahim-Verbaas, C. A., Ingelsson, E., Jacobsson, B., Joshi, P. K., Jugessur, A., Kaakinen, M., Kanoni, S., Karjalainen, J., Kolcic, I., Kristiansson, K., Kutalik, Z., Lahti, J., Lee, S. H., Lin, P., Lind, P. A., Liu, Y., Lohman, K., Loitfelder, M., McMahon, G., Vidal, P. M., Meirelles, O., Milani, L., Myhre, R., Nuotio, M.-L., Oldmeadow, C. J., Petrovic, K. E., Peyrot, W. J., Polašek, O., Quaye, L., Reinmaa, E., Rice, J. P., Rizzi, T. S., Schmidt, H., Schmidt, R., Smith, A. V., Smith, J. A., Tanaka, T., Terracciano, A., van der Loos, M. J. H. M., Vitart, V., Völzke, H., Wellmann, J., Yu, L., Zhao, W., Allik, J., Attia, J. R., Bandinelli, S., Bastardot, F., Beauchamp, J., Bennett, D. A., Berger, K., Bierut, L. J., Boomsma, D. I., Bültmann, U., Campbell, H., Chabris, C. F., Cherkas, L., Chung, M. K., Cucca, F., de Andrade, M., De Jager, P. L., De Neve, J.-E., Deary, I. J., Dedoussis, G. V., Deloukas, P., Dimitriou, M., Eiríksdóttir, G., Elderson, M. F., Eriksson, J. G., Evans, D. M., Faul, J. D., Ferrucci, L., Garcia, M. E., Grönberg, H., Guðnason, V., Hall, P., Harris, J. M., Harris, T. B., Hastie, N. D., Heath, A. C., Hernandez, D. G., Hoffmann, W., Hofman, A., Holle, R., Holliday, E. G., Hottenga, J.-J., Iacono, W. G., Illig, T., Järvelin, M.-R., Kähönen, M., Kaprio, J., Kirkpatrick, R. M., Kowgier, M., Latvala, A., Launer, L. J., Lawlor, D. A., Lehtimäki, T., Li, J., Lichtenstein, P., Lichtner, P., Liewald, D. C., Madden, P. A., Magnusson, P. K. E., Mäkinen, T. E., Masala, M., McGue, M., Metspalu, A., Mielck, A., Miller, M. B., Montgomery, G. W., Mukherjee, S., Nyholt, D. R., Oostra, B. A., Palmer, L. J., Palotie, A., Penninx, B. W. J. H., Perola, M., Peyser, P. A., Preisig, M., Räikkönen, K., Raitakari, O. T., Realo, A., Ring, S. M., Ripatti, S., Rivadeneira, F., Rudan, I., Rustichini, A., Salomaa, V., Sarin, A.-P., Schlessinger, D., Scott, R. J., Snieder, H., St Pourcain, B., Starr, J. M., Sul, J. H., Surakka, I., Svento, R., Teumer, A., Tiemeier, H., van Rooij, F. J. A., Van Wagoner, D. R., Vartiainen, E., Viikari, J., Vollenweider, P., Vonk, J. M., Waeber, G., Weir, D. R., Wichmann, H.-E., Widen, E., Willemsen, G., Wilson, J. F., Wright, A. F., Conley, D., Davey-Smith, G., Franke, L., Groenen, P. J. F., Hofman, A., Johannesson, M., Kardia, S. L. R., Krueger, R. F., Laibson, D., Martin, N. G., Meyer, M. N., Posthuma, D., Thurik, A. R., Timpson, N. J., Uitterlinden, A. G., van Duijn, C. M., Visscher, P. M., Benjamin, D. J., Cesarini, D., and Koellinger, P. D. (2013). GWAS of 126,559 Individuals Identifies Genetic Variants Associated with Educational Attainment. Science, 340:1467–1471.
- Rindermann, H. (2007). The g-Factor of International Cognitive Ability Comparisons: The Homogeneity of Results in PISA, TIMSS, PIRLS and IQ-Tests Across Nations. *European Journal of Personality*, 21:667–706.
- Rindermann, H. (2008). Relevance of Education and Intelligence at the National Level for the Economic Welfare of People. Intelligence, 36:127–142.
- Rodriguez, A., Järvelin, M.-R., Obel, C., Taanila, A., Miettunen, J., Moilanen, I., Henriksen, T. B., Pietiläinen, K., Ebeling, H., Kotimaa, A. J., Linnet, K. M., and Olsen, J. (2007). Do Inattention and Hyperactivity Symptoms equal Scholastic Impairment? Evidence from Three European Cohorts. *BMC Public Health*, 7:327.
- Rostowski, J. and Stacescu, B. (2006). The Wig and the Pith Helmet The Impact of "Legal School" versus Colonial Institutions on Economic Performance (Second Version). Technical report, CASE-Center for Social and Economic Research.
- Rouse, C. E. (1998). Private School Vouchers and Student Achievement: An Evaluation of the Milwaukee Parental Choice Program.

  The Quarterly Journal of Economics, 113(2):553–602.
- Sachs, J. D. and Malaney, P. (2002). The Economic and Social Burden of Malaria. Nature, 415:680-685.
- Sachs, J. D. and Warner, A. M. (2001). The Curse of Natural Resources. European Economic Review, 45:827–838.

- Scheffler, R. M., Hinshaw, S. P., Modrek, S., and Levine, P. (2007). The Global Market For ADHD Medications. *Health Affairs*, 26:450–457.
- Schuetz, G., Luedemann, E., West, M. R., and Woessmann, L. (2013). School Accountability, Autonomy, Choice, and the Equality of Educational Opportunities. In Windzio, M., editor, *Integration and Inequality in Educational Institutions*, pages 123–152. Springer, Dordrecht, Heidelberg, New York, London.
- Schuetz, G., Ursprung, H. W., and Woessmann, L. (2008). Education Policy and Equality of Opportunity. Kyklos, 61(2):279–308.
- Selzam, S., Dale, P. S., Wagner, R. K., DeFries, J. C., Cederlöf, M., O'Reilly, P. F., Krapohl, E., and Plomin, R. (2017a). Genome-Wide Polygenic Scores Predict Reading Performance Throughout the School Years. *Scientific Studies of Reading*, 21:334–349.
- Selzam, S., Krapohl, E., von Stumm, S., O'Reilly, P. F., Rimfeld, K., Kovas, Y., Dale, P. S., Lee, J. J., and Plomin, R. (2017b). Predicting Educational Achievement from DNA. *Molecular Psychiatry*, 22:267–272.
- Shakeshaft, N. G., Trzaskowski, M., McMillan, A., Rimfeld, K., Krapohl, E., Haworth, C. M. A., Dale, P. S., and Plomin, R. (2013). Strong Genetic Influence on a UK Nationwide Test of Educational Achievement at the End of Compulsory Education at Age 16. *PLoS One*, 8:e80341.
- Sharkey, P. (2010). The Acute Effect of Local Homicides on Children's Cognitive Performance. Proceedings of the National Academy of Sciences of the United States of America, 107(26):11733–11738.
- Shemyakina, O. (2011). The Effect of Armed Conflict on Accumulation of Schooling: Results from Tajikistan. *Journal of Development Economics*, 95:186–2000.
- Shiina, T., Inoko, H., and Kulski, J. K. (2004). An Update of the HLA Genomic Region, Locus Information and Disease Associations: 2004. *Tissue Antiqens*, 64:631–649.
- Singh, I. (2008). Beyond Polemics: Science and Ethics of ADHD. Nature Reviews Neuroscience, 9:957–964.
- Sniekers, S., Stringer, S., Watanabe, K., Jansen, P. R., Coleman, J. R. I., Krapohl, E., Taskesen, E., Hammerschlag, A. R., Okbay, A., Zabaneh, D., Amin, N., Breen, G., Cesarini, D., Chabris, C. F., Iacono, W. G., Ikram, M. A., Johannesson, M., Koellinger, P., Lee, J. J., Magnusson, P. K. E., McGue, M., Miller, M. B., Ollier, W. E. R., Payton, A., Pendleton, N., Plomin, R., Rietveld, C. A., Tiemeier, H., van Duijn, C. M., and Posthuma, D. (2017). Genome-Wide Association Meta-Analysis of 78,308 Individuals Identifies New Loci and Genes Influencing Human Intelligence. Nature Genetics, 49:1107–1112.
- Stock, J. and Yogo, M. (2005). Testing for Weak Instruments in Linear IV Regression, pages 80–108. Cambridge University Press, New York.
- Stoet, G. and Geary, D. C. (2017). Students in Countries with higher Levels of Religiosity perform lower in Science and Mathematics.

  Intelligence, 62:71–78.
- Strulik, H. and Werner, K. (2016). 50 Is The New 30 Long-Run Trends of Schooling and Retirement explained by Human Aging. *Journal of Economic Growth*, 21:165–187.
- Swanson, J. M., Kinsbourne, M., Nigg, J., Lanphear, B., Stefanatos, G. A., Volkow, N., Taylor, E., Casey, B. J., Castellanos, F. X., and Wadhwa, P. D. (2007). Etiological Subtypes of Attention-Deficit/Hyperactivity Disorder: Brain Imaging, Molecular Genetic and Environmental Factors and the Dopamine Hypothesis. *Neuropsychology Review*, 17:39–59.

- Tate, J. E., Burton, A. H., Boschi-Pinto, C., Steele, A. D., Duque, J., Parashar, U. D., and S, W. C. G. R. (2012). 2008 Estimate of Worldwide Rotavirus-Associated Mortality in Children younger than 5 years before the Introduction of Universal Rotavirus Vaccination Programmes: A Systematic Review and Meta-Analysis. LANCET Infectious Diseases, 12(2):136–141.
- Thomas, R., Sanders, S., Doust, J., Beller, E., and Glasziou, P. (2015). Prevalence of Attention-Deficit/Hyperactivity Disorder: A Systematic Review and Meta-Analysis. *Pediatrics*, 135(4):1–8.
- Thursz, M. R., Thomas, H. C., Greenwood, B. M., and Hill, A. V. S. (1997). Heterozygote Advantage for HLA Class-II Type in Hepatitis B Virus Infection. *Nature Genetics*, 17:11–12.
- Timimi, S. and Taylor, E. (2004). Adhd is Best Understood as a Cultural Construct. British Journal of Psychiatry, 184:8-9.
- Todd, P. E. and Wolpin, K. I. (2003). On the Specification and Estimation of the Production Function for Cognitive Achievement.

  The Economic Journal, 113(485):F3-F33.
- Todd, P. E. and Wolpin, K. I. (2007). The Production of Cognitive Achievement in Children: Home, School, and Racial Test Score Gaps. *Journal of Human Capital*, 1(1):91–136.
- Trachtenberg, E., Korber, B., Sollars, C., Kepler, T. B., Hraber, P. T., Hayes, E., Funkhouser, R., Fugate, M., Theiler, J., Hsu, Y. S., Kunstman, K., Wu, S., Phair, J., Erlich, H., and Wolinsky, S. (2003). Advantage of Rare HLA Supertype in HIV Disease Progression. *Nature Medicine*, 9:928–935.
- Van Tol, H. H. M., Bunzow, J. R., Guan, H.-C., Sunahara, R. K., Seeman, P., Niznik, H. B., and Civelli, O. (1991). Cloning of the Gene for a Human Dopamine D<sub>4</sub> Receptor with High Affinity for the Antipsychotic Clozapine. *Nature*, 350:610–614.
- Van Tol, H. H. M., Wu, C. M., Guan, H.-C., Ohara, K., Bunzow, J. R., Civelli, O., Kennedy, J., Seeman, P., Niznik, H. B., and Jovanovic, V. (1992). Multiple Dopamine D4 Receptor Variants in the Human Population. *Nature*, 358:149–152.
- Volken, T. (2003). IQ and the Wealth of Nations: A Critique of Richard Lynn and Tatu Vanhanen's Recent Book. European Sociological Review, 19(4):411–412.
- Walker, S. P., Wachs, T. D., Gardner, J. M., Lozoff, B., Wasserman, G. A., Pollitt, E., Carter, J. A., and the International Child Development Steering Group (2007). Child Development: Risk Factors for Adverse Outcomes in Developing Countries. *LANCET*, 369:145–157.
- Wang, E., Ding, Y.-C., Flodman, P., Kidd, J. R., Kidd, K. K., Grady, D. L., Ryder, O. A., Spence, M. A., Swanson, J. M., and Moyzis, R. K. (2004). The Genetic Architecture of Selection at the Human Dopamine Receptor D4 (DRD4) Gene Locus. American Journal of Human Genetics, 74:931–944.
- Wig, T., Hegre, H., and Regan, P. M. (2015). Updated Data on Institutions and Elections 1960–2012: Presenting the IAEP Dataset Version 2.0. Research & Politics, 2(2):1–11.
- Wilcox, A. J. (2001). On the Importance and the Unimportance of Birthweight. *International Journal of Epidemiology*, 30:1233–1241.
- Willcutt, E. G. (2012). The Prevalence of DSM-IV Attention-Deficit/Hyperactivity Disorder: A Meta-Analytic Review. *Neurotherapeutics*, 9:490–499.

- Willcutt, E. G., Nigg, J. T., Pennington, B. F., Solanto, M. V., Rohde, L. A., Tannock, R., Loo, S. K., Carlson, C. L., McBurnett, K., and Lahey, B. B. (2012). Validity of DSM–IV Attention Deficit/Hyperactivity Disorder Symptom Dimensions and Subtypes. Journal of Abnormal Psychology, 121:991–1010.
- Williams, J. and Taylor, E. (2006). The Evolution of Hyperactivity, Impulsivity and Cognitive Diversity. *Journal of the Royal Society Interface*, 3:399–413.
- Wolff, P. H., Tesfai, B., Egasso, H., and Aradom, T. (1995). The Orphans of Eritrea: A Comparison Study. *Journal of Child Psychology and Psychiatry*, 36(4):633–644.
- Wolraich, M. L., Hannah, J. N., Pinnock, T. Y., Baumgaertel, A., and Brown, J. (1996). Comparison of Diagnostic Criteria for Attention-Deficit Hyperactivity Disorder in a County-Wide Sample. *Journal of the American Academy of Child and Adolescent Psychiatry*, 35:319–324.
- Wooldridge, J. M. (2010). Econometric Analysis of Cross Section and Panel Data. The MIT Press, Cambridge, Massachusetts, second edition.
- Wright, S. P., Horn, S. P., and Sanders, W. L. (1997). Teacher and Classroom Context Effect on Student Achievement: Implications for Teacher Evaluation. *Journal of Personnel Evaluation in Education*, 11:57–67.
- Wu, J., Xiao, H., Sun, H., Zou, L., and Zhu, L.-Q. (2012). Role of Dopamine Receptors in ADHD: A Systematic Meta-Analysis.
  Molecular Neurobiology, 45:605–620.
- Zabaneh, D., Krapohl, E., Simpson, M. A., Miller, M. B., Iacono, W. G., McGue, M., Putallaz, M., Lubinski, D., Plomin, R., and Breen, G. (2016). Fine Mapping Genetic Associations Between the HLA Region and Extremely High Intelligence. Scientific Reports, 7:41182.

## A Regression Tables

Table 1: Meta-Regression of ADHD Prevalence Estimates (Assessing Study Methodological Characteristics)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Risk of Bias	DSM	Population	Full vs. Partial	Type of ADHD	Informant	Time	Region
	Score	Edition	Sample	ADHD Criteria	Measurement	of ADHD	Fixed Effects	Fixed Effects
	Dependent Vari	able: Study-Level	Estimated ADHD	Prevalence Rate				
			Ref	erence Category:	Low Risk of Bias	Score		
Very High Risk of Bias Score	-0.0316***	-0.0358***	-0.0373***	-0.0372***	-0.0352***	-0.0349***	-0.0310***	-0.0023
	(0.0056)	(0.0043)	(0.0030)	(0.0032)	(0.0036)	(0.0030)	(0.0049)	(0.0080)
High Risk of Bias Score	-0.0345**	-0.0393***	-0.0297***	-0.0301***	-0.0202***	-0.0137***	-0.0138***	-0.0117***
	(0.0143)	(0.0142)	(0.0095)	(0.0103)	(0.0073)	(0.0044)	(0.0038)	(0.0036)
Moderate Risk of Bias Score	-0.0319***	-0.0343***	-0.0258***	-0.0256***	-0.0196***	-0.0204***	-0.0220***	-0.0166***
	(0.0088)	(0.0105)	(0.0070)	(0.0069)	(0.0069)	(0.0060)	(0.0072)	(0.0050)
			Refere	ence Category: D	SM-IV Diagnostic	Criteria		
OSM-III Diagnostic Criteria		-0.0100	-0.0051	-0.0082	-0.0112	-0.0067	-0.0083	-0.0061
		(0.0127)	(0.0119)	(0.0120)	(0.0099)	(0.0119)	(0.0171)	(0.0204)
DSM-III-R Diagnostic Criteria		-0.0227***	-0.0194***	-0.0212***	-0.0253***	-0.0209**	-0.0218	-0.0129
		(0.0083)	(0.0069)	(0.0051)	(0.0058)	(0.0079)	(0.0133)	(0.0150)
			F	Reference Catego	y: Schooling Sam	ple		
Community Sample			0.0182*	0.0171	0.0279***	0.0159**	0.0119	0.0106
-			(0.0103)	(0.0114)	(0.0102)	(0.0063)	(0.0094)	(0.0079)
Whole Population			0.0191**	0.0198**	0.0376***	0.0169	0.0138	0.0096
•			(0.0081)	(0.0085)	(0.0088)	(0.0125)	(0.0103)	(0.0094)
			,	ce Category: Full	,	,	()	()
Full Diagnostic Criteria Met				0.0046	0.0278***	0.0301**	0.0340***	0.0256*
				(0.0124)	(0.0100)	(0.0116)	(0.0109)	(0.0137)
			R	Reference Categor	y: Diagnostic Rep	ort	,	,
nterview					-0.0122	-0.0149	-0.0199	-0.0130
					(0.0155)	(0.0158)	(0.0194)	(0.0199)
Symptom-only Checklist					0.0429***	0.0597***	0.0601***	0.0455***
					(0.0125)	(0.0103)	(0.0116)	(0.0091)
Jnclear					0.0268	0.0523***	0.0602***	0.0480***
					(0.0193)	(0.0164)	(0.0154)	(0.0153)
				Reference Cate	,	(0.0202)	(0.0-0-)	(010200)
Child					-87:	-0.0336**	-0.0319**	-0.0199
						(0.0133)	(0.0137)	(0.0189)
Parent						0.0126	0.0160	0.0041
						(0.0123)	(0.0107)	(0.0088)
reacher reacher						-0.0218*	-0.0232	-0.0192
						(0.0114)	(0.0163)	(0.0153)
AND Rule						-0.0504***	-0.0520***	-0.0505***
						(0.0111)	(0.0112)	(0.0130)
OR Rule						-0.0379***	-0.0406***	-0.0415***
or reac						(0.0085)	(0.0146)	(0.0116)
Unclear						-0.0483*	-0.0513*	-0.0404**
O 1101001						(0.0257)	(0.0286)	(0.0192)
Number of ADHD Studies	179	179	179	179	179	179	179	179
Number of ADHD Studies $R^2$	0.19	0.22	0.27	0.27	0.38	0.48	0.50	0.65
Time Fixed Effects	No	No	No	No	No	No	Yes	Yes
THE TIME THECES	110	110	110	110	110	110	100	1.02

Notes: This table shows the association between estimated ADHD prevalence rates and various study-level methodological characteristics in a sample of 179 individual ADHD studies conducted in 52 countries during the years 1981 to 2014. The method of estimation refers to a weighted meta-regression model where study-level observations are weighted according to their reported sample size. Time Fixed Effects refer to a set of indicator variables that take a value of 1 if the respective study was published within a given five-year interval (i.e., 1980–1984, 1985–1989,..., and 2010–2014). Region Fixed Effects refer to a set of regional controls for individual ADHD studies conducted in North America, Latin America and the Caribbean, Africa, Asia, the Middle East, and Europe. See the main text for additional details on data construction and sources. Constant term included but not shown.

Standard errors, clustered with respect to countries, are reported in parentheses.

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 2: Association Between the Country-Level ADHD Prevalence Estimate and DRD4 Exon III 2- and 7-Repeat Allele Frequency Measure

		Ü				I	1 1 1	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Baseline	Legal Origin	Major Religion	Economic	Health	Educational	State	Neolithic
	Specification	Effects	Effects	Development	System	System	Experience	Transition
	Dependent Vari	able: Country-Leve	el ADHD Prevalenc	ce Estimate				
$DRD4^{R2R7}$	0.2197**	0.3644***	0.4188***	0.4219***	0.4226***	0.4302**	0.4339**	0.4218**
(Ethnicity-Weighted)	(0.1036)	(0.1311)	(0.1286)	(0.1509)	(0.1556)	(0.1740)	(0.1617)	(0.1654)
ln Real GDP per Capita				0.0010	-0.0001	0.0089	0.0007	0.0026
				(0.0130)	(0.0150)	(0.0178)	(0.0195)	(0.0203)
Life Expectancy					0.0002	0.0025	0.0046	0.0041
					(0.0020)	(0.0022)	(0.0030)	(0.0037)
ln Human Capital Index						-0.1210	-0.1612	-0.1582
						(0.0980)	(0.1047)	(0.1103)
ln State History 1 AD – 1500 AD							-0.0269	-0.0306
(Ancestry-Adjusted)							(0.0200)	(0.0206)
ln Agricultural Transition Timing								0.0188
(Ancestry-Adjusted)								(0.0422)
Number of Countries	42	42	42	42	42	42	42	42
$R^2$	0.40	0.45	0.45	0.45	0.45	0.48	0.53	0.54
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Legal Origin Effects	No	Yes	Yes	Yes	Yes	Yes Yes		Yes
Major Religion Effects	No	No	Yes	Yes	Yes	Yes	Yes	Yes

Notes: The dependent variable is the country-level ADHD prevalence estimate calculated as the weighted average across individual ADHD studies within countries, where each study-level observation is weighted according to its reported sample size.  $DRD4^{R2R7}$  refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure (ethnicity-weighted). In Real GDP per Capita refers to the natural logarithm of real GDP per capita. Life Expectancy refers to the country's overall life expectancy at birth. In Human Capital Index refers to the natural logarithm of a country's human capital index, which combines data on average years of schooling and information on the assumed rate of return to educational years adjusted for decreasing returns. In State History refers to the natural logarithm of the country's experience with early state formation from 1 AD to 1500 AD (ancestry-adjusted). In Agricultural Transition Timing refers to the natural logarithm of the time elapsed between when the country began practicing sedentary agriculture and the year 2000 (ancestry-adjusted). Region Effects refer to region controls for North America, Latin America and the Caribbean, Africa, East Asia, Central Asia, South Asia, Middle East, Europe, and Oceania. Legal Origin Effects refer to a set of indicator variables that take a value of 1 if the country's legal tradition is British, French, German, or Scandinavian. Major Religion Effects refer to a set of country-level controls indicating the share of Muslims, Catholics, and Protestants in the population (measured in percent). See the main text for additional details on data construction and sources. Constant term included but not shown.

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 3: DRD4 III 2- and 7-Repeat Allele Frequency Measure and Student Achievement Test Score (Biogeographic and Historical Factors)

	(1)	(2)	(3)	(4)	(5)	(6)
	Baseline	Absolute	Climate	Percentage	Major Religion	Legal Origin
	Specification	Latitude	Effects	Arable Land	Effects	Effects
			ent Achievement Tes			
$DRD4^{R2R7}$	-2.8806***	-2.7416**	-2.3051**	-2.2348**	-3.6139***	-3.8765***
(Ethnicity-Weighted)	(0.8839)	(1.1567)	(1.0241)	(1.0396)	(1.1089)	(1.1361)
ln Absolute Latitude		0.0312	-0.0042	0.0086	-0.0371	-0.0906
		(0.1477)	(0.1714)	(0.1744)	(0.1706)	(0.1821)
Population in Area with Malaria			-0.1244	-0.1324	-0.1358	-0.0370
			(0.2422)	(0.2458)	(0.2731)	(0.2809)
Temperature			-0.0056	-0.0045	0.0001	-0.0046
			(0.0120)	(0.0119)	(0.0125)	(0.0141)
ln Percentage Arable Land Area				-0.0120	0.0047	0.0026
				(0.0392)	(0.0419)	(0.0401)
Percentage of Protestants					0.0060***	0.0054
					(0.0022)	(0.0037)
Percentage of Catholics					0.0039*	0.0022
					(0.0020)	(0.0020)
Percentage of Muslims					-0.0008	-0.0019
_					(0.0026)	(0.0027)
British Legal Origin					, ,	0.1314
						(0.1842)
French Legal Origin						0.2511
						(0.1600)
German Legal Origin						0.3861**
						(0.1754)
Scandinavian Legal Origin						0.0075
0 0						(0.2901)
Number of Countries	81	81	81	81	81	81
$R^2$	0.69	0.69	0.69	0.69	0.73	0.75
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
OPEC Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
TIMSS Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes

Notes: The dependent variable is the country-level student achievement test score outcome in the academic subjects mathematics and science. This variable has been constructed from individual student achievement test score outcomes from the PISA and TIMSS studies during the various student assessment waves in the years 1995 to 2015.  $DRD4^{R2R7}$  refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure (ethnicity-weighted). In Absolute Latitude refers to the natural logarithm of the absolute value of a country's approximate centroid latitude in decimal degrees. Population in Area with Malaria is the share of a country's population in 1995 residing in areas with malaria falciparum. Temperature is the country's mean temperature value (in degrees Celsius) during the period 1960 to 1990. In Percentage Arable Land Area refers to the natural logarithm of a country's percentage arable land area. Major Religion Effects refer to a set of country-level controls indicating the share of Muslims, Catholics, and Protestants in the population (measured in percent). Legal Origin Effects refer to a set of indicator variables that takes a value of 1 if the country's legal tradition is British, French, German, or Scandinavian. Region Fixed Effects refer to region controls for North America, Latin America and the Caribbean, Africa, East Asia, Central Asia, South Asia, Middle East, Europe, and Oceania. OPEC Fixed Effects refer to an indicator variable for member countries belonging to the Organization of the Petroleum Exporting Countries (OPEC). TIMSS Fixed Effects take a value of 1 if the student achievement test score outcome was derived from the TIMSS study and zero otherwise. See the main text for additional details on data construction and sources. Constant term included but not shown.

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 4: DRD4 III 2- and 7-Repeat Allele Frequency Measure and Student Achievement Test Score (Health-Related Factors)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Baseline	Low Birth	Infant	Life	Disability-Adjusted	Homicide	Health-Related
	Specification	Weight	Mortality	Expectancy	Life-Years	Rate	Factors
	Dependent Vari	able: Aggregate S	tudent Achievemer	nt Test Score			
$DRD4^{R2R7}$	-3.8765***	-3.6536***	-3.3784***	-3.2673***	-3.6231***	-4.2019***	-3.6049***
(Ethnicity-Weighted)	(1.1361)	(1.1311)	(0.9943)	(1.0611)	(1.0861)	(1.1830)	(1.0217)
Low Birth Weight		-0.0452**					-0.0038
		(0.0217)					(0.0236)
Infant Mortality Rate			-0.0284***				-0.0252***
			(0.0058)				(0.0092)
Life Expectancy				0.0665***			-0.0004
				(0.0172)			(0.0300)
Disability-Adjusted Life-Years					-2.2591***		-0.3429
					(0.8223)		(0.8738)
Homicide Rate						-0.0113*	-0.0078*
						(0.0061)	(0.0041)
Number of Countries	81	81	81	81	81	81	81
$R^2$	0.75	0.77	0.84	0.81	0.79	0.77	0.85
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
OPEC Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
TIMSS Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Biogeographic Effects	Effects Yes Yes		Yes	Yes	Yes	Yes	Yes
Major Religion Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Legal Origin Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Notes: The dependent variable is the country-level student achievement test score outcome in the academic subjects mathematics and science. This variable has been constructed from individual student achievement test score outcomes from the PISA and TIMSS studies during the various student assessment waves in the years 1995 to 2015.  $DRD4^{R2R7}$  refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure (ethnicity-weighted).  $Low\ Birth\ Weight\ refers$  to the share of newborns with a birth weight below 2,500 grams.  $Infant\ Mortality\ Rate\ refers$  to the number of infant deaths within one year per 1,000 live births.  $Life\ Expectancy\ refers$  to the country's overall life expectancy at birth.  $Disability\-Adjusted\ Life\-Years\ refer$  to the country's healthy life years lost due to disabilities, diseases, or deaths.  $Homicide\ Rate\ refers$  to the number of intentional homicides per 100,000 inhabitants of a country.  $Region\ Fixed\ Effects\ refer$  to region controls for North America, Latin America and the Caribbean, Africa, East Asia, Central Asia, South Asia, Middle East, Europe, and Oceania.  $OPEC\ Fixed\ Effects\ refer$  to an indicator variable for member countries belonging to the Organization of the Petroleum Exporting Countries (OPEC).  $TIMSS\ Fixed\ Effects\ take\ a\ value\ of\ 1\ if\ the\ student\ achievement\ test\ score\ outcome\ was\ derived from the TIMSS\ study\ and\ zero\ otherwise. <math>Biogeographic\ Effects\ include\ ln\ Absolute\ Latitude\ Population\ in\ Area\ with\ Malaria,\ Temperature\ and\ ln\ Percentage\ Arable\ Land\ Area.\ Major\ Religion\ Effects\ refer\ to\ a\ set\ of\ country-level\ controls\ indicating\ the\ share\ of\ Muslims,\ Catholics,\ and\ Protestants\ in\ the\ population\ (measured\ in\ percent).\ Legal\ Origin\ Effects\ refer\ to\ a\ set\ of\ indicator\ variables\ that\ takes\ a\ value\ of\ 1\ if\ the\ country's\ legal\ tradition\ is\ British,\ French,\ German,\ or\ Scandinavian.\ See\ the\ main\ text\ for\ additional\ details\ on\ data\ construction\$ 

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 5: DRD4 III 2- and 7-Repeat Allele Frequency Measure and Student Achievement Test Score (Economic Environment Factors)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Baseline	GDP per	Government	Fertility	Population	Trade	Economic	Health-Related
	Specification	Capita	Consumption	Rate	Growth	Openness	Environment	Factors
	Dependent Vari	able: Aggregate S	tudent Achievemen	t Test Score				
$DRD4^{R2R7}$	-3.8765***	-2.5740**	-3.7060***	-3.0536***	-4.0503***	-3.6383***	-2.7301***	-3.1287***
(Ethnicity-Weighted)	(1.1361)	(1.1515)	(1.1811)	(0.9907)	(1.1203)	(1.0032)	(0.9876)	(0.9881)
ln GDP per Capita		0.3374***					0.1418	0.0707
		(0.1194)					(0.1869)	(0.2036)
Government Consumption			-1.1979				-1.0378	-0.7169
(Percentage of GDP)			(1.4911)				(1.5954)	(1.5266)
Total Fertility Rate				-0.3834***			-0.1978	-0.1089
				(0.1018)			(0.1354)	(0.1303)
Population Growth					-0.1540**		-0.1515**	-0.0965
					(0.0618)		(0.0570)	(0.0651)
Trade Openness						0.0024*	0.0020	0.0011
(Percentage of GDP)						(0.0013)	(0.0014)	(0.0012)
Number of Countries	81	81	81	81	81	81	81	81
$R^2$	0.75	0.79	0.76	0.80	0.77	0.77	0.83	0.86
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
OPEC Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
TIMSS Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Biogeographic Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Major Religion Effects	Yes	Yes	Yes	Yes Yes		Yes	Yes	Yes
Legal Origin Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Health-Related Effects	No	No	No	No	No	No	No	Yes

Notes: The dependent variable is the country-level student achievement test score outcome in the academic subjects mathematics and science. This variable has been constructed from individual student achievement test score outcomes from the PISA and TIMSS studies during the various student assessment waves in the years 1995 to 2015.  $DRD4^{R2R7}$  refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure (ethnicity-weighted). In Real GDP per Capita refers to the natural logarithm of real GDP per capita. Government Consumption refers to the share of government consumption measured in percentage of total GDP. Total Fertility Rate refers to the country's average number of children expected to be born to a woman over her entire life course. Population Growth refers to the country's average population growth rate (measured in percent). Trade Openness refers to the country's amount of exports and imports over total GDP (measured in percent). Region Fixed Effects refer to region controls for North America, Latin America and the Caribbean, Africa, East Asia, Central Asia, South Asia, Middle East, Europe, and Oceania. OPEC Fixed Effects refer to an indicator variable for member countries belonging to the Organization of the Petroleum Exporting Countries (OPEC). TIMSS Fixed Effects take a value of 1 if the student achievement test score outcome was derived from the TIMSS study and zero otherwise. Biogeographic Effects include In Absolute Latitude, Population in Area with Malaria, Temperature, and In Percentage Arable Land Area. Major Religion Effects refer to a set of country-level controls indicating the share of Muslims, Catholics, and Protestants in the population (measured in percent). Legal Origin Effects refer to a set of indicator variables that takes a value of 1 if the country's legal tradition is British, French, German or Scandinavian. Health-Related Effects include Low Birth Weight, Infant Mortality Rate, Life Expectancy, Disability-Adjusted Life-Years, and Homicide Rate. See the main t

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 6: DRD4 III 2- and 7-Repeat Allele Frequency Measure and Student Achievement Test Score (Social Value Factors)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	Baseline	Female Political	Female Economic	Government	Property Rights	Freedom from	Number of	Social Value	Health-Related
	Specification	Empowerment	Empowerment	Structure	Protection	Corruption	Civil Conflicts	Factors	Factors
	Dependent Vari	able: Aggregate Stu	dent Achievement Te	est Score					
$DRD4^{R2R7}$	-3.8765***	-3.9575***	-3.8367***	-3.9037***	-3.4152***	-3.2612***	-3.5968***	-3.4752***	-3.6959***
(Ethnicity-Weighted)	(1.1361)	(1.1526)	(1.1686)	(1.1238)	(1.0941)	(1.0320)	(1.1388)	(1.0236)	(1.0802)
Women in Parliament		0.0027						0.0047	0.0045
(Proportion of Seats)		(0.0095)						(0.0076)	(0.0067)
Female Economic Participation			-0.0054					-0.0091	0.0084
(Share of Labor Force)			(0.0121)					(0.0106)	(0.0145)
Federalist Political System				0.1749*				0.1554*	0.0695
				(0.0908)				(0.0910)	(0.0771)
Property Rights Protection					0.0129***			0.0065	0.0061
(0 = low to 100 = high)					(0.0036)			(0.0068)	(0.0067)
Freedom from Corruption						0.0148***		0.0106	0.0031
(0 = low to 100 = high)						(0.0039)		(0.0070)	(0.0062)
ln Number of Civil Conflicts							-0.0175	0.0108	0.0129
							(0.0185)	(0.0177)	(0.0171)
Number of Countries	81	81	81	81	81	81	81	81	81
$R^2$	0.75	0.75	0.75	0.76	0.80	0.80	0.76	0.82	0.86
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
OPEC Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
TIMSS Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Biogeographic Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Major Religion Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Legal Origin Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Health-Related Effects	No	No	No	No	No	No	No	No	Yes

Notes: The dependent variable is the country-level student achievement test score outcome in the academic subjects mathematics and science. This variable has been constructed from individual student achievement test score outcomes from the PISA and TIMSS studies during the various student assessment waves in the years 1995 to 2015. DRD4\*R2\*R7 refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure (ethnicity-weighted). Women in Parliament refers to the proportion of seats held by women in national parliaments. Female Economic Participation refers to the percentage of women in the total labor force. Federal Political System refers to an indicator variable for a federalist government system. Property Rights Protection refers to an index of governmental and legislative efforts to protect private property. Freedom from Corruption refers to the country's freedom from corruption index. In Number of Civil Conflicts refers to the country's average number of internal (i.e., civil) conflict events. Region Fixed Effects refer to region controls for North America, Latin America and the Caribbean, Africa, East Asia, Central Asia, South Asia, Middle East, Europe, and Oceania. OPEC Fixed Effects refer to an indicator variable for member countries belonging to the Organization of the Petroleum Exporting Countries (OPEC). TIMSS Fixed Effects take a value of 1 if the student achievement test score outcome was derived from the TIMSS study and zero otherwise. Biogeographic Effects include the Absolute Latitude, Population in Area with Malaria, Temperature, and In Percentage Arable Land Area. Major Religion Effects refer to a set of country-level controls indicating the share of Muslims, Catholics, and Protestants in the population (measured in percent). Legal Origin Effects refer to a set of indicator variables that takes a value of 1 if the country's legal tradition is British, French, German, or Scandinavian. Health-Related Effects include Low Birth Weight, Infant Mortality Rate, Life Expectancy, Disab

<sup>\*:</sup> Significant at the 10% level. \*\*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 7: DRD4 III 2- and 7-Repeat Allele Frequency Measure and Student Achievement Test Scores (Educational Factors)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	Baseline	No	Primary	Secondary	Human	Student-Teacher	Educational	Private	Educational	Health-Related
	Specification	Education	Education	Education	Capital	Ratio	Expenditures	Enrollment	Factors	Factors
	Dependent Va	riable: Aggregat	e Student Achiev	vement Test Scor	e					
$DRD4^{R2R7}$	-3.5286***	-3.5475***	-3.4669***	-3.4787***	-3.5627***	-3.5658***	-3.3426**	-3.5305***	-3.4990**	-3.0533***
(Ethnicity-Weighted)	(1.2096)	(1.2597)	(1.2178)	(1.1835)	(1.2114)	(1.2415)	(1.2725)	(1.2237)	(1.3584)	(1.0223)
No Education		0.0016							0.0156	0.0318***
(Percentage of Population)		(0.0080)							(0.0105)	(0.0098)
Primary Education			0.0025						0.0085	0.0087
(Percentage of Population)			(0.0058)						(0.0069)	(0.0065)
Secondary Education				-0.0024					-0.0002	0.0085
(Percentage of Population)				(0.0069)					(0.0091)	(0.0087)
ln Human Capital Index					0.1486				1.0904	-0.2441
					(0.6533)				(0.7833)	(0.6422)
Student-Teacher Ratio						-0.0077			0.0035	0.0082
(Secondary Education)						(0.0142)			(0.0146)	(0.0114)
Educational Expenditures							0.0821		0.1036	0.0839
							(0.0803)		(0.0922)	(0.0610)
Private Enrollment Rate								0.0003	0.0003	0.0012
(Primary Education, in Percent)								(0.0031)	(0.0032)	(0.0045)
Number of Countries	73	73	73	73	73	73	73	73	73	73
$R^2$	0.78	0.78	0.78	0.78	0.78	0.78	0.79	0.78	0.79	0.91
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
OPEC Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
TIMSS Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Biogeographic Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Major Religion Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Legal Origin Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Health-Related Effects	No	No	No	No	No	No	No	No	No	Yes

Notes: The dependent variable is the country-level student achievement test score outcome in the academic subjects mathematics and science. This variable has been constructed from individual student achievement test score outcomes from the PISA and TIMSS studies during the various student assessment waves in the years 1995 to 2015. DRD4R2R7 refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure (ethnicity-weighted). No Education refers to the country's percentage of population aged 15 and above with primary educational attainment. Brimary Education refers to the country's percentage of population aged 15 and above with secondary Education refers to the country's percentage of population aged 15 and above with secondary educational attainment. In Human Capital Index refers to the natural logarithm of a country's human capital index, which combines data on average years of schooling and information Student-Teacher Ratio refers to the average classroom size in secondary education, i.e., the number of students per trained teacher. Educational Expenditures refer to the country's percentage of GDP devoted to educational services. Private Enrollment Rate refers to the country's share of privately enrolled students in primary education. Region Fixed Effects refer to region controls for North America, Latin America and the Caribbean, Africa, East Asia, Central Asia, South Asia, Middle East, Europe, and Oceania. OPEC Fixed Effects refer to an indicator variable for member countries belonging to the Organization of the Petroleum Exporting Countries (OPEC). TIMSS Fixed Effects take a value of 1 if the student achievement test score outcome was derived from the TIMSS study and zero otherwise. Biogeographic Effects include In Absolute Latitude, Population in Area with Malaria, Temperature, and In Percentage Arable Land Area. Major Religion Effects refer to a set of country-level controls indicating the share of Muslims, Catholics, and Protestants in the population (measured in percent). Legal Orig

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 8: DRD4 III 2- and 7-Repeat Allele Frequency Measure and Student Achievement Test Score (Genetic and Diversity Factors)

	(1)	(2)	(3)	(4)	(5)	(6)
	Baseline	Genetic	HLA Genetic	Ethnic	Religious	Genetic and
	Specification	Diversity	Diversity	Diversity	Diversity	Diversity Factors
	Dependent Varia	ble: Aggregate Stud	lent Achievement Test	Score		
$DRD4^{R2R7}$	-3.8765***	-4.4028***	-3.8706***	-3.4633***	-3.8991***	-3.9026***
(Ethnicity-Weighted)	(1.1361)	(1.1370)	(1.2064)	(1.1519)	(1.1233)	(1.1942)
Predicted Genetic Diversity		-4.9385				-5.9396
(Ancestry-Adjusted)		(3.8201)				(5.1445)
HLA Diversity			0.0572			2.4985
(Ethnicity-Weighted)			(3.3450)			(4.9590)
Ethnic Diversity				-0.4882*		-0.4554
				(0.2498)		(0.2730)
Religious Diversity					0.0523	0.0811
					(0.2368)	(0.2323)
Number of Countries	81	81	81	81	81	81
$R^2$	0.75	0.76	0.75	0.77	0.75	0.78
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
OPEC Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
TIMSS Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes
Biogeographic Effects	Yes	Yes	Yes	Yes	Yes	Yes
Major Religion Effects	Yes	Yes	Yes	Yes	Yes	Yes
Legal Origin Effects	Yes	Yes	Yes	Yes	Yes	Yes

Notes: The dependent variable is the country-level student achievement test score outcome in the academic subjects mathematics and science. This variable has been constructed from individual student achievement test score outcomes from the PISA and TIMSS studies during the various student assessment waves in the years 1995 to 2015.  $DRD4^{R2R7}$  refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure (ethnicity-weighted). Predicted Genetic Diversity refers to a country-level measure of overall genetic diversity (ancestry-adjusted). HLA Diversity refers to a country-level measure of genetic diversity associated with a set of genes located in the HLA region (ethnicity-weighted). Ethnic Diversity refers to a country-level measure of ethno-linguistic diversity. Religious Diversity refers to a country-level measure of religious diversity. Region Fixed Effects refer to region controls for North America, Latin America and the Caribbean, Africa, East Asia, Central Asia, South Asia, Middle East, Europe, and Oceania. OPEC Fixed Effects refer to an indicator variable for member countries belonging to the Organization of the Petroleum Exporting Countries (OPEC). TIMSS Fixed Effects take a value of 1 if the student achievement test score outcome was derived from the TIMSS study and zero otherwise. Biogeographic Effects include In Absolute Latitude, Population in Population Population

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 9: DRD4 III 2- and 7-Repeat Allele Frequency Measure and Student Achievement Test Score (Stratified Analysis According to PISA Subjects)

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	Baseline	Health-Related	Economic	Social Value	Schooling	Educational	Health-Related
	Specification	Factors	Environment	Factors	Sample	Factors	Factors
	Devel A Deve			Ashimum and Treat C			
$DRD4^{R2R7}$	-4.2957***	ndent Variable: Aggre -5.8645***	-5.9801***	-6.1522***	-4.1031***	-3.9787***	-6.5177**
(Ethnicity-Weighted)	(1.1115)	(1.8266)	(1.7623)	(1.5758)	(1.2278)	(1.3475)	(2.3417)
Number of Countries	63	63	63	63	57	57	57
$\mathbb{R}^2$	0.83	0.88	0.91	0.92	0.84	0.85	0.92
	Panel B – Deper	ndent Variable: Aggre	egate PISA Student	Achievement Mathe	matics Score		
$DRD4^{R2R7}$	-4.0803***	-6.0946***	-6.0408***	-6.4830***	-3.7640***	-3.7785**	-7.0899***
(Ethnicity-Weighted)	(1.1956)	(1.8999)	(1.6885)	(1.6363)	(1.3167)	(1.4983)	(2.2092)
Number of Countries	63	63	63	63	57	57	57
$\mathbb{R}^2$	0.80	0.85	0.90	0.91	0.84	0.85	0.92
	Panel C – Deper	ndent Variable: Aggre	egate PISA Student	Achievement Science	e Score		
$DRD4^{R2R7}$	-4.0774***	-5.8312***	-6.1372***	-6.0382***	-3.9233***	-3.6014**	-6.2710**
$({\bf Ethnicity\text{-}Weighted})$	(1.1537)	(1.8128)	(1.8795)	(1.5921)	(1.2086)	(1.3198)	(2.3373)
Number of Countries	63	63	63	63	57	57	57
$R^2$	0.82	0.88	0.90	0.92	0.83	0.84	0.91
Region Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
OPEC Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Biogeographic Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Major Religion Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
legal Origin Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Health-Related Effects	No	Yes	Yes	Yes	No	No	Yes
Economic Environment Effects	No	No	Yes	No	No	No	No
Social Value Effects	No	No	No	Yes	No	No	No
Educational Effects	No	No	No	No	No	Yes	Yes

Notes: The dependent variable is the country-level student achievement test score outcome in the academic subjects mathematics and science. This variable has been constructed from individual student achievement test score outcomes from the PISA and TIMSS studies during the various student assessment waves in the years 1995 to 2015. DRD4R2R7 refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure (ethnicity-weighted). Predicted Genetic Diversity refers to a country-level measure of overall genetic diversity (ancestry-adjusted). HLA Diversity refers to a country-level measure of genetic diversity associated with a set of genes located in the HLA region (ethnicityweighted). Ethnic Diversity refers to a country-level measure of ethno-linguistic diversity. Religious Diversity refers to a country-level measure of religious diversity. Region Fixed Effects refer to region controls for North America, Latin America and the Caribbean, Africa, East Asia, Central Asia, South Asia, Middle East, Europe, and Oceania. OPEC Fixed Effects refer to an indicator variable for member countries belonging to the Organization of the Petroleum Exporting Countries (OPEC). TIMSS Fixed Effects take a value of 1 if the student achievement test score outcome was derived from the TIMSS study and zero otherwise. Biogeographic Effects include In Absolute Latitude, Population in Area with Malaria, Temperature, and In Percentage Arable Land Area. Major Religion Effects refer to a set of country-level controls indicating the share of Muslims, Catholics, and Protestants in the population (measured in percent). Legal Origin Effects refer to a set of indicator variables that takes a value of 1 if the country's legal tradition is British, French, German, or Scandinavian. Health-Related Effects include Low Birth Weight, Infant Mortality Rate, Life Expectancy, Disability-Adjusted Life-Years, and Homicide Rate. Economic Environment Effects include In Real GDP per Capita, Government Consumption, Total Fertility Rate, Population Growth, and Trade Openness. Social Value Effects include Women in Parliament, Female Economic Participation, Federal Political System, Property Rights Protection, Freedom from Corruption, and In Number of Civil Conflicts. Educational Effects include No Education, Primary Education, Secondary Education, In Human Capital Index, Student-Teacher Ratio, Educational Expenditures, and Private Enrollment Rate. See the main text for additional details on data construction and sources. Constant term included but not shown. Robust standard errors are reported in parentheses.

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

Table 10: ADHD Prevalence Rate and Student Achievement Test Score (Instrumental Variables Estimation Approach)

					*		* * /	
Specification	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Estimator	OLS	OLS	OLS	OLS	2SLS	2SLS	2SLS	2SLS
TIMSS Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Continent Fixed Effects	No	Yes	No	Yes	No	Yes	No	Yes
Country Controls	No	No	Yes	Yes	No	No	Yes	Yes
	Panel A = D	lenendent Variable	: Aggregate Studen	t Achievement Tes	t Score			
ADHD Prevalence Rate	-6.4754***	-4.9478***	-3.7442**	-4.0975***	-6.3282***	-9.3471***	-5.2160***	-5.8155***
TID II D I TOVALORICO TRAVO	(1.3327)	(1.5553)	(1.3645)	(1.2962)	(2.3765)	(2.9513)	(1.5008)	(1.3908)
	(1.0021)	(1.0000)	(1.0010)	(112002)	(2.0.00)	(2.0010)	(1.0000)	(1.0000)
Number of Countries	37	37	37	37	37	37	37	37
Partial $\mathbb{R}^2$	0.31	0.26	0.70	0.72	0.31	0.05	0.69	0.70
			nostics: Testing fo	r Underidentification	on and Weak Identifi			
	Instrumented	Variable			ADHD Preva-	ADHD Preva-	ADHD Preva-	ADHD Preva-
					lence Rate	lence Rate	lence Rate	lence Rate
	Excluded Inst	ruments			$DRD4^{R2R7}$	$DRD4^{R2R7}$	$DRD4^{R2R7}$	$DRD4^{R2R7}$
					State History	State History	State History	State History
					Neolithic Tran-	Neolithic Tran-	Neolithic Tran-	Neolithic Tran-
					sition	sition	sition	sition
	Kleibergen-Pa	ap rk LM Statisti	с		4.0819	3.0741	10.6780	11.7508
	p-value				0.2528	0.3803	0.0136	0.0083
	Hansen J Stat	tistic			1.2362	1.5281	3.2706	1.2119
	p-value				0.5390	0.4658	0.1949	0.5456
	G D 11	I W II P Co or or			4.5000	F 0750	0.010	0.0000
		gg-Donald Wald F Statistic		4.7868 0.7399	5.3750 0.6589	9.0107	8.0292 0.3061	
		$H_0$ : Maximum relative OLS bias > 10% ( $p-value$ ) $H_0$ : Maximum relative OLS bias > 20% ( $p-value$ )				0.2104		
		: Maximum relative OLS bias $> 20\%$ $(p-value)$ : Maximum relative OLS bias $> 30\%$ $(p-value)$			0.2789 0.1250	0.2057 0.0826	0.0204 0.0044	0.0405 0.0103
	n <sub>0</sub> : Maximi	im relative OLS bia	s > 30% (p - value)	)	0.1250	0.0620	0.0044	0.0103
	Kleibergen-Pa	ap rk Wald F Sta	tistic		2.9290	6.2063	6.0950	6.5389

Notes: This table reports coefficient estimates from an instrument variables (IV) estimation approach on the association between student achievement test scores and the country-level ADHD prevalence rate. The results presented in columns (1) to (4) refers to the baseline OLS estimates. Columns (5) to (8) reports the corresponding 2SLS estimates. All specifications assess the sensitivity of the main findings to the inclusion of TIMSS fixed effects, columns (1) and (5), continent fixed effects, columns (2) and (6), a set of country controls, columns (3) and (7), and to the inclusion of continent fixed effects and country controls, columns (4) and (8). TIMSS Fixed Effects take a value of 1 if the student achievement test scores were derived from the TIMSS study and zero otherwise. Continent Fixed Effects refer to continent dummies for the Americas, Africa, Asia, and Europe. Country Controls refer to a set of Biogeographical Effects (i.e., In absolute latitude, share of the population living in area with malaria, and percentage of arable land area), Legal Origin Effects (i.e., dummy variables for British, French, German or Scandinavian legal origin), and Major Religion Effects (i.e., percentage of Muslims, Catholics, and Protestants in the population). See the main text for additional details on data construction and sources.

0.9334

0.6110

0.3916

0.5403

0.1288

0.0445

0.5562

0.1374

0.4936

0.1056

Panel A reports the regression coefficient associated with the country-level ADHD prevalence rate across the various model specifications. For the 2SLS estimates, the ADHD prevalence rate is treated as endogenous and instrumented using a set of excluded instrumental variables (i.e., the country-level  $DRD4^{R2R7}$  measure, ln State History Experience, and ln Agricultural Transition Timing). Panel B reports the first stage IV diagnostics regarding underidentification and weak identification of IV in the 2SLS estimates. The critical values for assessing the weakness of IV are tabulated in Stock and Yogo (2005) for various combinations regarding the number of endogenous variables, instruments, and the relative size of the OLS bias. Even though these values only apply to the non-robust Cragg-Donald Wald F statistic assuming the presence of i.i.d. errors, we follow the suggestion in Baum et al. (2007, p. 490) and apply these critical values with caution to the Kleibergen-Paap Wald F statistic which is robust to the presence of non-i.i.d. errors.

The Partial  $R^2$  refers to the model's  $R^2$  after partialling out the TIMSS and continent fixed effects, respectively, from the regression model. Robust standard errors are reported in parentheses.

 $H_0$ : Maximum relative OLS bias > 10% (p-value)

 $H_0$ : Maximum relative OLS bias > 20% (p-value)

 $H_0$ : Maximum relative OLS bias > 30%~(p-value)

<sup>\*:</sup> Significant at the 10% level. \*\*: Significant at the 5% level. \*\*\*: Significant at the 1% level.

# B Descriptive Statistics

 ${\bf Table~11:~Summary~Statistics~for~the~ADHD~Meta-Regression~Sample}$ 

Variable	N	Mean	SD	Minimum	Maximum
ADHD Prevalence Rate	179	0.0730	0.0527	0.0020	0.3450
Risk of Bias Score: Very High $(0 \le RoB \le 1)$	179	0.0168	0.1287	0	1
Risk of Bias Score: High $(2 \le RoB \le 3)$	179	0.2291	0.4214	0	1
Risk of Bias Score: Moderate $(4 \le RoB \le 5)$	179	0.5754	0.4957	0	1
Risk of Bias Score: Low $(6 \le RoB \le 8)$	179	0.1788	0.3842	0	1
DSM-III Diagnostic Criteria	179	0.0559	0.2303	0	1
DSM-III-R Diagnostic Criteria	179	0.1285	0.3356	0	1
DSM-IV Diagnostic Criteria	179	0.8156	0.3889	0	1
Population Sample: Community	179	0.1676	0.3746	0	1
Population Sample: School	179	0.7374	0.4413	0	1
Population Sample: Whole Population	179	0.0950	0.2940	0	1
Full Diagnostic Criteria Met: Yes	179	0.4190	0.4948	0	1
Type of ADHD Measurement: Interview	179	0.4358	0.4972	0	1
Type of ADHD Measurement: Diagnostic Report	179	0.1285	0.3356	0	1
Type of ADHD Measurement: Symptom-only Checklist	179	0.3966	0.4906	0	1
Type of ADHD Measurement: Unclear	179	0.0391	0.1944	0	1
Informant of ADHD: Child	179	0.0447	0.2072	0	1
informant of ADHD: Parent	179	0.2570	0.4382	0	1
informant of ADHD: Teacher	179	0.2123	0.4101	0	1
informant of ADHD: And Rule	179	0.1006	0.3016	0	1
informant of ADHD: Or Rule	179	0.0670	0.2508	0	1
Informant of ADHD: Unclear	179	0.0168	0.1287	0	1
Informant of ADHD: Clinician	179	0.3017	0.4603	0	1
Five-Year Indicator: 1980 to 1984	179	0.0112	0.1054	0	1
Five-Year Indicator: 1985 to 1989	179	0.0279	0.1652	0	1
Five-Year Indicator: 1990 to 1994	179	0.0615	0.2408	0	1
Five-Year Indicator: 1995 to 1999	179	0.1341	0.3417	0	1
Five-Year Indicator: 2000 to 2004	179	0.1508	0.3589	0	1
Five-Year Indicator: 2005 to 2009	179	0.3631	0.4823	0	1
Five-Year Indicator: 2010 to 2014	179	0.2514	0.4350	0	1
Region Dummy: Latin America and Caribbean	179	0.1285	0.3356	0	1
Region Dummy: North America	179	0.1955	0.3977	0	1
Region Dummy: Africa	179	0.0447	0.2072	0	1
Region Dummy: East Asia	179	0.1229	0.3292	0	1
Region Dummy: Central Asia	179	0.0168	0.1287	0	1
Region Dummy: South Asia	179	0.0335	0.1805	0	1
Region Dummy: Middle East	179	0.1453	0.3533	0	1
Region Dummy: Europe	179	0.2793	0.4499	0	1
Region Dummy: Oceania	179	0.0335	0.1805	0	1

Table 12: Pairwise Correlations for the ADHD Meta-Regression Sample

Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)
(1) ADHD Prevalence Rate	1.0000																	
(2) Risk of Bias Score: Very High $(0 \le RoB \le 1)$	-0.0659	1.0000																
(3) Risk of Bias Score: High $(2 \le RoB \le 3)$	0.0914	-0.0712	1.0000															
(4) Risk of Bias Score: Moderate (4 $\leq RoB \leq 5$ )	-0.0771	-0.1520	-0.6345	1.0000														
(5) Risk of Bias Score: Low $(6 \le RoB \le 8)$	0.0212	-0.0609	-0.2543	-0.5432	1.0000													
(6) DSM-III Diagnostic Criteria	-0.0790	-0.0318	-0.1326	0.1597	-0.0500	1.0000												
(7) DSM-III-R Diagnostic Criteria	-0.1853	0.0799	-0.1298	-0.0079	0.1258	-0.0934	1.0000											
(8) DSM-IV Diagnostic Criteria	0.2067	-0.0502	0.1906	-0.0878	-0.0790	-0.5117	-0.8076	1.0000										
(9) Population Sample: Community	-0.1535	-0.0586	-0.0310	0.0526	-0.0142	0.0211	0.1406	-0.1338	1.0000									
(10) Population Sample: School	0.1825	-0.0210	0.0835	-0.0245	-0.0529	-0.0207	-0.1123	0.1092	-0.7520	1.0000								
(11) Population Sample: Whole Population	-0.0784	0.1061	-0.0859	-0.0302	0.0975	0.0042	-0.0105	0.0066	-0.1454	-0.5429	1.0000							
(12) Full Diagnostic Criteria Met: Yes	-0.0601	-0.1109	0.0491	0.0422	-0.0712	0.1878	0.1476	-0.2387	0.1949	-0.0851	-0.1206	1.0000						
(13) Type of ADHD Measurement: Interview	-0.1070	-0.1147	0.0572	0.0255	-0.0572	0.1787	0.1339	-0.2214	0.2391	-0.1669	-0.0541	0.9207	1.0000					
(14) Type of ADHD Measurement: Diagnostic Report	-0.0990	0.0799	-0.0107	-0.0079	-0.0049	-0.0934	-0.0975	0.1395	0.0065	-0.3020	0.4451	-0.3261	-0.3374	1.0000				
(15) Type of ADHD Measurement: Symptom-only Checklist	0.1931	0.0721	-0.0615	-0.0197	0.0688	-0.1475	-0.0724	0.1499	-0.2109	0.3281	-0.2237	-0.6423	-0.7125	-0.3113	1.0000			
(16) Type of ADHD Measurement: Unclear	-0.0428	-0.0263	0.0272	-0.0016	-0.0189	0.0764	0.0087	-0.0527	-0.0905	0.1204	-0.0654	-0.1713	-0.1773	-0.0775	-0.1636	1.0000		
(17) Informant of ADHD: Child	-0.1062	-0.0282	-0.1179	0.0764	0.0402	-0.0526	0.0785	-0.0366	0.0477	0.0062	-0.0701	0.0355	0.0826	-0.0831	-0.0096	-0.0436	1.0000	
(18) Informant of ADHD: Parent	0.0069	0.0228	0.0445	-0.0121	-0.0408	-0.0874	-0.1494	0.1807	0.1811	-0.4045	0.3764	-0.2921	-0.2332	0.5001	-0.0587	-0.1186	-0.1272	1.0000
(19) Informant of ADHD: Teacher	0.1235	0.0386	-0.0554	-0.0239	0.0787	-0.0073	0.0864	-0.0703	-0.2329	0.3098	-0.1682	-0.4132	-0.4287	-0.0360	0.4169	0.1067	-0.1123	-0.3053
(20) Informant of ADHD: And Rule	-0.0601	0.1011	-0.0496	0.0241	-0.0106	-0.0813	-0.0729	0.1111	-0.1003	0.1151	-0.0450	-0.1710	-0.2564	-0.1284	0.3364	0.0284	-0.0723	-0.1966
(21) Informant of ADHD: Or Rule	-0.0358	-0.0350	0.0134	-0.0409	0.0498	0.1293	-0.1029	0.0122	0.0591	-0.0939	0.0656	-0.0013	-0.0103	-0.1029	0.0110	0.1764	-0.0580	-0.1576
(22) Informant of ADHD: Unclear	0.0224	-0.0170	0.1360	-0.0639	-0.0609	-0.0318	0.0799	-0.0502	-0.0586	0.0779	-0.0423	-0.0227	-0.0270	-0.0501	-0.0169	0.1982	-0.0282	-0.0768
(23) Informant of ADHD: Clinician	-0.0161	-0.0858	0.0472	0.0228	-0.0525	0.1051	0.1114	-0.1583	0.0635	0.0603	-0.1714	0.7493	0.7479	-0.2524	-0.5329	-0.1326	-0.1422	-0.3865
(24) Five-Year Indicator: 1980 to 1984	-0.0879	-0.0139	-0.0579	0.0913	-0.0496	0.4370	-0.0408	-0.2236	-0.0477	0.0634	-0.0344	0.0175	0.0138	-0.0408	-0.0862	0.2527	-0.0230	-0.0625
(25) Five-Year Indicator: 1985 to 1989	-0.0695	-0.0221	-0.0924	0.0770	0.0094	0.6969	-0.0651	-0.3566	0.0147	-0.0529	0.0607	0.1996	0.1929	-0.0651	-0.1374	-0.0342	-0.0367	-0.0997
(26) Five-Year Indicator: 1990 to 1994	-0.1303	0.1478	-0.0288	-0.0626	0.0627	-0.0622	0.6664	-0.5382	0.0097	0.0470	-0.0829	0.0184	0.0097	-0.0287	0.0303	-0.0516	0.0572	-0.0973
(27) Five-Year Indicator: 1995 to 1999	-0.0368	-0.0514	-0.0974	0.0395	0.0732	0.0471	0.3389	-0.3203	0.0429	-0.0260	-0.0156	0.0314	0.0179	-0.1511	0.0831	0.0052	-0.0851	-0.1189
(28) Five-Year Indicator: 2000 to 2004	0.0347	-0.0550	0.0303	0.0778	-0.1152	-0.0346	-0.1152	0.1199	0.0616	-0.0323	-0.0300	0.0217	0.1018	0.0248	-0.1184	-0.0045	-0.0156	0.0379
(29) Five-Year Indicator: 2005 to 2009	0.0408	-0.0081	0.0584	-0.0330	-0.0188	-0.1837	-0.2552	0.3290	-0.0900	0.1074	-0.0465	-0.0055	-0.0544	-0.0122	0.0289	0.0874	0.0616	0.0610
(30) Five-Year Indicator: 2010 to 2014	0.0749	0.0247	0.0519	-0.0754	0.0321	-0.1410	-0.2225	0.2755	0.0158	-0.0932	0.1198	-0.1267	-0.1197	0.1623	0.0566	-0.1169	-0.0007	0.1013
(31) Region Dummy: Latin America and Caribbean	0.1574	-0.0501	0.1483	-0.0417	-0.0920	-0.0934	-0.1474	0.1826	0.0065	0.0015	-0.0105	0.1476	0.1339	-0.1474	-0.0042	-0.0775	-0.0831	-0.0730
(32) Region Dummy: North America	0.0063	0.0454	-0.1346	-0.0325	0.1744	0.0641	0.1474	-0.1652	0.1559	-0.3140	0.2727	-0.1332	-0.1208	0.4421	-0.1406	-0.0995	-0.0385	0.2581
(33) Region Dummy: Africa	-0.0642	-0.0282	-0.1179	0.1311	-0.0304	-0.0526	-0.0831	0.1028	-0.0247	0.0676	-0.0701	0.0355	-0.0810	-0.0831	0.1562	-0.0436	-0.0468	-0.0653
(34) Region Dummy: East Asia	-0.1419	0.0837	0.0794	-0.0571	-0.0414	0.0571	0.0597	-0.0853	-0.0769	0.1460	-0.1213	0.2339	0.2201	-0.0929	-0.1644	0.0123	-0.0810	-0.0255
(35) Region Dummy: Central Asia	0.0075	-0.0170	0.0324	0.0241	-0.0609	-0.0318	-0.0501	0.0621	-0.0586	0.0779	-0.0423	-0.0227	-0.0270	-0.0501	-0.0169	0.1982	-0.0282	-0.0768
(36) Region Dummy: South Asia	-0.0272	-0.0243	-0.0276	0.0344	-0.0059	0.2250	-0.0715	-0.0715	-0.0005	0.0406	-0.0603	0.0306	0.0241	-0.0715	-0.0241	0.1225	0.1099	-0.0385
(37) Region Dummy: Middle East	0.3462	-0.0538	0.0394	-0.1271	0.1387	-0.1003	-0.1109	0.1551	-0.1425	0.2100	-0.1335	-0.0609	-0.0745	-0.1583	0.1843	-0.0014	0.0643	-0.1336
(38) Region Dummy: Europe	-0.2071	0.0157	-0.0134	0.0813	-0.0955	-0.0430	0.0958	-0.0572	-0.0127	0.0036	0.0107	-0.1249	-0.0700	-0.0530	0.0806	0.0671	0.1064	-0.0242
(39) Region Dummy: Oceania	-0.1137	-0.0243	-0.0276	0.0972	-0.0869	0.0898	0.0212	-0.0715	0.1657	-0.1710	0.0455	-0.0323	0.0241	0.1140	-0.0876	-0.0376	-0.0403	0.1036

Table 12: continued

							Table	12: cont	inued												
Variable	(19)	(20)	(21)	(22)	(23)	(24)	(25)	(26)	(27)	(28)	(29)	(30)	(31)	(32)	(33)	(34)	(35)	(36)	(37)	(38)	(39)
(1) ADHD Prevalence Rate																					
(2) Risk of Bias Score: Very High $(0 \leq RoB \leq 1)$																					
(3) Risk of Bias Score: High $(2 \le RoB \le 3)$																					
(4) Risk of Bias Score: Moderate (4 $\leq RoB \leq 5)$																					
(5) Risk of Bias Score: Low $(6 \le RoB \le 8)$																					
(6) DSM-III Diagnostic Criteria																					
(7) DSM-III-R Diagnostic Criteria																					
(8) DSM-IV Diagnostic Criteria																					
(9) Population Sample: Community																					
(10) Population Sample: School																					
(11) Population Sample: Whole Population																					
(12) Full Diagnostic Criteria Met: Yes																					
(13) Type of ADHD Measurement: Interview																					
(14) Type of ADHD Measurement: Diagnostic Report																					
(15) Type of ADHD Measurement: Symptom-only Checklist																					
(16) Type of ADHD Measurement: Unclear																					
(17) Informant of ADHD: Child																					
(18) Informant of ADHD: Parent																					
(19) Informant of ADHD: Teacher	1.0000																				
(20) Informant of ADHD: And Rule	-0.1736	1.0000																			
(21) Informant of ADHD: Or Rule	-0.1392	-0.0896	1.0000																		
(22) Informant of ADHD: Unclear	-0.0678	-0.0437	-0.0350	1.0000																	
(23) Informant of ADHD: Clinician	-0.3412	-0.2198	-0.1762	-0.0858	1.0000																
(24) Five-Year Indicator: 1980 to 1984	0.0748	-0.0355	-0.0285	-0.0139	0.0459	1.0000															
(25) Five-Year Indicator: 1985 to 1989	-0.0880	-0.0567	0.2257	-0.0221	0.1102	-0.0180	1.0000														
(26) Five-Year Indicator: 1990 to 1994	0.1516	-0.0856	-0.0686	0.1478	-0.0161	-0.0272	-0.0434	1.0000													
(27) Five-Year Indicator: 1995 to 1999	0.1165	-0.0225	-0.0399	-0.0514	0.0986	-0.0418	-0.0667	-0.1007	1.0000												
(28) Five-Year Indicator: 2000 to 2004	-0.0661	0.0148	0.1367	-0.0550	-0.0389	-0.0448	-0.0714	-0.1078	-0.1658	1.0000											
(29) Five-Year Indicator: 2005 to 2009	0.0057	-0.0207	-0.0166	-0.0081	-0.0660	-0.0803	-0.1280	-0.1932	-0.2971	-0.3182	1.0000										
(30) Five-Year Indicator: 2010 to 2014	-0.1119	0.1060	-0.1038	0.0247	-0.0161	-0.0616	-0.0982	-0.1483	-0.2280	-0.2442	-0.4376	1.0000									
(31) Region Dummy: Latin America and Caribbean	-0.1177	0.1492	0.0306	-0.0501	0.1114	-0.0408	-0.0651	-0.0983	-0.0531	0.2113	0.0225	-0.0686	1.0000								
(32) Region Dummy: North America	-0.0148	-0.1180	-0.0195	-0.0644	-0.1092	-0.0524	0.0874	0.1085	-0.0286	0.1858	-0.1965	0.0065	-0.1893	1.0000							
(33) Region Dummy: Africa	0.0861	0.1974	-0.0580	-0.0282	-0.0833	-0.0230	-0.0367	-0.0553	-0.0058	-0.0156	0.0616	-0.0007	-0.0831	-0.1066	1.0000						
(34) Region Dummy: East Asia	-0.1527	-0.0686	0.0357	0.0837	0.1988	-0.0398	0.0398	0.1168	-0.0474	-0.0627	0.0712	-0.0600	-0.1437	-0.1846	-0.0810	1.0000					
(35) Region Dummy: Central Asia	-0.0678	0.1011	0.1390	-0.0170	0.0090	-0.0139	-0.0221	-0.0334	-0.0514	0.0666	-0.0986	0.1250	-0.0501	-0.0644	-0.0282	-0.0489	1.0000				
(36) Region Dummy: South Asia	-0.0208	0.0409	-0.0499	-0.0243	0.0128	0.2755	0.1568	-0.0477	-0.0733	-0.0785	0.0530	-0.0364	-0.0715	-0.0918	-0.0403	-0.0697	-0.0243	1.0000			
(37) Region Dummy: Middle East	0.0962	0.0203	0.0797	0.0697	-0.0637	-0.0438	-0.0699	-0.0395	-0.0691	-0.1294	0.0514	0.1631	-0.1583	-0.2032	-0.0892	-0.1543	-0.0538	-0.0768	1.0000		
(38) Region Dummy: Europe	0.1640	-0.1254	-0.1171	0.0157	-0.0294	0.0523	-0.1055	-0.0556	0.1935	-0.1580	0.0736	-0.0451	-0.2391	-0.3069	-0.1347	-0.2331	-0.0813	-0.1159	-0.2566	1.0000	
(39) Region Dummy: Oceania	-0.0967	0.0409	0.0742	-0.0243	-0.0548	-0.0198	0.1568	0.0816	0.0178	0.0082	-0.0761	-0.0364	-0.0715	-0.0918	-0.0403	-0.0697	-0.0243	-0.0347	-0.0768	-0.1159	1.0000

Table 13: Summary Statistics for the ADHD and  $DRD4^{R2R7}$  Country Sample

Variable	N	Mean	SD	Minimum	Maximum
ADHD Prevalence Rate	42	0.0625	0.0368	0.0066	0.1647
$DRD4^{R2R7}$	42	0.2291	0.0508	0.1134	0.3473
Legal Origin: British	42	0.3095	0.4679	0	1
Legal Origin: French	42	0.4524	0.5038	0	1
Legal Origin: German	42	0.0952	0.2971	0	1
Legal Origin: Scandinavian	42	0.0952	0.2971	0	1
Legal Origin: Socialist	42	0.0476	0.2155	0	1
Major Religion Share: Protestantism	42	17.5595	27.9005	0	97.8000
Major Religion Share: Catholicism	42	28.7952	36.9618	0	96.9000
Major Religion Share: Islam	42	21.1886	37.3047	0	99.4000
ln Real GDP per Capita	42	9.6824	0.9373	6.7218	11.1580
Life Expectancy	42	72.8356	7.3760	47.5213	80.3303
ln Human Capital Index	42	0.9174	0.2778	0.1495	1.2631
ln State History 1 AD - 1500 AD	42	-1.2456	0.4751	-2.6539	-0.6198
ln Agricultural Transition Timing	42	8.7044	0.3158	7.9137	9.1874
Region Dummy: Latin America and Caribbean	42	0.1429	0.3542	0	1
Region Dummy: North America	42	0.0476	0.2155	0	1
Region Dummy: Africa	42	0.1190	0.3278	0	1
Region Dummy: East Asia	42	0.1190	0.3278	0	1
Region Dummy: Central Asia	42	0.0238	0.1543	0	1
Region Dummy: South Asia	42	0.0238	0.1543	0	1
Region Dummy: Middle East	42	0.1429	0.3542	0	1
Region Dummy: Europe	42	0.3333	0.4771	0	1
Region Dummy: Oceania	42	0.0476	0.2155	0	1

Table 14: Pairwise Correlations for the ADHD and  $DRD4^{R2R7}$  Country Sample

Variable	(1)	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(11)	(18) (19)	9) (20)	(21)	(22)	(23)	(24)
(1) ADHD Prevalence Rate	1.0000																						
(2) $DRD4^{R2R7}$	0.1942	1.0000																					
(3) Legal Origin: British	-0.1706	0.2089	1.0000																				
(4) Legal Origin: French	0.3274	-0.0329	-0.6085	1.0000																			
(5) Legal Origin: German	-0.1410	-0.1606	-0.2172	-0.2949	1.0000																		
(6) Legal Origin: Scandinavian	-0.0626	-0.0062	-0.2172	-0.2949	-0.1053	1.0000																	
(7) Legal Origin: Socialist	-0.1143	-0.1467	-0.1497	-0.2032	-0.0725	-0.0725	1.0000																
(8) Major Religion Share: Protestantism	-0.1444	0.1870	-0.0235	-0.4669	0.0955	0.8364	-0.1424	1.0000															
(9) Major Religion Share: Catholicism	0.2134	0.6223	-0.1521	0.3926	-0.0508	-0.2505	-0.1720	-0.2068	1.0000														
(10) Major Religion Share: Islam	0.2162	-0.6021	-0.1444	0.3904	-0.1858	-0.1856	-0.0870	-0.3376	-0.4188	1.0000													
(11) ln Real GDP per Capita	-0.0845	0.1352	0.0430	-0.2535	0.2099	0.2629	-0.1524	0.3905	0.0717	-0.1944	1.0000												
(12) Life Expectancy	-0.0983	0.2070	-0.1055	-0.1209	0.2227	0.2222	-0.1016	0.2607	0.2221	-0.3688	0.8354 1	1.0000											
(13) ln Human Capital Index	-0.2380	0.3205	0.1150	-0.4820	0.3377	0.2903	0.0114	0.4942	0.1248	-0.5868	0.8190 0	0.8341 1.	1.0000										
(14) ln State History 1 AD - 1500 AD	-0.2530	-0.2427	0.0595	0.0515	0.2686	-0.2974	-0.2099	-0.2526	-0.1225	0.1812	-0.0436 0	0.1312 -0.	-0.0904 1.	1.0000									
(15) ln Agricultural Transition Timing	0.1011	-0.2537	-0.1217	0.2700	-0.0877	-0.2370	0.0808	-0.3541	-0.0802	0.3293	0.1691 0	0.3220 -0.	-0.0083 0.	0.4579 1.	1.0000								
(16) Region Dummy: Latin America and Caribbean	0.4326	0.4863	-0.2733	0.4492	-0.1325	-0.1325	-0.0913	-0.2250	0.6861	-0.2258	-0.2168 -0	-0.0381 -0.	-0.1355 -0.	-0.2956 -0.	-0.1552 1.0	1.0000							
(17) Region Dummy: North America	-0.0251	0.1465	0.3340	-0.2032	-0.0725	-0.0725	-0.0500	0.1544	0.0582	-0.1243	0.2048 0	0.1444 0.	0.2680 0.	0.0554 0.	0.0104 -0.0	-0.0913 1.0	1.0000						
(18) Region Dummy: Africa	-0.0812	-0.2067	0.0719	0.1090	-0.1193	-0.1193	-0.0822	-0.0773	-0.2425	0.3051	-0.5728 -0	-0.6918 -0.	-0.5605 -0.	-0.0204 -0.	-0.4958 -0.	-0.1501 -0.0	-0.0822 1.0000	000					
(19) Region Dummy: East Asia	-0.2302	-0.1959	0.0719	-0.3341	0.3817	-0.1193	0.2630	-0.1787	-0.2641	-0.1978	-0.0351 0	0.1278 0.	0.0644 0.	0.3272 0.	0.0460 -0.	-0.1501 -0.0	-0.0822 -0.1351	1.0000	0				
(20) Region Dummy: Central Asia	0.0543	-0.2890	-0.1046	0.1718	-0.0507	-0.0507	-0.0349	-0.0995	-0.1227	0.3305	-0.0519 -0	-0.1020 -0.	-0.1483 0.	0.1147 0.	0.2348 -0.0	-0.0638 -0.0	-0.0349 -0.0574	574 -0.0574	74 1.0000	0			
(21) Region Dummy: South Asia	-0.0747	-0.1092	0.2333	-0.1419	-0.0507	-0.0507	-0.0349	-0.0932	-0.1176	-0.0406	-0.3292 -0	-0.2471 -0.	-0.2310 0.	0.0832 0.	0.1704 -0.0	-0.0638 -0.0	-0.0349 -0.0574	574 -0.0574	74 -0.0244	4 1.0000	0		
(22) Region Dummy: Middle East	0.3107	-0.3879	0.0210	0.1758	-0.1325	-0.1325	-0.0913	-0.2568	-0.3102	0.6662	0.1303 -0	-0.0530 -0.	-0.2418 0.	0.0668 0.	0.4718 -0.	-0.1667 -0.1	-0.0913 -0.1501	1021 -0.1501	0.0638	8 -0.0638	1.0000	0	
(23) Region Dunnny: Europe	-0.2655	0.1708	-0.2549	-0.1353	0.1147	0.4588	0.0791	0.4780	0.1333	-0.3804	0.4500 0	0.4298 0.	0.5165 -0.	-0.1590 -0.	-0.0191 -0.3	-0.2887 -0.	-0.1581 -0.2599	99 -0.2599	99 -0.1104	м -0.1104	4 -0.2887	7 1.0000	0
(24) Region Dummy: Oceania	-0.1205	0.2109	0.3340	-0.2032	-0.0725	-0.0725	-0.0500	0.1066	-0.0284	-0.1279	0.1386 0	0.1615 0.	0.2347 0.	0.0644 -0.	-0.0946 -0.0	-0.0913 -0.0	-0.0500 -0.0822	322 -0.0822	22 -0.0349	9 -0.0349	9 -0.0913	3 -0.1581	1.0000

 ${\bf Table~15:~Summary~Statistics~for~the~81-Country~Sample}$ 

Variable	N	Mean	SD	Minimum	Maximum
SATS	81	4.4252	0.6240	2.9675	5.5807
Indicator: TIMSS	81	0.2222	0.4183	0	1
$DRD4^{R2R7}$	81	0.2323	0.0660	0.0980	0.4583
ln Absolute Latitude	81	3.3822	0.7434	0.3124	4.1589
Population in Area with Malaria	81	0.1115	0.2592	0	1
Temperature	81	14.4297	8.5190	-7.9294	27.2440
ln Percentage Arable Land Area	81	2.3522	1.1883	-2.3010	4.0291
Major Religion Share: Protestantism	81	11.6901	22.4612	0	97.8000
Major Religion Share: Catholicism	81	34.0333	38.3069	0	96.9000
Major Religion Share: Islam	81	21.1746	35.2328	0	99.4000
Legal Origin: British	81	0.1852	0.3909	0	1
Legal Origin: French	81	0.4321	0.4985	0	1
Legal Origin: German	81	0.0617	0.2422	0	1
Legal Origin: Scandinavian	81	0.0494	0.2180	0	1
Legal Origin: Socialist	81	0.2716	0.4476	0	1
Low Birth Weight	81	8.0984	2.8299	4.1000	18.6333
Infant Mortality Rate	81	17.1026	13.3284	3.1615	60.9462
Life Expectancy	81	72.3863	5.0010	57.4585	80.3303
Disability-Adjusted Life-Years	81	0.3272	0.1156	0.1793	0.9579
Homicide Rate	81	6.5526	11.3658	0.4333	65.3867
n GDP per Capita	81	9.4649	0.8080	7.7414	11.1580
Government Consumption	81	0.1973	0.0597	0.0743	0.3407
Total Fertility Rate	81	2.4207	1.0354	1.3262	5.2726
Population Growth	81	1.0896	1.2494	-1.2606	6.7407
Trade Openness	81	83.5695	47.3345	21.1660	356.3131
Women in Parliament	81	16.9584	9.6070	0	44.1650
Female Economic Participation	81	39.1951	9.5224	13.7139	48.8440
Government Structure: Federalist	81	0.2963	0.4595	0	1
Property Rights Protection	81	57.4621	22.6865	10	91.7500
Freedom from Corruption	81	50.4475	22.4963	18.4200	94.7350
n Number of Civil Conflicts	81	-2.2126	3.2428	-4.6052	4.2197
Predicted Genetic Diversity	81	0.7242	0.0234	0.6434	0.7563
HLA Genetic Diversity	81	0.3248	0.0168	0.2649	0.3529
Ethnic Diversity	81	0.3547	0.2175	0.0020	0.7517
Religious Diversity	81	0.4127	0.2381	0.0035	0.8603
Region Dummy: Latin America and Caribbean	81	0.1605	0.3694	0	1
Region Dummy: North America	81	0.0247	0.1561	0	1
Region Dummy: Africa	81	0.0988	0.3002	0	1
Region Dummy: East Asia	81	0.1111	0.3162	0	1
Region Dummy: Central Asia	81	0.0741	0.2635	0	1
Region Dummy: Middle East	81	0.0988	0.3002	0	1
Region Dummy: Europe	81	0.4074	0.4944	0	1
Region Dummy: Oceania	81	0.0247	0.1561	0	1
OPEC Dummy	81	0.0741	0.2635	0	1

 ${\bf Table~16:~Pairwise~Correlations~for~the~81-Country~Sample}$ 

Variable	(1)	(2)	(2)	(4)	(5)	Table I	(7)	(9)	(0)	(10)	ountry Sa		(12)	(14)	(15)	(16)	(17)	(10)	/10)	(20)	(91)	(22)
(1) SATS	1.0000	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)
(1) SALS (2) Indicator: TIMSS	-0.3720	1.0000																				
(2) Indicator: 118/55 (3) DRD4 <sup>R2R7</sup>		-0.1570	1.0000																			
(4) In Absolute Latitude	-0.0548 0.4022	-0.1370	-0.4613	1.0000																		
(5) Population in Area with Malaria	-0.3478	0.3418	0.4592	-0.4955	1.0000																	
(6) Temperature	-0.5591	0.2844	0.3243	-0.7692	0.5137	1.0000																
(7) In Percentage Arable Land Area	0.2473	-0.2536	0.1380	0.3379	-0.1636	-0.2808	1.0000															
(8) Major Religion Share: Protestantism	0.3578	-0.1393	0.0939	0.2720	-0.1198	-0.2808	0.0273	1.0000														
(9) Major Religion Share: Catholicism	-0.0046	-0.2165	0.4824	-0.1206	0.0446	0.1043	0.1773	-0.1945	1.0000													
(10) Major Religion Share: Islam	-0.4169	0.3067	-0.4261	-0.0726	0.0642	0.3064	-0.3463	-0.2910	-0.4868	1.0000												
(11) Legal Origin: British	0.0374	0.1274	0.1521	-0.2747	0.0864	0.1764	-0.1811	0.1227	-0.1616	-0.0936	1.0000											
(12) Legal Origin: French	-0.4058	0.0733	0.1861	-0.2538	0.1744	0.4827	-0.1282	-0.3529	0.4051	0.2975	-0.4158	1.0000										
(13) Legal Origin: German	0.3247	-0.1371	-0.0391	0.1335	-0.1110	-0.1617	0.1006	0.1166	0.0147	-0.1538	-0.1223	-0.2237	1.0000									
(14) Legal Origin: Scandinavian	0.2424	-0.1218	-0.0146	0.2244	-0.0987	-0.3061	-0.0333	0.7855	-0.2002	-0.1372	-0.1087	-0.1988	-0.0585	1.0000								
(15) Legal Origin: Socialist	0.1256	-0.0593	-0.3118	0.3411	-0.1616	-0.4551	0.2627	-0.1599	-0.2204	-0.0996	-0.2911	-0.5326	-0.1566	-0.1392	1.0000							
(16) Low Birth Weight	-0.5509	0.3893	0.1446	-0.4504	0.3878	0.6046	-0.0878	-0.2642	0.0317	0.2486	0.2269	0.3255	-0.1339	-0.2744	-0.3545	1.0000						
(17) Infant Mortality Rate	-0.7660	0.4345	-0.0950	-0.3022	0.3915	0.3495	-0.1543	-0.2491	-0.2044	0.3644	0.0222	0.1212	-0.2482	-0.2294	0.0917	0.5351	1.0000					
(18) Life Expectancy	0.7030	-0.5296	0.0677	0.2949	-0.4016	-0.3168	0.1647	0.2225	0.2177	-0.2900	-0.0619	-0.0297	0.2763	0.2495	-0.1838	-0.5040	-0.8996	1.0000				
(19) Disability-Adjusted Life-Years	-0.2938	0.2993	-0.0382	0.0451	0.1832	-0.0779	0.0669	0.1003	-0.1568	-0.1525	0.1611	-0.2747	-0.1384	-0.0663	0.2724	0.2341	0.5700	-0.7234	1.0000			
(20) Homicide Rate	-0.4679	0.2705	0.1991	-0.3470	0.2848	0.3210	-0.0832	-0.0817	0.3287	-0.1644	0.0283	0.2063	-0.1327	-0.1090	-0.1296	0.3156	0.3551	-0.4101	0.2196	1.0000		
(21) ln GDP per Capita	0.6312	-0.4026	-0.0747	0.3193	-0.4570	-0.3149	-0.0877	0.3836	0.0689	-0.1807	0.1907	-0.1662	0.2628	0.2747	-0.2575	-0.3249	-0.7494	0.7170	-0.3692	-0.3582	1.0000	
(22) Government Consumption	-0.0521	0.1322	-0.4609	0.3621	-0.2450	-0.3176	0.0618	-0.1081	-0.3955	0.2557	-0.2656	-0.2250	-0.2820	-0.0122	0.6411	-0.2190	0.1190	-0.2424	0.2616	-0.2036	-0.2838	1.0000
(23) Total Fertility Rate	-0.7303	0.5597	0.0244	-0.4087	0.5663	0.6173	-0.5188	-0.2187	-0.1120	0.5732	0.0963	0.4175	-0.2383	-0.1453	-0.3494	0.5400	0.6431	-0.5993	0.1454	0.3141	-0.5054	-0.0360
(24) Population Growth	-0.4875	0.2939	0.0370	-0.4738	0.3834	0.6380	-0.6279	-0.1824	-0.0702	0.5551	0.1996	0.4417	-0.1295	-0.1173	-0.5390	0.4523	0.3048	-0.2286	-0.1648	0.1581	-0.0484	-0.2544
(25) Trade Openness	0.2172	-0.0602	-0.0005	-0.2603	-0.0338	0.0555	-0.0983	-0.0554	-0.0491	-0.0112	0.1510	-0.1757	-0.0947	-0.0568	0.1427	-0.0189	-0.1852	0.1082	-0.0897	-0.1305	0.1565	0.1175
(26) Women in Parliament	0.3842	-0.3461	0.3197	0.2407	-0.1663	-0.3497	0.2736	0.6274	0.2606	-0.5125	-0.0151	-0.1782	0.1083	0.5362	-0.1081	-0.2947	-0.3432	0.3920	-0.0581	-0.0236	0.3122	-0.2462
(27) Female Economic Participation	0.4448	-0.3100	0.1991	0.2164	-0.1365	-0.5260	0.4267	0.3430	0.1415	-0.7822	0.1096	-0.5940	0.1024	0.1914	0.4172	-0.3186	-0.1656	0.1360	0.3296	-0.0336	0.1109	-0.0398
(28) Government Structure: Federalist	0.3089	-0.1517	0.0963	0.1129	-0.1949	-0.2203	0.1211	0.0189	0.1369	-0.2902	0.1779	-0.1294	0.1706	-0.0231	-0.0923	-0.1413	-0.3596	0.3223	-0.1170	-0.0861	0.2861	-0.1743
(29) Property Rights Protection	0.6028	-0.3092	0.1380	0.1809	-0.2919	-0.2611	-0.0333	0.5294	0.1497	-0.3746	0.3568	-0.1662	0.3164	0.3170	-0.4521	-0.1685	-0.6055	0.5866	-0.2233	-0.2334	0.7533	-0.3826
(30) Freedom from Corruption	0.6144	-0.2851	0.1117	0.1979	-0.3000	-0.2708	-0.1212	0.6041	0.0862	-0.2981	0.3183	-0.1840	0.2717	0.4251	-0.4271	-0.2736	-0.6275	0.6423	-0.2818	-0.2879	0.7928	-0.3630
(31) In Number of Civil Conflicts	-0.4313	0.2331	0.1520	-0.2870	0.2707	0.2706	0.1127	-0.2473	-0.0916	0.3013	0.0124	0.2157	-0.1904	-0.1692	-0.0656	0.3443	0.4455	-0.4604	0.1974	0.1921	-0.4757	0.0698
(32) Predicted Genetic Diversity	0.0718	0.1049	-0.5937	0.5394	-0.3914	-0.2868	0.1261	0.0825	-0.4022	0.3486	0.0006	-0.2056	-0.0947	0.0543	0.2532	-0.0716	-0.0283	-0.0038	0.1938	-0.3934	0.2015	0.4338
(33) HLA Genetic Diversity	0.4179	-0.1324	-0.3001	0.5325	-0.5023	-0.3956	0.1170	0.2960	0.0080	-0.0156	0.0489	-0.1027	0.1099	0.1848	-0.0778	-0.2830	-0.4229	0.4524	-0.1028	-0.3315	0.5491	0.0026
(34) Ethnic Diversity	-0.4308	0.1265	0.1128	-0.3364	0.2059	0.2344	-0.2261	-0.1668	-0.0905	0.2394	0.1438	0.0958	-0.2262	-0.2865	0.0297	0.2059	0.3633	-0.4193	0.2051	0.1758	-0.2216	0.1007
(35) Religious Diversity	0.1845	-0.0073	0.0290	-0.0223	-0.0703	-0.1156	0.0895	0.1712	-0.1420	-0.2915	0.3769	-0.4621	0.1772	-0.1747	0.1746	-0.0483	-0.1003	0.0181	0.2090	-0.0249	0.1223	0.0553
(36) Region Dummy: Latin America and Caribbean	-0.3920	-0.0719	0.3273	-0.4207	0.1287	0.3713	-0.1384	-0.1598	0.5874	-0.2533	-0.1219	0.4334	-0.1121	-0.0997	-0.2670	0.1650	0.1493	-0.0952	-0.1171	0.6204	-0.2510	-0.3501
(37) Region Dummy: North America	0.1669	-0.0850	0.0719	0.1042	-0.0689	-0.2844	-0.0091	0.1776	0.0178	-0.0930	0.3338	-0.1388	-0.0408	-0.0363	-0.0972	-0.0593	-0.1306	0.1651	-0.0792	-0.0424	0.2112	-0.1676
(38) Region Dummy: Africa	-0.4092	0.3207	-0.1061	-0.1059	0.0790	0.2823	-0.0745	-0.0015	-0.2191	0.2880	0.1618	0.1289	-0.0849	-0.0755	-0.2021	0.4005	0.5116	-0.5031	0.4647	0.0394	-0.2782	-0.0284
(39) Region Dummy: East Asia	0.1297	0.0000	0.2352	-0.4532	0.3733	0.2261	-0.0975	-0.1392	-0.2097	-0.0783	0.1348	-0.1498	0.2358	-0.0806	-0.0393	0.1287	0.0634	-0.0806	-0.0941	-0.0990	-0.1680	-0.1656
(40) Region Dummy: Central Asia	-0.1529	-0.0378	-0.4375	0.1329	-0.0904	-0.2159	0.0656	-0.1439	-0.2478	0.2606	-0.1348	-0.1516	-0.0725	-0.0645	0.3572	0.0187	0.3599	-0.2153	0.1051	-0.0179	-0.2223	0.2224
(41) Region Dummy: Middle East	-0.2317	0.3207	-0.2985	-0.0109	0.1116	0.2846	-0.4326	-0.1684	-0.2494	0.6307	-0.0513	0.2960	-0.0849	-0.0755	-0.2021	0.1076	0.0078	-0.0542	-0.2411	-0.0857	0.0512	0.2539
(42) Region Dummy: Europe	0.5655	-0.2619	0.0274	0.5630	-0.3589	-0.5718	0.4660	0.2894	0.1193	-0.3978	-0.2012	-0.3175	0.1005	0.2749	0.3410	-0.4718	-0.5743	0.4659	0.0153	-0.3254	0.4352	0.1937
(43) Region Dummy: Oceania	0.1972	-0.0850	0.1070	0.0264	-0.0689	0.0332	-0.0773	0.1355	-0.0413	-0.0958	0.3338	-0.1388	-0.0408	-0.0363	-0.0972	-0.1060	-0.1387	0.1829	-0.1090	-0.0749	0.1568	-0.0913
(44) OPEC Dummy	-0.2368	0.1890	-0.0496	-0.1304	0.0504	0.3169	-0.3295	-0.1357	-0.2446	0.5381	-0.0135	0.2291	-0.0725	-0.0645	-0.1727	0.0175	0.1235	-0.1295	-0.1585	-0.0734	0.1127	0.0251
(44) OPEC Dummy	-0.2368	0.1890	-0.0496	-0.1304	0.0504	0.3169	-0.3295	-0.1357	-0.2446	0.5381	-0.0135	0.2291	-0.0725	-0.0645	-0.1727	0.0175	0.1235	-0.1295	-0.1585	-0.0734	0.1127	0.0251

#### Table 16: continued

Mathematic   Mat																							
Continue	Variable	(23)	(24)	(25)	(26)	(27)	(28)	(29)	(30)	(31)	(32)	(33)	(34)	(35)	(36)	(37)	(38)	(39)	(40)	(41)	(42)	(43)	(44)
Continue	(1) SATS																						
Continue	(2) Indicator: TIMSS																						
Page	(3) $DRD4^{R2R7}$																						
Page	(4) In Absolute Latitude																						
1	(5) Population in Area with Malaria																						
Page	(6) Temperature																						
Page	(7) In Percentage Arable Land Area																						
Paris	(8) Major Religion Share: Protestantism																						
1	(9) Major Religion Share: Catholicism																						
Control   Cont	(10) Major Religion Share: Islam																						
Control   Cont	(11) Legal Origin: British																						
14 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(12) Legal Origin: French																						
Control   Cont	(13) Legal Origin: German																						
Contingent   Con	(14) Legal Origin: Scandinavian																						
1	(15) Legal Origin: Socialist																						
Properties	(16) Low Birth Weight																						
14 September 14 September 15 September 16 September 16 September 16 September 17 September 17 September 17 September 17 September 17 September 17 September 18 September 18 September 18 September 18 September 18 September 18 September 19 September 29 Se	(17) Infant Mortality Rate																						
9 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(18) Life Expectancy																						
Companies	(19) Disability-Adjusted Life-Years																						
14 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(20) Homicide Rate																						
14 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(21) ln GDP per Capita																						
14 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(22) Government Consumption																						
14 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(23) Total Fertility Rate	1.0000																					
Common Franction   Common Fran	(24) Population Growth	0.7693	1.0000																				
Programment Recommender Federication 1978   3.78	(25) Trade Openness	-0.1410	0.0642	1.0000																			
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(26) Women in Parliament	-0.4114	-0.3641	-0.0282	1.0000																		
Property Rights Protection   G-374   G-378	(27) Female Economic Participation	-0.6775	-0.7180	0.0591	0.4947	1.0000																	
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(28) Government Structure: Federalist	-0.3389	-0.2252	-0.2252	0.1306	0.2165	1.0000																
19 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(29) Property Rights Protection	-0.3742	-0.0723	0.1737	0.4184	0.2557	0.2860	1.0000															
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(30) Freedom from Corruption	-0.3425	-0.0076	0.2042	0.5075	0.2125	0.1975	0.9274	1.0000														
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	(31) ln Number of Civil Conflicts	0.2729	0.0810	-0.2369	-0.2166	-0.2459	-0.1352	-0.4839	-0.4968	1.0000													
(4) Ethnic Diversity 0.290 0.270 0.120 0.230 0.164 0.150 0.280 0.2	(32) Predicted Genetic Diversity	-0.0979	-0.1235	0.0209	-0.1118	-0.1289	-0.0384	-0.0324	0.0145	-0.1202	1.0000												
1	(33) HLA Genetic Diversity	-0.3211	-0.1966	0.0022	0.3343	0.0607	0.1247	0.4125	0.5048	-0.3773	0.6426	1.0000											
(36) Region Dummy: Latin America and Caribban (27) Region Dummy: North America (37) Region Dummy: North America (38) Region Dummy: Morth America (38) Region Dummy: East Asia (38) Region Dummy: Central Asia (38) Region Dummy: Morth America (38) Region Dummy: Morth	(34) Ethnic Diversity	0.2901	0.2720	0.1129	-0.2339	-0.1648	-0.1197	-0.3152	-0.2866	0.3091	-0.0500	-0.3006	1.0000										
(37) Region Dummy: North America -0.096	(35) Religious Diversity	-0.2798	-0.2469	0.0738	0.0825	0.3883	0.2100	0.1558	0.1262	-0.1168	0.0985	0.1063	0.1334	1.0000									
(38) Region Dummy: Africa 0.3855 0.215 0.255 0.215 0.2159 0.225 0.2179 0.225 0	(36) Region Dummy: Latin America and Caribbean	0.2108	0.1415	-0.2008	0.0573	-0.0604	0.1582	-0.1342	-0.1925	0.0669	-0.5688	-0.3793	0.1046	-0.1421	1.0000								
(39) Region Dummy: East Asia 0.036 0.129 0.240 0.1317 0.087 0.087 0.087 0.087 0.087 0.088	(37) Region Dummy: North America	-0.0960	-0.0072	-0.1392	0.0259	0.1104	0.2452	0.2229	0.2255	-0.1181	-0.0112	0.1714	0.1815	0.2335	-0.0696	1.0000							
(40) Region Dummy: Central Asia	(38) Region Dummy: Africa	0.3855	0.2154	-0.0636	-0.1179	-0.2398	-0.2148	-0.0995	-0.0922	0.2513	0.2199	0.0315	0.0972	-0.0323	-0.1447	-0.0527	1.0000						
(41) Region Dummy: Middle East 0.5203 0.609 0.0076 0.4509	(39) Region Dummy: East Asia	0.0365	0.1299	0.2420	-0.1317	0.0879	-0.0574	-0.0357	-0.0714	0.0946	-0.4885	-0.4183	0.0017	0.0058	-0.1546	-0.0563	-0.1170	1.0000					
(41) Region Dummy: Middle East 0.5203 0.609 0.0076 0.4509		-0.0424	-0.1594	-0.0197	-0.1736	0.1287	-0.1835	-0.2678	-0.3050	0.0887	0.1642	-0.1074	0.0673	0.0333	-0.1237	-0.0450	-0.0936	-0.1000	1.0000				
(42) Region Dummy: Europe -0.6543 -0.6075 0.1313 0.4197 0.4567 0.1713 0.2922 0.3185 0.2922 0.3185 0.2923 0.3185 0.2923 0.3185 0.2924 0.2924 0.2924 0.2924 0.2924 0.2924 0.2924 0.2924 0.		0.5203	0.6099	-0.0076	-0.4509	-0.7493	-0.2148	-0.1824	-0.1145	0.0287	0.2898	0.0281	0.2141	-0.1001	-0.1447	-0.0527	-0.1096	-0.1170	-0.0936	1.0000			
(43) Region Dummy: Oceania -0.0748 0.0153 -0.1215 0.1629 0.0928 0.0710 0.2358 0.2780 -0.1181 -0.0444 0.1065 -0.0808 0.2712 -0.0696 -0.0253 -0.0527 -0.0563 -0.0450 -0.0527 -0.1319 1.0000		-0.6543	-0.6075	0.1313	0.4197	0.4567	0.1773	0.2922	0.3185	-0.2532	0.3578	0.4841	-0.3359	0.0056	-0.3625	-0.1319	-0.2745	-0.2932	-0.2345	-0.2745	1.0000		
																					-0.1319	1.0000	
	(44) OPEC Dummy		0.4722	-0.0879	-0.3322	-0.5607	-0.1835	-0.1852	-0.0972	0.1364	0.0912	-0.0603	0.2619	-0.2434	-0.1237	-0.0450	0.0644	0.0500	-0.0800		-0.2345	-0.0450	1.0000

Table 17: Summary Statistics for the 73-Country Sample

Variable	N	Mean	SD	Minimum	Maximum
SATS	73	4.4443	0.6393	2.9675	5.5807
Indicator: TIMSS	73	0.2192	0.4166	0	1
$DRD4^{R2R7}$	73	0.2360	0.0665	0.0980	0.4583
ln Absolute Latitude	73	3.3604	0.7757	0.3124	4.1589
Population in Area with Malaria	73	0.1119	0.2559	0	1
Temperature	73	14.4781	8.7231	-7.9294	27.2440
ln Percentage Arable Land Area	73	2.3723	1.1032	-0.5777	4.0291
Major Religion Share: Protestantism	73	12.4863	23.4060	0	97.8000
Major Religion Share: Catholicism	73	36.5671	39.2930	0	96.9000
Major Religion Share: Islam	73	19.2184	34.5871	0	99.4000
Legal Origin: British	73	0.1918	0.3964	0	1
Legal Origin: French	73	0.4521	0.5011	0	1
Legal Origin: German	73	0.0685	0.2543	0	1
Legal Origin: Scandinavian	73	0.0548	0.2292	0	1
Legal Origin: Socialist	73	0.2329	0.4256	0	1
Low Birth Weight	73	8.1661	2.8880	4.1000	18.6333
Infant Mortality Rate	73	16.7928	13.2230	3.1615	60.9462
Life Expectancy	73	72.4116	5.1666	57.4585	80.3303
Disability-Adjusted Life-Years	73	0.3288	0.1205	0.1793	0.9579
Homicide Rate	73	6.9954	11.8844	0.4333	65.3867
ln GDP per Capita	73	9.4895	0.8251	7.7414	11.1580
Government Consumption	73	0.1922	0.0580	0.0743	0.3407
Total Fertility Rate	73	2.4360	1.0178	1.3262	5.1311
Population Growth	73	1.1204	1.2016	-1.2167	6.7407
Trade Openness	73	83.9198	49.5013	21.1660	356.3131
Women in Parliament	73	17.4489	9.6758	0	44.1650
Female Economic Participation	73	39.3555	9.3171	13.7139	48.8440
Government Structure: Federalist	73	0.3014	0.4620	0	1
Property Rights Protection	73	59.3927	21.9252	10	91.7500
Freedom from Corruption	73	51.8108	22.1564	18.4200	94.735
In Number of Civil Conflicts	73	-2.4176	3.2445	-4.6052	4.2197
No Education	73	14.536	15.9471	0.2229	67.1353
Primary Education	73	33.4813	13.9370	5.1631	66.7962
Secondary Education	73	38.6610	14.4345	12.4997	68.8693
In Human Capital Index	73	0.9419	0.2143	0.3958	1.2631
Student-Teacher Ratio	73	16.1612	5.6989	8.8166	33.6070
Educational Expenditures	73	4.4710	1.1693	1.5938	7.1392
Private Enrollment Rate		10.9734	13.8670	0.2273	71.9354
Predicted Genetic Diversity	73 73	0.7226	0.0240	0.6434	0.7563
HLA Genetic Diversity	73			0.2649	0.3529
v		0.3244	0.0175		
Ethnic Diversity  Policious Diversity	73 73	0.3552	0.2211	0.0020	0.7517
Religious Diversity	73	0.3936	0.2357	0.0035	0.8603
Region Dummy: Latin America and Caribbean	73	0.1781	0.3852	0	1
Region Dummy: North America	73	0.0274	0.1644	0	1
Region Dummy: Africa	73	0.1096	0.3145	0	1
Region Dummy: East Asia	73	0.1233	0.3310	0	1
Region Dummy: Central Asia	73	0.0548	0.2292	0	1
Region Dummy: Middle East	73	0.0822	0.2766	0	1
Region Dummy: Europe	73	0.4110	0.4954	0	1
Region Dummy: Oceania	73	0.0137	0.1170	0	1
OPEC Dummy	73	0.0822	0.2766	0	1

Table 18: Pairwise Correlations for the 73-Country Sample

							Tai	ne 16. 1	airwise C	Offeracion	s for the i	3-Countr	у запіріе												
Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)	(23)	(24)	(25)
(1) SATS	1.0000																								
(2) Indicator: TIMSS	-0.4118	1.0000																							
(3) $DRD4^{R2R7}$	-0.1115	-0.1823	1.0000																						
(4) In Absolute Latitude	0.4318	-0.0851	-0.4715	1.0000																					
(5) Population in Area with Malaria	-0.3687	0.3015	0.5288	-0.5015	1.0000																				
(6) Temperature	-0.6099	0.2761	0.3418	-0.7717	0.4949	1.0000																			
(7) In Percentage Arable Land Area	0.2790	-0.1928	0.1364	0.3391	-0.0076	-0.2171	1.0000																		
(8) Major Religion Share: Protestantism	0.3404	-0.1399	0.0618	0.2911	-0.1211	-0.4024	0.0225	1.0000																	
(9) Major Religion Share: Catholicism	-0.0362	-0.2250	0.4715	-0.1038	0.0610	0.0994	0.1875	-0.2313	1.0000																
(10) Major Religion Share: Islam	-0.4173	0.2857	-0.3896	-0.0798	-0.0072	0.3077	-0.3107	-0.2769	-0.4853	1.0000															
(11) Legal Origin: British	-0.0166	0.1625	0.1190	-0.2708	0.1105	0.1582	-0.2056	0.0946	-0.2003	-0.0555	1.0000														
(12) Legal Origin: French	-0.4273	0.0510	0.2025	-0.2361	0.1242	0.4732	-0.0570	-0.3790	0.3998	0.3114	-0.4425	1.0000													
(13) Legal Origin: German	0.3271	-0.1437	-0.0563	0.1430	-0.1193	-0.1686	0.1096	0.1091	-0.0024	-0.1503	-0.1321	-0.2463	1.0000												
(14) Legal Origin: Scandinavian	0.2428	-0.1276	-0.0288	0.2341	-0.1060	-0.3174	-0.0423	0.7886	-0.2219	-0.1340	-0.1173	-0.2187	-0.0653	1.0000											
(15) Legal Origin: Socialist	0.1924	-0.0569	-0.3001	0.3186	-0.1208	-0.4330	0.2159	-0.1317	-0.1633	-0.1530	-0.2684	-0.5004	-0.1494	-0.1327	1.0000										
(16) Low Birth Weight	-0.5790	0.4422	0.1837	-0.4540	0.3941	0.6185	-0.0710	-0.2783	0.0215	0.2373	0.2503	0.3112	-0.1452	-0.2899	-0.3566	1.0000									
(17) Infant Mortality Rate	-0.7990	0.5263	-0.0174	-0.3434	0.4606	0.4118	-0.2232	-0.2414	-0.1871	0.3426	0.0681	0.1637	-0.2583	-0.2387	0.0266	0.5399	1.0000								
(18) Life Expectancy	0.7068	-0.5665	0.0292	0.3116	-0.4208	-0.3470	0.1884	0.2104	0.2077	-0.2700	-0.1108	-0.0314	0.2816	0.2541	-0.1650	-0.5056	-0.9156	1.0000							
(19) Disability-Adjusted Life-Years	-0.2993	0.3456	-0.0368	0.0361	0.2289	-0.0528	0.0174	0.1038	-0.1661	-0.1451	0.1865	-0.2773	-0.1439	-0.0703	0.2766	0.2375	0.5868	-0.7319	1.0000						
(20) Homicide Rate	-0.4919	0.2994	0.1934	-0.3446	0.3131	0.3369	-0.1104	-0.0947	0.3157	-0.1511	0.0266	0.2034	-0.1445	-0.1192	-0.1138	0.3214	0.3853	-0.4225	0.2159	1.0000					
(21) ln GDP per Capita	0.6318	-0.4446	-0.1200	0.3633	-0.5460	-0.3860	-0.0333	0.3725	0.0456	-0.1842	0.1499	-0.2103	0.2641	0.2771	-0.1991	-0.3673	-0.7800	0.7285	-0.3650	-0.3781	1.0000				
(22) Government Consumption	0.0027	0.1365	-0.4572	0.3477	-0.2317	-0.2961	0.0053	-0.0668	-0.3827	0.2436	-0.2245	-0.2137	-0.2833	0.0079	0.6258	-0.1986	0.0989	-0.2446	0.2788	-0.1951	-0.2231	1.0000			
(23) Total Fertility Rate	-0.7898	0.5674	0.0492	-0.4028	0.5098	0.6048	-0.4408	-0.2324	-0.1263	0.5798	0.1194	0.3749	-0.2606	-0.1598	-0.3109	0.5455	0.7429	-0.6389	0.1873	0.3393	-0.6081	0.0229	1.0000		
(24) Population Growth	-0.5540	0.3000	0.0551	-0.4745	0.3282	0.6223	-0.6056	-0.2120	-0.1068	0.5722	0.2061	0.3944	-0.1494	-0.1352	-0.4944	0.4346	0.3695	-0.2565	-0.1445	0.1736	-0.1304	-0.2131	0.7559	1.0000	
(25) Trade Openness	0.2429	-0.0826	0.0106	-0.2668	-0.0455	0.0666	-0.1108	-0.0469	-0.0492	-0.0328	0.1877	-0.1968	-0.0978	-0.0591	0.1471	-0.0210	-0.2102	0.1276	-0.0952	-0.1360	0.1813	0.1052	-0.1617	0.0720	1.0000
(26) Women in Parliament	0.3658	-0.3560	0.2808	0.2592	-0.1311	-0.3609	0.2441	0.6302	0.2454	-0.5045	-0.0639	-0.1563	0.1000	0.5505	-0.1126	-0.2980	-0.3509	0.3919	-0.0670	-0.0407	0.3142	-0.2272	-0.4008	-0.3761	-0.0207
(27) Female Economic Participation	0.4606	-0.2820	0.2039	0.2096	-0.0465	-0.5175	0.3606	0.3561	0.1556	-0.8290	0.0965	-0.5689	0.1060	0.2026	0.4076	-0.3310	-0.2350	0.1509	0.3222	-0.0471	0.1483	-0.0404	-0.6591	-0.7132	0.0771
(28) Government Structure: Federalist	0.2688	-0.2036	0.0458	0.1259	-0.1927	-0.2501	0.1297	-0.0077	0.1228	-0.2874	0.1350	-0.1167	0.1765	-0.0270	-0.0793	-0.1120	-0.3388	0.3001	-0.1069	-0.0896	0.2843	-0.1773	-0.3440	-0.2382	-0.2237
(29) Property Rights Protection	0.6075	-0.3322	0.0792	0.2515	-0.3736	-0.3555	0.0226	0.5258	0.0962	-0.3640	0.3093	-0.2378	0.3223	0.3254	-0.3759	-0.2198	-0.6366	0.6065	-0.2282	-0.2863	0.7483	-0.2794	-0.4888	-0.1766	0.2189
(30) Freedom from Corruption	0.6109	-0.3324	0.0501	0.2590	-0.3938	-0.3624	-0.0616	0.6058	0.0462	-0.2939	0.2691	-0.2387	0.2751	0.4413	-0.3716	-0.3168	-0.6427	0.6584	-0.2792	-0.3267	0.7829	-0.2738	-0.4482	-0.0899	0.2454
(31) ln Number of Civil Conflicts	-0.4204	0.2771	0.2260	-0.3485	0.3672	0.3528	0.0465	-0.2254	-0.0552	0.3157	0.0704	0.2838	-0.1841	-0.1635	-0.2017	0.4033	0.4474	-0.4707	0.1907	0.2208	-0.4376	-0.0521	0.3779	0.1845	-0.2655
(32) No Education	-0.6018	0.4537	-0.0697	-0.3460	0.2608	0.6115	-0.3062	-0.2983	-0.2511	0.7632	0.0018	0.4750	-0.1635	-0.2050	-0.3529	0.4454	0.5874	-0.4932	0.1060	0.1163	-0.4128	-0.0080	0.7429	0.6694	-0.0999
(33) Primary Education	-0.0686	-0.2461	0.4014	-0.3281	0.2868	0.3663	0.2268	-0.1198	0.3793	-0.2891	-0.0538	0.2586	-0.0771	0.0254	-0.2220	0.1564	-0.0901	0.1067	-0.2056	0.2768	-0.1599	-0.3050	-0.0648	-0.0785	-0.0191
(34) Secondary Education	0.4548	-0.1600	-0.2595	0.4776	-0.3909	-0.6924	0.1073	0.2426	-0.0438	-0.3952	-0.0171	-0.5992	0.1755	0.0931	0.5665	-0.4502	-0.3028	0.2063	0.1571	-0.2713	0.3495	0.3003	-0.5224	-0.5212	0.1791
(35) ln Human Capital Index	0.6999	-0.4606	-0.0677	0.4935	-0.4115	-0.7481	0.2275	0.4324	0.0411	-0.5780	0.0384	-0.6032	0.3120	0.2501	0.3535	-0.5320	-0.6360	0.5493	-0.0704	-0.3564	0.5880	0.1049	-0.7378	-0.5944	0.1233
(36) Student-Teacher Ratio	-0.4439	0.3070	0.2224	-0.5008	0.4409	0.5331	-0.1044	-0.2371	0.0659	0.1304	0.1347	0.3066	0.0071	-0.2471	-0.3577	0.5634	0.4782	-0.4311	0.0197	0.4300	-0.5226	-0.3012	0.5097	0.3932	-0.0732
(37) Educational Expenditures	0.2463	0.0510	-0.1464	0.2907	-0.1065	-0.2975	-0.0966	0.5181	-0.2551	0.0626	0.2136	-0.3857	-0.0012	0.4155	0.0323	-0.3271	-0.1412	0.1098	0.0981	-0.2167	0.2102	0.2159	-0.0254	-0.0606	0.0532
(38) Private Enrollment Rate	-0.1877	-0.1051	0.1380	-0.2272	-0.0356	0.3226	-0.2088	-0.1215	0.2208	0.0555	0.0540	0.3873	-0.1578	-0.0926	-0.3622	0.2470	-0.0225	0.0559	-0.1793	0.1091	0.1332	-0.1554	0.1201	0.3232	0.0166
(39) Predicted Genetic Diversity	0.1134	0.1179	-0.5896	0.5345	-0.4171	-0.2858	0.1368	0.1177	-0.3793	0.3455	0.0331	-0.1992	-0.0788	0.0730	0.2115	-0.0615	-0.0573	0.0079	0.2021	-0.3840	0.2535	0.3981	-0.0892	-0.1101	0.0198
(40) HLA Genetic Diversity	0.4178	-0.1458	-0.3340	0.5444	-0.5410	-0.4281	0.1431	0.2982	0.0136	-0.0133	0.0243	-0.0985	0.1180	0.1933	-0.0812	-0.2850	-0.4269	0.4455	-0.0916	-0.3256	0.5559	0.0055	-0.3420	-0.2288	0.0162
(41) Ethnic Diversity	-0.4425	0.0836	0.1114	-0.3610	0.2093	0.2733	-0.2566	-0.1619	-0.0800	0.2604	0.1953	0.1196	-0.2361	-0.2986	-0.0208	0.2606	0.4165	-0.4298	0.2137	0.1829	-0.2065	0.0741	0.3249	0.3613	0.0956
(42) Religious Diversity	0.2105	-0.0062	0.0660	-0.0426	-0.0478	-0.1286	0.1151	0.2026	-0.1226	-0.3590	0.3963	-0.4881	0.2116	-0.1669	0.1691	-0.0053	-0.0986	-0.0107	0.2532	0.0013	0.1538	-0.0215	-0.2838	-0.2682	0.0907
(43) Region Dummy: Latin America and Caribbean	-0.4216	-0.0735	0.3201	-0.4163	0.1382	0.3837	-0.1673	-0.1793	0.5798	-0.2483	-0.1358	0.4405	-0.1262	-0.1121	-0.2565	0.1613	0.1713	-0.1004	-0.1256	0.6146	-0.2759	-0.3431	0.2214	0.1447	-0.2079
(44) Region Dummy: North America	0.1669	-0.0889	0.0660	0.1101	-0.0739	-0.2941	-0.0135	0.1741	0.0075	-0.0905	0.3445	-0.1524	-0.0455	-0.0404	-0.0925	-0.0653	-0.1350	0.1678	-0.0823	-0.0491	0.2132	-0.1674	-0.1056	-0.0123	-0.1417
(45) Region Dummy: Africa	-0.4341	0.3442	-0.1314	-0.0977	0.0844	0.2904	-0.0915	-0.0136	-0.2493	0.3311	0.1633	0.1219	-0.0951	-0.0845	-0.1933	0.4078	0.5552	-0.5182	0.4683	0.0268	-0.2994	-0.0000	0.4106	0.2285	-0.0669
(46) Region Dummy: East Asia	0.1231	0.0028	0.2268	-0.4504	0.4009	0.2323	-0.1184	-0.1547	-0.2413	-0.0633	0.1348	-0.1732	0.2282	-0.0903	-0.0095	0.1250	0.0767	-0.0847	-0.1006	-0.1146	-0.1859	-0.1479	0.0337	0.1337	0.2430
(47) Region Dummy: Central Asia	-0.1243	0.0179	-0.3894	0.1168	-0.0645	-0.2023	0.0404	-0.1242	-0.2208	0.2442	-0.1173	-0.0977	-0.0653	-0.0580	0.2946	0.0107	0.2619	-0.1890	0.0975	-0.0063	-0.1843	0.1864	0.0011	-0.1051	-0.0335
(48) Region Dummy: Middle East	-0.2132	0.3237	-0.2850	0.0087	0.0093	0.2561	-0.3640	-0.1573	-0.2721	0.6556	-0.0191	0.2293	-0.0811	-0.0721	-0.1649	0.0841	0.0298	-0.0559	-0.2367	-0.0749	0.0457	0.2734	0.4874	0.5946	-0.0193
(49) Region Dummy: Europe	0.6093	-0.3079	-0.0200	0.5785	-0.3676	-0.5788	0.4724	0.3084	0.1330	-0.4081	-0.1947	-0.3111	0.1042	0.2883	0.3303	-0.4756	-0.6073	0.4989	0.0001	-0.3434	0.4969	0.1870	-0.6815	-0.6241	0.1296
(50) Region Dummy: Oceania	0.1397	-0.0624	0.0859	0.0540	-0.0519	-0.0466	-0.0680	0.1288	-0.0540	-0.0659	0.2419	-0.1070	-0.0320	-0.0284	-0.0649	-0.0952	-0.0956	0.1171	-0.0773	-0.0588	0.0886	-0.0198	-0.0485	-0.0090	-0.0616
(**/ ***8*** - ********																									

Table 18: continued

										Table 1	18: contin	ued														
Variable	(26)	(27)	(28)	(29)	(30)	(31)	(32)	(33)	(34)	(35)	(36)	(37)	(38)	(39)	(40)	(41)	(42)	(43)	(44)	(45)	(46)	(47)	(48)	(49)	(50)	(51)
(1) SATS																										
(2) Indicator: TIMSS																										
(3) $DRD4^{R2R7}$																										
(4) In Absolute Latitude																										
(5) Population in Area with Malaria																										
(6) Temperature																										
(7) In Percentage Arable Land Area																										
(8) Major Religion Share: Protestantism																										
(9) Major Religion Share: Catholicism																										
(10) Major Religion Share: Islam																										
(11) Legal Origin: British																										
(12) Legal Origin: French																										
(13) Legal Origin: German																										
(14) Legal Origin: Scandinavian																										
(15) Legal Origin: Socialist																										
(16) Low Birth Weight																										
(17) Infant Mortality Rate																										
(18) Life Expectancy																										
(19) Disability-Adjusted Life-Years																										
(20) Homicide Rate																										
(21) ln GDP per Capita																										
(22) Government Consumption																										
(23) Total Fertility Rate																										
(24) Population Growth																										
(25) Trade Openness																										
(26) Women in Parliament	1.0000																									
(27) Female Economic Participation	0.4803	1.0000																								
(28) Government Structure: Federalist	0.0903		1.0000																							
(29) Property Rights Protection	0.4160	0.3009	0.2854	1.0000																						
(30) Freedom from Corruption	0.5141	0.2557	0.1702	0.9254	1.0000																					
(31) In Number of Civil Conflicts	-0.2009		-0.1225		-0.4388	1.0000																				
(32) No Education	-0.4849	-0.7559	-0.3587	-0.4110	-0.3520	0.4622	1.0000																			
(33) Primary Education	0.2468	0.0991		-0.0591	-0.1018	0.0963	-0.1866	1.0000																		
			0.0833					-0.5239	1.0000																	
(34) Secondary Education	0.1812	0.5456	0.1628	0.2806	0.2502		-0.6313		0.7840	1.0000																
(35) In Human Capital Index	0.4236	0.6554				-0.4927		-0.2206																		
(36) Student-Teacher Ratio	-0.1985 0.3258	-0.3242 0.0704			-0.3287 0.4122	0.4685 -0.1732	0.3901			-0.5633	1.0000 -0.2770	1.0000														
(37) Educational Expenditures				0.2828				-0.2855	0.2387																	
(38) Private Enrollment Rate	-0.0208	-0.2284		0.1838		-0.0056			-0.1766		0.1100		1.0000													
(39) Predicted Genetic Diversity	-0.0732		-0.0158	0.0511		-0.1967		-0.4020	0.2980			0.2494		1.0000												
(40) HLA Genetic Diversity	0.3463	0.0748	0.1052	0.4457	0.5281	-0.3937	-0.1777	-0.2243	0.3100	0.3353	-0.4911	0.3932	-0.0805	0.6620	1.0000											
(41) Ethnic Diversity	-0.2580	-0.1849	-0.1336	-0.3083	-0.2919	0.3190	0.2907		-0.1338	-0.2553		-0.1078		-0.0545		1.0000										
(42) Religious Diversity	0.1183	0.4659	0.2013	0.2307	0.1818	-0.1865	-0.3577	-0.1645		0.4298	-0.0176	0.1066	0.0124	0.0525	0.0856	0.1631	1.0000									
(43) Region Dummy: Latin America and Caribbean	0.0368	-0.0738		-0.1892	-0.2370	0.1008	-0.0157			-0.2516		-0.3591	0.3464		-0.3777		-0.1148	1.0000								
(44) Region Dummy: North America	0.0185	0.1162	0.2555	0.2286	0.2313	-0.1139	-0.1443	-0.2915	0.1628	0.2401	-0.0154	0.1710	-0.0385	0.0003	0.1778	0.1881	0.2627	-0.0781	1.0000							
(45) Region Dummy: Africa	-0.1421	-0.2660		-0.1403	-0.1210	0.2887	0.6010		-0.2229	-0.4791	0.2858	0.2473		0.2524	0.0402	0.1006		-0.1633		1.0000						
(46) Region Dummy: East Asia	-0.1579	0.0888	-0.0647		-0.1001		-0.0050		-0.1492			-0.1514		-0.4799		0.0009		-0.1746		-0.1316						
(47) Region Dummy: Central Asia	-0.1539	0.0582		-0.2285	-0.2636	-0.0110		-0.1246	0.1863			-0.2115		0.1355			-0.0193				-0.0903					
(48) Region Dummy: Middle East	-0.4001	-0.7155	-0.1965	-0.1948	-0.1153	0.0315	0.4551	-0.2235	-0.2261	-0.3190	0.0024	0.0222	0.2401	0.2876	0.0209	0.2853	-0.1818	-0.1393	-0.0502	-0.1050	-0.1122	-0.0721	1.0000			
(49) Region Dummy: Europe	0.4184	0.4823		0.3818			-0.5149	-0.0621	0.5077		-0.6188		-0.1964		0.5094		0.0420				-0.3132	-0.2011	-0.2500			
(50) Region Dummy: Oceania	0.1573	0.0785		0.1751	0.2260	-0.0800	-0.1058		-0.0799		-0.0233	0.0998	-0.0393	-0.0621	0.0245	0.0224	0.2102		-0.0198		-0.0442		-0.0353		1.0000	
(51) OPEC Dummy	-0.3645	-0.6119	-0.1965	-0.2294	-0.1230	0.1634	0.4881	-0.1819	-0.2649	-0.3766	-0.0194	-0.0117	0.1734	0.1154	-0.0545	0.2721	-0.2358	-0.1393	-0.0502	0.0547	0.0395	-0.0721	0.6368	-0.2500	-0.0353	1.0000

 ${\bf Table\ 19:\ Summary\ Statistics\ for\ the\ PISA\ Country\ Sample}$ 

Variable	N	Mean	SD	Minimum	Maximum
SATS: PISA Mathematics and Science Score	63	4.5485	0.5988	3.2333	5.5807
SATS: PISA Mathematics Score	63	4.5256	0.6079	3.2073	5.6661
SATS: PISA Science Score	63	4.5837	0.5851	3.2574	5.4935
$DRD4^{R2R7}$	63	0.2378	0.0617	0.1076	0.4583
n Absolute Latitude	63	3.4181	0.8004	0.3124	4.1589
Population in Area with Malaria	63	0.0645	0.1807	0	0.9539
Temperature	63	13.1427	8.3394	-7.9294	26.9546
n Percentage Arable Land Area	63	2.5123	0.9404	0.1131	3.9845
Major Religion Share: Protestantism	63	13.3524	24.5111	0	97.8000
Major Religion Share: Catholicism	63	38.4397	38.4874	0	96.9000
Major Religion Share: Islam	63	15.434	29.9561	0	99.4000
Legal Origin: British	63	0.1587	0.3684	0	1
Legal Origin: French	63	0.4127	0.4963	0	1
Legal Origin: German	63	0.0794	0.2725	0	1
Legal Origin: Scandinavian	63	0.0635	0.2458	0	1
Legal Origin: Socialist	63	0.2857	0.4554	0	1
Low Birth Weight	63	7.5131	2.2285	4.1000	14.1429
Infant Mortality Rate	63	14.0261	11.0949	3.1615	53.1154
Life Expectancy	63	73.7934	3.8593	65.0924	80.3303
Disability-Adjusted Life-Years	63	0.3089	0.0678	0.1793	0.5602
Homicide Rate	63	4.9193	7.2414	0.4333	42.4533
n GDP per Capita	63	9.6377	0.7070	7.7625	11.1580
Government Consumption	63	0.1931	0.0584	0.0743	0.3407
Total Fertility Rate	63	2.1129	0.7404	1.3262	4.8534
Population Growth	63	0.8945	1.1698	-1.2606	6.7407
Trade Openness	63	85.0844	52.4673	21.1660	356.3131
Women in Parliament	63	18.7248	9.4059	0	44.1650
Female Economic Participation	63	40.7632	7.9675	14.5475	48.6768
Government Structure: Federalist	63	0.3333	0.4752	0	1
Property Rights Protection	63	61.1878	22.7514	11.4286	91.7500
Freedom from Corruption	63	53.8541	23.3216	20.075	94.7350
n Number of Civil Conflicts	63	-2.6142	3.0438	-4.6052	3.5838
Predicted Genetic Diversity	63	0.7229	0.0226	0.6434	0.7518
HLA Genetic Diversity	63	0.3260	0.0179	0.2649	0.3529
Ethnic Diversity	63	0.3401	0.2171	0.0020	0.7456
Religious Diversity	63	0.4136	0.2350	0.0049	0.8241
Region Dummy: Latin America and Caribbean	63	0.1746	0.3827	0	1
Region Dummy: North America	63	0.0317	0.1767	0	1
Region Dummy: Africa	63	0.0476	0.2147	0	1
Region Dummy: East Asia	63	0.1111	0.3168	0	1
Region Dummy: Central Asia	63	0.0794	0.2725	0	1
Region Dummy: Middle East	63	0.0476	0.2147	0	1
Region Dummy: Europe	63	0.4762	0.5034	0	1
Region Dummy: Oceania	63	0.0317	0.1767	0	1
OPEC Dummy	63	0.0476	0.2147	0	1

 ${\bf Table~20:~Pairwise~Correlations~for~the~PISA~Country~Sample}$ 

Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)	(19)	(20)	(21)	(22)
(1) SATS: PISA Mathematics and Science Score	1.0000																					
(2) SATS: PISA Mathematics Score	0.9914	1.0000																				
(3) SATS: PISA Science Score	0.9907	0.9698	1.0000																			
(4) $DRD4^{R2R7}$	0.0126	-0.0104	0.0594	1.0000																		
(5) In Absolute Latitude	0.3451	0.3468	0.3306	-0.4418	1.0000																	
(6) Population in Area with Malaria	-0.1713	-0.1951	-0.1455	0.5522	-0.5199	1.0000																
(7) Temperature	-0.4723	-0.4823	-0.4523	0.3355	-0.7875	0.4778	1.0000															
(8) In Percentage Arable Land Area	0.2502	0.2653	0.2192	-0.0482	0.4773	-0.0584	-0.2613	1.0000														
(9) Major Religion Share: Protestantism	0.4502	0.4453	0.4418	0.0436	0.3125	-0.1725	-0.4272	-0.0155	1.0000													
(10) Major Religion Share: Catholicism	-0.0284	-0.0215	0.0025	0.3472	-0.0706	-0.0376	0.1300	0.0602	-0.2650	1.0000												
(11) Major Religion Share: Islam	-0.4991	-0.4801	-0.5245	-0.4050	-0.1065	0.0066	0.2191	-0.2543	-0.2650	-0.4651	1.0000											
(12) Legal Origin: British	0.2460	0.2233	0.2409	0.2197	-0.2815	0.0609	0.1168	-0.1979	0.0637	-0.1222	-0.1084	1.0000										
(13) Legal Origin: French	-0.4643	-0.4672	-0.4298	0.1856	-0.2489	0.0963	0.4775	-0.1436	-0.3493	0.4479	0.2069	-0.3641	1.0000									
(14) Legal Origin: German	0.3271	0.3401	0.3091	-0.0744	0.1289	-0.1056	-0.1438	0.0953	0.1025	-0.0171	-0.1507	-0.1275	-0.2461	1.0000								
(15) Legal Origin: Scandinavian	0.2351	0.2392	0.2242	-0.0413	0.2268	-0.0936	-0.3174	-0.0928	0.8060	-0.2581	-0.1344	-0.1131	-0.2183	-0.0764	1.0000							
(16) Legal Origin: Socialist	-0.0156	-0.0040	-0.0324	-0.3132	0.2995	-0.0405	-0.3576	0.3096	-0.1672	-0.2397	0.0248	-0.2747	-0.5302	-0.1857	-0.1647	1.0000						
(17) Low Birth Weight	-0.4246	-0.3997	-0.4449	0.0492	-0.4689	0.1842	0.5742	-0.0182	-0.4000	0.0409	0.3108	0.1273	0.3623	-0.1173	-0.3299	-0.2496	1.0000					
(18) Infant Mortality Rate	-0.7581	-0.7181	-0.7892	-0.1313	-0.2732	0.3218	0.3282	-0.0871	-0.3841	-0.1964	0.5808	-0.2071	0.1924	-0.2599	-0.2426	0.2443	0.4326	1.0000				
(19) Life Expectancy	0.6645	0.6439	0.6716	0.0882	0.2795	-0.3045	-0.2573	0.0205	0.3726	0.2089	-0.4713	0.1857	-0.0663	0.3027	0.2743	-0.4071	-0.3977	-0.8526	1.0000			
(20) Disability-Adjusted Life-Years	-0.1194	-0.1092	-0.1296	-0.1961	0.2923	-0.0340	-0.3629	0.3592	-0.0331	-0.2042	0.0082	-0.2397	-0.2746	-0.1904	-0.0583	0.6386	0.0125	0.3606	-0.6150	1.0000		
(21) Homicide Rate	-0.4916	-0.5008	-0.4733	0.0373	-0.4351	0.2664	0.3736	-0.2923	-0.1675	0.2860	-0.0608	-0.0138	0.2505	-0.1722	-0.1366	-0.0851	0.3114	0.3889	-0.4294	0.1378	1.0000	
(22) ln GDP per Capita	0.6717	0.6748	0.6598	-0.0562	0.3265	-0.4810	-0.3557	0.0024	0.4611	0.1045	-0.3076	0.2697	-0.1661	0.2721	0.2952	-0.3593	-0.2387	-0.7784	0.7599	-0.3184	-0.2999	1.0000
(23) Government Consumption	-0.1663	-0.1665	-0.1756	-0.4296	0.3354	-0.2360	-0.2853	0.1551	-0.0719	-0.3677	0.3088	-0.2580	-0.2315	-0.3091	0.0046	0.6435	-0.0940	0.2038	-0.3631	0.5325	-0.1760	-0.3174
(24) Total Fertility Rate	-0.7153	-0.7310	-0.6821	0.0872	-0.4209	0.3318	0.5700	-0.5015	-0.2726	-0.0447	0.6125	-0.0592	0.4968	-0.2591	-0.1233	-0.2718	0.4032	0.5918	-0.4546	-0.2229	0.3281	-0.5220
(25) Population Growth	-0.3941	-0.3928	-0.3771	0.0910	-0.4805	0.2301	0.5812	-0.5984	-0.1976	-0.0058	0.5311	0.1225	0.4608	-0.1092	-0.0997	-0.4821	0.3723	0.2070	-0.0498	-0.5606	0.1767	-0.0215
(26) Trade Openness	0.1947	0.2211	0.1663	0.0033	-0.3049	-0.0161	0.1162	-0.0970	-0.0591	-0.0675	0.0654	0.1844	-0.1529	-0.1065	-0.0662	0.1170	0.0708	-0.1618	0.0619	-0.1338	-0.1910	0.1255
(27) Women in Parliament	0.4314	0.4290	0.4229	0.2274	0.2755	-0.1059	-0.3228	0.1192	0.6343	0.1534	-0.4279	-0.0304	-0.1501	0.0713	0.5775	-0.1662	-0.4079	-0.3880	0.4707	-0.1269	-0.1993	0.3546
(28) Female Economic Participation	0.5220	0.5428	0.4940	0.1163	0.2820	-0.0241	-0.5227	0.3944	0.3354	0.0358	-0.6615	0.1203	-0.6082	0.0821	0.2101	0.4031	-0.3723	-0.2320	0.1595	0.4004	-0.1148	0.1545
(29) Government Structure: Federalist	0.2140	0.1925	0.2411	0.1191	0.0528	-0.1003	-0.1426	-0.0139	0.0155	0.1641	-0.2836	0.2457	-0.0456	0.1661	-0.0460	-0.2236	0.0228	-0.2694	0.2379	-0.0797	0.0437	0.2727
(30) Property Rights Protection	0.7061	0.6975	0.7096	0.1127	0.2181	-0.3523	-0.3028	-0.0041	0.5463	0.1113	-0.3772	0.3886	-0.1454	0.3134	0.3188	-0.5155	-0.2163	-0.7533	0.7273	-0.4160	-0.3193	0.8062
(31) Freedom from Corruption	0.6962	0.6840	0.6991	0.1154	0.2159	-0.3399	-0.2975	-0.1019	0.6206	0.0646	-0.3311	0.3481	-0.1673	0.2573	0.4309	-0.4859	-0.3140	-0.7591	0.8074	-0.4877	-0.3780	0.8308
(32) In Number of Civil Conflicts	-0.4260	-0.4193	-0.4353	0.1248	-0.3118	0.3739	0.2600	-0.0717	-0.2926	-0.1164	0.3318	0.0656	0.1622	-0.1936	-0.1717	-0.0213	0.2828	0.4760	-0.4952	0.2270	0.2723	-0.3972
(33) Predicted Genetic Diversity	0.0842	0.0883	0.0494	-0.5608	0.5921	-0.5490	-0.3908	0.3564	0.1000	-0.2460	0.2670	-0.1392	-0.1805	-0.0956	0.0797	0.3235	-0.0676	-0.0818	0.0425	0.3079	-0.2569	0.2200
(34) HLA Genetic Diversity	0.4293	0.4185	0.4026	-0.2699	0.5189	-0.5891	-0.4170	0.2213	0.3043	0.1024	-0.1083	0.0357	-0.0734	0.0986	0.1808	-0.1055	-0.3043	-0.4776	0.5373	-0.0870	-0.3130	0.5723
(35) Ethnic Diversity	-0.4633	-0.4601	-0.4482	0.2149	-0.4042	0.2965	0.2725	-0.3202	-0.2358	0.0014	0.2074	0.1590	0.1207	-0.2399	-0.3107	0.0512	0.3475	0.3996	-0.4565	0.1382	0.4245	-0.2686
(36) Religious Diversity	0.2976	0.3019	0.2622	-0.0391	-0.0071	-0.1720	-0.1666	0.0915	0.1121	-0.1649	-0.2879	0.3875	-0.4769	0.2047	-0.2036	0.1936	-0.0308	-0.1653	0.1209	0.0666	-0.0495	0.1392
(37) Region Dummy: Latin America and Caribbean	-0.4861	-0.4914	-0.4521	0.2643	-0.4321	0.1691	0.4413	-0.3414	-0.1822	0.5393	-0.2227	-0.0854	0.4637	-0.1350	-0.1198	-0.2909	0.2913	0.2721	-0.2161	-0.1431	0.7025	-0.2968
(38) Region Dummy: North America	0.1607	0.1470	0.1648	0.0715	0.1021	-0.0651	-0.3030	-0.0442	0.1731	-0.0007	-0.0898	0.4169	-0.1518	-0.0532	-0.0471	-0.1145	-0.0379	-0.1283	0.1773	-0.1045	-0.0348	0.2305
(39) Region Dummy: Africa	-0.2481	-0.2578	-0.2456	-0.1528	-0.0362	-0.0804	0.2282	0.0270	-0.1200	-0.1630	0.4229	-0.0971	0.2667	-0.0657	-0.0582	-0.1414	0.1200	0.1981	-0.2032	-0.0718	-0.0862	-0.1504
(40) Region Dummy: East Asia	0.1630	0.1553	0.1564	0.2683	-0.5346	0.5638	0.3708	-0.1026	-0.1478	-0.3308	0.0120	0.2611	-0.1938	0.2699	-0.0921	-0.1118	0.1087	0.0187	-0.0557	-0.1957	-0.1563	-0.1714
(41) Region Dummy: Central Asia	-0.2885	-0.2394	-0.3487	-0.5102	0.1181	-0.0483	-0.1866	0.0273	-0.1564	-0.2893	0.4810	-0.1275	-0.1268	-0.0862	-0.0764	0.3343	0.1180	0.5798	-0.4315	0.3097	0.0649	-0.3133
(42) Region Dummy: Middle East	-0.2507	-0.2534	-0.2413	-0.2253	-0.0057	-0.0804	0.1947	-0.3294	-0.1160	-0.1488	0.4427	-0.0971	0.2667	-0.0657	-0.0582	-0.1414	0.2257	0.0393	-0.0162	-0.1932	-0.0174	-0.0045
(43) Region Dummy: Europe	0.5103	0.5098	0.5138	0.0025	0.5760	-0.3429	-0.5652	0.5037	0.3107	0.1118	-0.4034	-0.2402	-0.2828	0.0728	0.2731	0.3116	-0.4507	-0.5398	0.3927	0.2710	-0.3818	0.4321
(44) Region Dummy: Oceania	0.1967	0.1765	0.1992	0.1142	0.0197	-0.0651	0.0668	-0.1424	0.1292	-0.0678	-0.0934	0.4169	-0.1518	-0.0532	-0.0471	-0.1145	-0.1055	-0.1393	0.2037	-0.1625	-0.0928	0.1597
(45) OPEC Dummy	-0.3001	-0.3066	-0.2830	0.1265	-0.1947	0.1700	0.3257	-0.3059	-0.1053	-0.2165	0.4730	-0.0971	0.2667	-0.0657	-0.0582	-0.1414	0.0779	0.2582	-0.2184	-0.0757	-0.0607	-0.0203

#### Table 20: continued

Variable	(23)	(24)	(25)	(26)	(27)	(28)	(29)	(30)	(31)	(32)	(33)	(34)	(35)	(36)	(37)	(38)	(39)	(40)	(41)	(42)	(43)	(44)	(45)
(1) SATS: PISA Mathematics and Science Score																							
(2) SATS: PISA Mathematics Score																							
(3) SATS: PISA Science Score																							
(4) $DRD4^{R2R7}$																							
(5) In Absolute Latitude																							
(6) Population in Area with Malaria																							
(7) Temperature																							
(8) In Percentage Arable Land Area																							
(9) Major Religion Share: Protestantism																							
(10) Major Religion Share: Catholicism																							
(11) Major Religion Share: Islam																							
(12) Legal Origin: British																							
(13) Legal Origin: French																							
(14) Legal Origin: German																							
(15) Legal Origin: Scandinavian																							
(16) Legal Origin: Socialist																							
(17) Low Birth Weight																							
(18) Infant Mortality Rate																							
(19) Life Expectancy																							
(20) Disability-Adjusted Life-Years																							
(21) Homicide Rate																							
(22) ln GDP per Capita																							
(23) Government Consumption	1.0000																						
(24) Total Fertility Rate	0.0016	1.0000																					
(25) Population Growth	-0.2444	0.7301	1.0000																				
(26) Trade Openness	0.1242	-0.1113	0.1178	1.0000																			
(27) Women in Parliament	-0.2243	-0.3368	-0.2997	-0.0476	1.0000																		
(28) Female Economic Participation	-0.0317	-0.7343	-0.7538	-0.0106	0.4375	1.0000																	
(29) Government Structure: Federalist	-0.2562	-0.1782	-0.0964	-0.3110	0.0876	0.1274	1.0000																
(30) Property Rights Protection	-0.3542	-0.4278	-0.0771	0.1454	0.4177	0.1920	0.3169	1.0000															
(31) Freedom from Corruption	-0.3532	-0.3810	0.0050	0.1848	0.5375	0.1893	0.1985	0.9277	1.0000														
(32) In Number of Civil Conflicts	0.0620	0.2822	0.0960	-0.1795	-0.2930	-0.2516	-0.1284	-0.4259	-0.4591	1.0000													
(33) Predicted Genetic Diversity	0.4593	-0.2365	-0.2288	0.0466	-0.0029	-0.0386	-0.0774	-0.0073	0.0222	-0.1855	1.0000												
(34) HLA Genetic Diversity	-0.0312	-0.3772	-0.2162	-0.0135	0.3874	0.1192	0.0881	0.4288	0.5066	-0.4147	0.6586	1.0000											
(35) Ethnic Diversity	0.0479	0.3787	0.3023	0.1693	-0.2868	-0.1790	-0.1225	-0.2878	-0.3014	0.2657	-0.2012	-0.3934	1.0000										
(36) Religious Diversity	0.0448	-0.3977	-0.2825	0.0678	0.0126	0.3759	0.1964	0.1621	0.1190	-0.1686	0.0382	0.1077	0.0245	1.0000									
(37) Region Dummy: Latin America and Caribbean	-0.3465	0.3982	0.2287	-0.2292	-0.0102	-0.1546	0.2070	-0.1906	-0.2355	0.1180	-0.5137	-0.3625	0.2248	-0.1392	1.0000								
(38) Region Dummy: North America	-0.1822	-0.0771	0.0216	-0.1484	-0.0042	0.1145	0.2561	0.2235	0.2213	-0.1194	-0.0027	0.1710	0.2196	0.2690	-0.0833	1.0000							
(39) Region Dummy: Africa	0.0472	0.2627	0.1214	0.0148	-0.1083	-0.4425	-0.1581	-0.1109	-0.0853	0.1596	0.1259	0.0137	-0.0616	-0.1864	-0.1028	-0.0405	1.0000						
(40) Region Dummy: East Asia	-0.2461	0.0378	0.1670	0.2603	-0.2080	0.0295	-0.0357	-0.0389	-0.0632	0.1092	-0.5000	-0.4276	0.0503	0.1015	-0.1626	-0.0640	-0.0791	1.0000					
(41) Region Dummy: Central Asia	0.2277	0.0985	-0.0732	-0.0227	-0.2230	0.0836	-0.2076	-0.3389	-0.3578	0.2035	0.1799	-0.1193	0.1660	0.0296	-0.1350	-0.0532	-0.0657	-0.1038	1.0000				
(42) Region Dummy: Middle East	0.3098	0.4692	0.6239	0.0627	-0.3803	-0.6565	-0.1581	-0.1107	-0.0868	0.0060	0.2551	0.0149	0.1555	-0.0932	-0.1028	-0.0405	-0.0500	-0.0791	-0.0657	1.0000			
(43) Region Dummy: Europe	0.2395	-0.6496	-0.5819	0.0875	0.4146	0.4505	-0.0000	0.2852	0.3101	-0.2553	0.4610	0.5012	-0.3814	-0.0591	-0.4385	-0.1726	-0.2132	-0.3371	-0.2799	-0.2132	1.0000		
(44) Region Dummy: Oceania	-0.0932	-0.0433	0.0490	-0.1302	0.1554	0.0905	0.0640	0.2382	0.2791	-0.1194	-0.0420	0.1017	-0.0800	0.3126	-0.0833	-0.0328	-0.0405	-0.0640	-0.0532	-0.0405	-0.1726	1.0000	
(45) OPEC Dummy	0.0030	0.4147	0.4982	-0.0758	-0.2579	-0.5221	-0.1581	-0.1829	-0.1007	0.2378	-0.0192	-0.0970	0.2768	-0.2887	-0.1028	-0.0405	0.3000	0.1581	-0.0657	0.3000	-0.2132	-0.0405	1.0000

# C Data Description and Sources

# Dependent Variable

Aggregate Student Achievement Test Score Outcome. This variable refers to the country-level aggregate student achievement test score outcome during the years 1995 to 2015. The variable is constructed based on International Student Achievement Tests (ISATs) of the various PISA (i.e., PISA 2000, 2003, 2006, 2009, 2012, and 2015) and TIMSS (i.e., TIMSS 1995, 1999, 2003, 2007, 2011, and 2015) assessment waves.

The first step in the construction of this variable involves the aggregation of individual student-level test score outcomes by country and wave, respectively, for the main academic subjects mathematics and science. It is worth mentioning that we disregard the construction of reading achievement test score outcomes, as this subject is only available in the PISA but not TIMSS assessment waves. The aggregation is performed on behalf of the provided population weights in both ISATs in order to maintain representativity of aggregated student achievement test score outcomes in the general population. This procedure yields an aggregate value for average student achievement for each combination of the ISAT, assessment wave, and academic subject. Then, we calculate country-level averages of the mathematics and science test score outcomes across the available set of PISA and TIMSS assessment waves, as proposed by Rindermann (2008).

Once country-level averages of student test score outcomes across the main subjects mathematics and science have been constructed, we take the unweighted mean of both values to arrive at a combined student achievement test score outcome. If available, we give preference to the aggregate PISA test score outcome. In order to increase the sample size, we use the aggregate TIMSS test score outcome in cases with missing country-level PISA test score observations. To avoid a possible bias through systematic differences between scores in the two international student achievement tests, we include a re-codification indicator that equals 1 if the respective test score outcome originates from TIMSS and zero otherwise throughout the empirical analysis.

The corresponding data on international student achievement test score outcomes of PISA and TIMSS are freely available online at <a href="http://www.oecd.org/pisa/">http://www.oecd.org/pisa/</a> and <a href="https://timssandpirls.bc.edu/">https://timssandpirls.bc.edu/</a>, respectively.

## Genetic Variables

DRD4 Exon III 2- and 7-Repeat Allele Frequency. This variable refers to the country-level DRD4 exon III 2- and 7-repeat allele frequency measure. The construction of this variable is based on population genome data of DRD4 exon III allele frequencies across a worldwide sample of 120 populations in 59 countries. The raw data have been compiled by Gören (2016) from an extensive list of individual molecular genetic, population, and candidate gene studies of DRD4 exon III polymorphism. Since contemporary countries are generally comprised of various ethnic groups on at least some of which population genome data of DRD4 exon III allele frequencies have not yet been compiled, a matching scheme is needed for assigning ethnic groups to the population genome data in Gören (2016). The methodology developed in Gören (2017a) assigns the entire distribution of ethnic groups in the Alesina et al. (2003) ethnicity data to the population genome data in Gören (2016) based on the historical relationship between ethno-linguistic groups, as provided by the Ethnologue database (Global Mapping International, 2010). In cases where the Alesina et al. (2003) ethnicity data refers to universal ethnic groups (e.g., 'black' or 'white') or to groups of mixed ancestry (e.g., 'Mestizo' or 'Amerindian'), the corresponding genetic composition of DRD4 exon III allele frequencies were derived through combinations of population genome data in Gören (2016). After successful completion of the matching procedure, the ethnic-specific DRD4 exon III 2- and 7-repeat allele frequency measures were aggregated to the country level using the corresponding shares of ethnic groups within a particular country as the weighting scheme. See Gören (2017a) for additional details on data construction and sources.

Genetic Diversity. This variable refers to a country-level measure of overall genetic diversity. The calculation of this variable

is based on the predictive power of migratory distance from East Africa on between-population genetic diversity (expressed as expected heterozygosity). This association has been uncovered by Ramachandran et al. (2005) based on a sample of 53 indigenous populations from the Human Genome Diversity Cell Line Panel from the Human Genome Diversity Project-Centre d'Etude du Polymorphisme Humain (HGDP-CEPH) (Cann et al., 2002). This finding is in strong support of a serial founder effect since the pre-historic exodus of anatomically modern humans originating from East Africa – i.e., Addis Ababa, Ethiopia (9N, 38E) – and subsequently inhabiting the entire world through five intermediate land-restricted migration routes, i.e., Cairo, Egypt (30N, 31E), Istanbul, Turkey (41N, 28E), Phnom Penh, Cambodia (11N, 104E), Anadyr, Russia (64N, 177E), and Prince Rupert, Canada (54N, 130W). The argumentation is that populations further away from the cradle of humanity – i.e., Addis Ababa, Ethiopia (9N, 38E) – show a concomitant decrease in their level of genetic diversity, which is in line with the hypothesis that subsequent founder populations only carry with them a fraction of their parents' genetic composition.

The empirical evidence regarding a serial founder effect uncovered in Ramachandran et al. (2005) were used by Ashraf and Galor (2013) in the construction of their country-level genetic diversity measure. To accomplish this task, Ashraf and Galor (2013) first construct the migratory distance from East Africa to the country's modern capital city, restricting the pre-historic migration routes through the aforementioned five intermediate way points. In the second step, the country-level expected heterozygosity estimate were inferred from the regression coefficient associated with migratory distance from East Africa in the original 53 HGDP-CEPH population sample. Given the fact that contemporary countries have undergone large population movements since the post-Columbian period (i.e., 1500 AD), Ashraf and Galor (2013) propose the use of an ancestry-adjusted version of the country-level genetic diversity measure. This variable refers to a weighted estimate of the various country-level genetic diversity measures, where the weights correspond to the shares of a country's contemporary population that can trace their ancestry back to countries prior to the post-Columbian era (i.e., prior to the year 1500 AD). See Ashraf and Galor (2013) for additional details on data construction and sources.

HLA Genetic Diversity. This variable refers to a country-level genetic diversity measure (expressed as expected heterozygosity) associated with a set of genes located in the human leukocyte antigen (HLA) region. The construction of this variable is based on population genome data for 156 single-nucleotide polymorphisms (SNPs) within 19 HLA genes in a sample of 51 distinct populations, as provided by the Allele Frequency Database (ALFRED) (Rajeevan et al., 2012). Next, the distribution of the relevant ethnolinguistic groups within countries in the Alesina et al. (2003) ethnicity data were matched to the set of 51 populations for which HLA heterozygosity data have been provided by ALFRED. This matching procedure employs the historical relationship between ethnic groups, as indicated by the Ethnologue database (Global Mapping International, 2010). It is worth mentioning that the HLA genetic composition of additional ethnic groups (e.g., black or mestizo) were constructed through combinations of HLA population genome data. Then, the ethnic-specific HLA expected heterozygosity measures were aggregated to the country level using the corresponding ethnic population shares as the weighting scheme. See Cook (2015) for additional details on data construction and sources.

## **Diversity Variables**

Ethnic Diversity. This variable refers to a country-level measure of ethno-linguistic diversity, calculated using a slight variation of the Herfindahl-Hirschman concentration index based on population shares of the various ethno-linguistic groups within a particular country. This index is restricted to the unit interval, expressing the probability that two randomly selected individuals within a particular country will belong to entirely different ethnic groups. Thus, higher values of this index would indicate a more ethnically diverse country. See Alesina et al. (2003) for additional details on data construction and sources.

Religious Diversity. This variable refers to a country-level measure of religious diversity, calculated using a slight variation of the Herfindahl-Hirschman concentration index based on population shares of the various religious groups within a particular country.

This index is restricted to the unit interval, expressing the probability that two randomly selected individuals within a particular country will belong to entirely different religious groups. Thus, higher values of this index would indicate a country with greater religious diversity. See Alesina et al. (2003) for additional details on data construction and sources.

## Biogeographic Variables

**Absolute Latitude.** This variable refers to the country's approximate centroid absolute latitude in decimal degrees. This variable is part of *The Center for International Development* research datasets at Harvard University. The corresponding data are available online at <a href="http://www.cid.harvard.edu/ciddata/ciddata.html">http://www.cid.harvard.edu/ciddata/ciddata.html</a>.

Population in Area with Malaria. This variable refers to the share of a country's population residing in areas subject to malaria falciparum. The variable combines historical maps of malaria prevalence with data on the distribution of the world's population in the year 1994. The values range from 0 (no malaria prevalence) to 100 (high malaria prevalence). This variable is part of *The Center for International Development* research datasets at Harvard University. The corresponding data are available online at <a href="http://www.cid.harvard.edu/ciddata/ciddata.html">http://www.cid.harvard.edu/ciddata/ciddata.html</a>.

Temperature. This variable refers to the country's mean temperature value (in Degree Celsius) during the period 1960 to 1990. The raw data are constructed using the G-ECON database with a spatial resolution of 1 decimal degree latitude × longitude from monthly temperature observations during the period 1960 to 1990 provided by the Climatic Research Unit database at the University of East Anglia. See Nordhaus (2006); New et al. (2002) for additional details on data construction and sources. The corresponding country-level temperature values across 1 decimal degree grid cells within the country's national borders have been constructed by Ashraf and Galor (2013).

Percentage Arable Land Area. This variable measures the percentage of a country's arable land area that meets the criteria of the Food and Agricultural Organization of the United Nations (FAO), which are defined as followed: "Arable land refers to land under temporary crops (double cropped areas are counted only once), temporary meadows for mowing or pasture, land under market and kitchen gardens and land temporarily fallow (less than five years). The abandoned land resulting from shifting cultivation is not included". This variable is part of the World Bank's World Development Indicators database. The corresponding data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

Region Fixed Effects. This variable refers to a set of indicator variables that equals 1 if the respective country is located in one of the main regional classifications according to the definition of the World Bank (i.e., North America, Central and South America, Sub-Saharan Africa, East Asia, Central Asia, South Asia, Middle East, Europe, or Oceania) and zero otherwise.

### Historical Variables

**Legal Origin Effects.** This variable refers to a vector of indicator variables that equals 1 if the respective country's legal origin is British, French, German, Scandinavian, or Socialist and zero otherwise. The corresponding data are available from the La Porta et al. (1999) study.

Major Religion Effects. This variable refers to the set of a country's major religion shares for Protestants, Catholics, and Muslims (as percentage of population in 1980). The residual indicates the membership to any other – or no religion at all. The

corresponding data are available from the La Porta et al. (1999) study.

Agricultural Transition Timing. This variable refers to the time elapsed between when the country began practicing sedentary agriculture and the year 2000. See Putterman (2008) for additional details regarding data construction and sources. This variable has been ancestry-adjusted to correct for international migration flows since the year 1500 AD based on the World Migration Matrix data, as reported by Putterman and Weil (2010).

State History. This variable measures the experience of present-day countries with early state formation. The data for the ancestry-adjusted measure of state history from 1 AD to 1500 AD is constructed as follows: The data on state history, as discussed in Chanda and Putterman (2007), begin in 1 AD and continue in half-century intervals until the year 1500 AD. When constructing the index, more weights are assigned to most recent half-centuries of state experiences, and earlier half-centuries are discounted by 5% (e.g.,  $\frac{1}{[1.05]}$ ,  $\frac{1}{[1.05]^2}$ , etc.).

This measure is then normalized to be in the range of 0 (the lowest possible experience of early state formation) and 1 (the highest possible experience with early state formation) for ease of interpretation. Finally, an ancestry-adjusted measure of early state formation for each country is constructed based on post-1500 AD international migration flows data, as reported by Putterman and Weil (2010).

#### Health Variables

Percentage of Low Birth Weight. This variable refers to the percentage of new-born babies with a birth weight of less than 2,500 grams. According to the World Health Organization International Classification of Diseases criteria, this threshold denotes abnormally low weight at birth as a result of being born too soon or of slow growth, i.e., being too small for gestational age. The data are downloaded from the World Bank's *Health Nutrition and Population Statistics* database and were originally gathered as part of the UNICEF State of the World's Children reports. The data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

Infant Mortality Rate. This variable measures the number of infant deaths for both sexes in the first year after birth per 1,000 live births. The data are provided as part of the World Bank's World Development Indicators database and are gathered by the UN Inter-agency Group for Child Mortality Estimation, relying largely on vital registration systems and/or estimations based on censuses and surveys. See <a href="http://www.childmortality.org/">http://www.childmortality.org/</a> for additional details on data construction and sources. The corresponding data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

Life Expectancy. This variable refers to the years of life expectancy at birth for both sexes. This measure can be interpreted as the years an individual can be expected to live if the life conditions or mortality remain unchanged. It provides a summary of mortality patterns throughout all age groups in a given year. This variable is part of the World Bank's World Development Indicators database. The corresponding data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

Disability-Adjusted Life-Years. This variable is used to quantify the burden of disease from mortality and morbidity and corresponds to the years of healthy life lost. According to the World Health Organization, the sum of disability adjusted life years (DALY) can be interpreted as the gap between an ideal health situation (no diseases and disability) and the actual health status. For details on the calculation of the variable see the project website, (http://www.who.int/healthinfo/global\_burden\_disease/metrics\_daly/en/). The data are available from the World Health Organization data repository at http://www.who.int/gho/database/en/.

Homicide Rate. This variable measures the country's number of intentional homicides per 100,000 people. The data are provided by the United Nations Office on Drugs and Crime (UNODC) and was gathered between 2000 and 2015 as supplied by the respective member countries. Homicide is defined by the UNODC as "[...] unlawful death purposefully inflicted on a person by another person. Data on intentional homicide should also include serious assault leading to death and death as a result of a terrorist attack. It should exclude attempted homicide, manslaughter, death due to legal intervention, justifiable homicide in self-defence and death due to armed conflict". For additional details on data construction and sources, see the corresponding website at <a href="https://data.unodc.org/">https://data.unodc.org/</a>.

### **Economic Environment Variables**

GDP per Capita. This variable refers to the purchasing power parity (PPP) adjusted output-side real GDP per capita (expressed in 2011 USD). It is calculated using the output-side real GDP at chained PPPs divided by population size. Both variables are part of the Penn World Tables database, Version 9.0. See Feenstra et al. (2015) for additional details on data construction and sources. The corresponding data are freely available online at <a href="https://www.rug.nl/ggdc/productivity/pwt/">https://www.rug.nl/ggdc/productivity/pwt/</a>.

Government Consumption. This variable refers to the country's the share of government consumption in percentage of total GDP (expressed in 2011 USD). This variable is part of the Penn World Tables database, Version 9.0. See Feenstra et al. (2015) for additional details on data construction and sources. The corresponding data are freely available online at <a href="https://www.rug.nl/ggdc/productivity/pwt/">https://www.rug.nl/ggdc/productivity/pwt/</a>.

Fertility Rate. This variable refers to a country's total number of children born to women throughout the entire life course. As defined by the United Nations, the total rate of fertility represents the number of children expected to be born to a woman until the end of her childbearing years, using age-specific fertility rates for the population and the corresponding year. This variable is part of the World Bank's World Development Indicators database. The corresponding data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

**Population Growth.** This variable refers to the country's average annual population growth rate in percent. This variable is constructed using the country's population size, which is part of the Penn World Tables database, Version 9.0. See Feenstra et al. (2015) for additional details on data construction and sources. The corresponding data are freely available online at <a href="https://www.rug.nl/ggdc/productivity/pwt/">https://www.rug.nl/ggdc/productivity/pwt/</a>.

**Trade Openness.** This variable refers to the sum of exports and imports of goods and services over total GDP (measured in % of GDP). This variable is part of the World Bank's World Development Indicators database. The corresponding data are available online from the Quality of Government (QoG) database (Dahlberg et al., 2018).

### Social Value Variables

Women in Parliament. This variable refers to the proportion of seats held by women in single or lower chambers in national parliaments (as percentage of total seats). This variable is part of the World Bank's World Development Indicators database. The corresponding data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

**Female Economic Participation.** This variable refers to the country's share of female population aged 15 and above of the total labour force. This variable is part of the World Bank's World Development Indicators database. The corresponding data are

available online at http://data.worldbank.org/.

Federalist Political System. This variable refers to an indicator variable that equals 1 if the respective country is classified as having a federalist political system and zero otherwise. It is based on a categorical variable taken from the *Institutions and Elections Programme* (IAEP), indicating whether a country's governmental structure is a federalist system, unitary system or a confederation. See Wig et al. (2015) for additional details on data construction and sources. The corresponding data are available online at https://havardhegre.net/iaep/.

**Property Rights Protection.** This variable refers to a country's degree of property rights protection. The relevant metric ranges from 0 = low to 100 = high. This variable is part of the Heritage Foundation database at <a href="https://www.heritage.org/index/property-rights">https://www.heritage.org/index/property-rights</a>. The corresponding data are available online from the *Quality of Government* (QoG) database (Dahlberg et al., 2018).

Freedom from Corruption. This variable refers to Transparency International's Corruption Perception Index (CPI), indicating the country's level of freedom from corruption. The relevant metric ranges from 0 = low to 100 = high. This variable is part of the Heritage Foundation database at <a href="https://www.heritage.org/index/freedom-from-corruption">https://www.heritage.org/index/freedom-from-corruption</a>. The corresponding data are available online from the Quality of Government (QoG) database (Dahlberg et al., 2018).

Number of Civil Conflicts. This variable refers to the number of internal armed conflicts. This variable has been constructed from the UCDP/PRIO Armed Conflict database. The corresponding data are available online at <a href="http://ucdp.uu.se/downloads/">http://ucdp.uu.se/downloads/</a>.

## **Educational Variables**

Share of Educational Attainment. This vector refers to a set of controls indicating the share of a country's population aged 15 and above with no, primary, or secondary educational attainment. See Barro and Lee (2013) for additional details on data construction and sources. The corresponding data are available online at <a href="http://www.barrolee.com/">http://www.barrolee.com/</a>.

Human Capital Index. This variable refers to a human capital index. This index has been constructed from the Penn World Tables database, Version 9.0, by combining data on average years of schooling, as presented by Barro and Lee (2013); Cohen and Soto (2007); Cohen and Leker (2014), and the expected returns to education, as compiled by Psacharopoulos (1994). The latter accounts for possible diminishing returns to schooling quantity. This way, a lower return is assigned to higher education, as measured by threshold values for primary  $(s \le 4)$ , secondary  $(4 < s \le 8)$  and tertiary (s > 8) schooling. See Feenstra et al. (2015) for additional details on data construction and sources. The corresponding data are available online at https://www.rug.nl/ggdc/productivity/pwt/.

**Student-Teacher Ratio.** This variable refers to the country's average class size, i.e., the ratio of the number of students to the number of trained teachers in secondary schooling. This variable is part of the World Bank's World Development Indicators database. The corresponding data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

**Private Enrollment Rate.** This variable refers to the share of pupils enrolled in privately organized educational facilities in primary education. This variable is part of the World Bank's *World Development Indicators* database. The corresponding data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

**Educational Expenditure.** This variable refers to the country's percentage of GDP devoted to educational services. This variable is part of the World Bank's *World Development Indicators* database. The corresponding data are available online at <a href="http://data.worldbank.org/">http://data.worldbank.org/</a>.

## ADHD Study-Level Methodological Variables

The following study-level methodological characteristics are taken from or are constructed based on information in the metaanalysis on estimated ADHD prevalence rates compiled by Thomas et al. (2015). See the main text for additional details on data construction and sources.

**ADHD Prevalence Estimate.** This variable refers to estimated ADHD prevalence rates (measured as the share of individuals diagnosed with ADHD) across individual ADHD studies or countries, respectively.

Risk of Bias Effects. This variable refers to a set of indicators for a possible Risk of Bias (RoB) in estimated ADHD prevalence rates. Specifically, each individual ADHD study was evaluated according to eight RoB questionnaires regarding the representativeness of the sample, the sampling frame, whether individuals were randomly selected, minimum non-response bias, whether observations were directly collected from subjects, and whether data collection and construction fulfills a minimum of measurement reliability and validity. The calculated RoB score ranges from 0 (= very high risk of bias) to 8 (= low risk of bias score). Given the RoB score, each individual ADHD study was classified as having a very high RoB score (i.e.,  $0 \le RoB \le 1$ ), high RoB score (i.e.,  $0 \le RoB \le 1$ ), moderate RoB score (i.e.,  $0 \le RoB \le 1$ ), or low RoB score (i.e.,  $0 \le RoB \le 1$ ).

**DSM Diagnostic Criteria Effects.** This variable refers to a set of indicators for the DSM diagnostic criteria used in the respective study. Each indicator is assigned the value of 1 if the diagnostic criterion used for the prevalence estimate is DSM-III, DSM-III-R, or DSM-IV, respectively, and zero otherwise.

**Population Sample Type.** This variable refers to a set of indicators for the source of the sampled population. Each indicator is assigned the value of 1 if the sampled population originates from a schooling, community, or national representative survey sample, respectively.

Full Diagnostic Criteria Met. This indicator variable takes on the value of 1 if the full diagnostic criteria of the various editions of the DSM in the assessment of individual ADHD cases are met and zero otherwise.

**Type of ADHD Measurement.** This variable refers to a set of indicator variables for the type of ADHD measurement. Each indicator is set to 1 if diagnosed ADHD cases are measured using a diagnostic report, an interview, or a symptom-only checklist, respectively.

**Informant of ADHD.** This variable refers to a set of indicator variables for the type of informant reporting ADHD-related symptoms in probands. This comprises indicators for the reported ADHD cases being confirmed by child, parent, teacher, or clinician.

# **Zuletzt erschienen /previous publications:**

V-408-18	Bernhard C. Dannemann, Erkan Gören, The Educational Burden of ADHD:
	Evidence From Student Achievement Test Scores
V-407-18	<b>Jürgen Bitzer, Erkan Gören</b> , Foreign Aid and Subnational Development: A Grid Cell Analysis
V-406-17	Christoph Böhringer, Jan Schneider, Marco Springmann, Economic and
V-405-17	Environmental Impacts of Raising Revenues for Climate Finance from Public Sources <b>Erhard Glötzl, Florentin Glötzl, Oliver Richters</b> , From constrained optimization to constrained dynamics: extending analogies between economics and mechanics
V-404-17	Heinz, Welsch, Jan Kühling, How Green Self Image Affects Subjective Well-Being: Pro-Environmental Values as a Social Norm
V-403-17	Achim Hagen, Jan Schneider, Boon or Bane? Trade Sanctions and the Stability of International Environmental Agreements
V-402-17	Erkan Gören, The Role of Novelty-Seeking Traits in Contemporary Knowledge Creation
V-401-17	Heinz Welsch, Jan Kühling, Divided We Stand: Immigration Attitudes, Identity, and
V-400-17	Subjective Well-Being Christoph Böhringer, Thomas F. Rutherford, Paris after Trump: An inconvenient
W 200 17	insight
V-399-17	Frank Pothen, Heinz Welsch, Economic Development and Material Use
V-398-17	Klaus Eisenack, Marius Paschen, Designing long-lived investments under uncertain and ongoing change
V-397-16	<b>Marius Paschen</b> , The effect of intermittent renewable supply on the forward premium in German electricity markets
V-396-16	<b>Heinz Welsch, Philipp Biermann</b> , Poverty is a Public Bad: Panel Evidence from Subjective Well-being Data
V-395-16	Philipp Biermann, How Fuel Poverty Affects Subjective Well-Being: Panel Evidence from Germany
V-394-16	Heinz Welsch, Electricity Externalities, Siting, and the Energy Mix: A Survey
V-393-16	Leonhard Kähler, Klaus Eisenack, Strategic Complements in International Environmental Agreements: a New Island of Stability
V-392-16	Christoph Böhringer, Xaquin Garcia-Muros, Ignacio Cazcarro, Iñaki Arto, The Efficiency Cost of Protective Measures in Climate Policy
V-391-16	Achim Hagen, Juan-Carlos Altamirano-Cabrera, Hans-Peter Weikard, The Influence of Political Pressure Groups on the Stability of International Environmental
V-390-16	Agreements Christoph Böhringer, Florian Landis, Miguel Angel Tovar Reaños, Costeffectiveness and Incidence of Renewable Energy Promotion in Germany
V-389-16	Carsten Helm, Mathias Mier, Efficient diffusion of renewable energies: A roller-coaster ride
V-388-16	Christoph Böhringer, Jan Schneider, Emmanuel Asane-Otoo, Trade In Carbon and The Effectiveness of Carbon Tariffs
V-387-16	Achim Hagen, Leonhard Kähler, Klaus Eisenack, Transnational Environmental
V-386-15	Agreements with Heterogeneous Actors  Jürgen Bitzer, Erkan Gören, Sanne Hiller, Absorption of Foreign Knowledge:  Firms', Parafite of Franksking Laurisanska
V-385-15	Firms' Benefits of Employing Immigrants Klaus Eisenack, Julien Minnemann, Paul Neetzow, Felix Reutter, Contributions
V-384-15	to the institutional economics of the energy transition Christoph Böhringer, Xaquín Garcia-Muros, Mikel Gonzalez-Eguino, Luis Rey,

V-382-15 Christoph Böhringer, Brita Bye, Taran Fæhn, Knut Einar Rosendahl, Output-based rebating of carbon taxes in the neighbor's backyard
 V-381-15 Christoph Böhringer, Markus Bortolamedi, Sense and No(n)-Sense of Energy

Christoph Böhringer, Edward J. Balistreri, Thomas F. Rutherford, Carbon

V-381-15 **Christoph Bohringer, Markus Bortolamedi**, Sense and No(n)-Sense of Energy Security Indicators

US Climate Policy: A Critical Assessment of Intensity Standards

policy and the structure of global trade

V-383-15

V-380-15	Christoph Böhringer, Knut Einar Rosendahl, Halvor Briseid Storrøsten, Mitigating carbon leakage:Combining output-based rebating with a consumption tax
V-379-15	Jan Micha Steinhäuser, Klaus Eisenack, Spatial incidence of large-scale power plant curtailment costs
V-378-15	Carsten Helm, Franz Wirl, Climate policies with private information: The case for unilateral action
V-377-15	<b>Klaus Eisenack</b> , Institutional adaptation to cooling water scarcity in the electricity sector under global warming
V-376-15	Christoph Böhringer, Brita Bye, Taran Fæhn, and Knut Einar Rosendahl, Targeted carbon tariffs – Carbon leakage and welfare effects
V-375-15	<b>Heinz Welsch, Philipp Biermann</b> , Measuring Nuclear Power Plant Externalities Using Life Satisfaction Data: A Spatial Analysis for Switzerland
V-374-15	<b>Erkan Gören</b> , The Relationship Between Novelty-Seeking Traits And Comparative Economic Development
V-373-14	Charlotte von Möllendorff, Heinz Welsch, Measuring Renewable Energy Externalities: Evidence from Subjective Well-Being Data
V-372-14	<b>Heinz Welsch, Jan Kühling</b> , Affective States and the Notion of Happiness: A Preliminary Analysis
V-371-14	Carsten Helm, Robert C. Schmidt, Climate cooperation with technology investments and border carbon adjustment
V-370-14	Christoph Böhringer, Nicholas Rivers, Hidemichi Yonezawa, Vertical fiscal externalities and the environment
V-369-14	<b>Heinz Welsch, Philipp Biermann,</b> Energy Prices, Energy Poverty, and Well-Being: Evidence for European Countries
V-368-14	Marius Paschen, Dynamic Analysis of the German Day-Ahead Electricity Spot Market
V-367-14	Heinz Welsch, Susana Ferreira, Environment, Well-Being, and Experienced Preference
V-366-14	Erkan Gören, The Biogeographic Origins of Novelty-Seeking Traits
V-365-14	Anna Pechan, Which Incentives Does Regulation Give to Adapt Network Infrastructure to Climate Change? - A German Case Study  Chairteanh Böhningen André Miller, Jon Schneiden Couken Teriffe Benieted
V-364-14	Christoph Böhringer, André Müller, Jan Schneider, Carbon Tariffs Revisited Christoph Böhringer, Alexander Cuntz, Diemtar Harhoff, Emmanuel A. Otoo,
V-363-14	The Impacts of Feed-in Tariffs on Innovation: Empirical Evidence from Germany
V-362-14	Christoph Böhringer, Nicholas Rivers, Thomas Ruhterford, Randall Wigle,
V-302-14	Sharing the burden for climate change mitigation in the Canadian federation
V-361-14	Christoph Böhringer, André Müller, Environmental Tax Reforms in Switzerland A
V-301-14	Computable General Equilibrium Impact Analysis
V-360-14	Christoph Böhringer, Jared C. Carbone, Thomas F. Rutherford,
7 300 11	The Strategic Value of Carbon Tariffs
V-359-13	Heinz Welsch, Philipp Biermann, Electricity Supply Preferences in Europe:
, 50, 10	Evidence from Subjective Well-Being Data
V-358-13	Heinz Welsch, Katrin Rehdanz, Daiju Narita, Toshihiro Okubo, Well-being
	effects of a major negative externality: The case of Fukushima
V-357-13	Anna Pechan, Klaus Eisenack, The impact of heat waves on electricity spot markets
V-356-13	Heinz Welsch, Jan Kühling, Income Comparison, Income Formation, and
	Subjective Well-Being: New Evidence on Envy versus Signaling
V-355-13	Christoph Böhringer, Knut Einar Rosendahl, Jan Schneider, Unilateral Climate
	Policy: Can Opec Resolve the Leakage Problem?
V-354-13	Christoph Böhringer, Thomas F. Rutherford, Marco Springmann; Clean-
	Development Investments:An Incentive-Compatible CGE Modelling Framework
V-353-13	Erkan Gören, How Ethnic Diversity affects Economic Development?
V-352-13	<b>Erkan Gören,</b> Economic Effects of Domestic and Neighbouring Countries' Cultural Diversity
V-351-13	<b>Jürgen Bitzer, Erkan Gören</b> , Measuring Capital Services by Energy Use: An Empirical Comparative Study